UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of report (Date of earliest event reported): May 30, 2008

MASTEC, INC.

(Exact Name of Registrant as Specified in Its Charter)

Florida

(State or Other Jurisdiction of Incorporation)

Florida 0-08106 65-0829355

(State or other jurisdiction (Commission File (IRS Employer of incorporation) Number) Identification No.)

800 S. Douglas Road, 12th Floor, Coral Gables, Florida 33134

(Address of Principal Executive Offices) (Zip Code)

(305) 599-1800

(Registrant's Telephone Number, Including Area Code)

N/A

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

ITEM 1.01 Entry into a Material Definitive Agreement.

ITEM 2.01 Completion of Acquisition or Disposition of Assets.

ITEM 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

On May 30, 2008 (the "Closing Date"), MasTec, Inc., a Florida corporation ("MasTec") through its subsidiary MasTec North America, Inc., a Florida corporation (the "Buyer") entered into a Stock Purchase Agreement (the "Purchase Agreement"), dated as of May 1, 2008, with Alan B. Roberts (the "Seller"), pursuant to which the Buyer purchased all of the issued and outstanding shares of capital stock (the "Shares") of Pumpco, Inc. ("Pumpco"), a Texas corporation engaged in midstream oil and gas pipeline construction (the "Acquisition").

Pursuant to the terms of the Purchase Agreement, the purchase price for the Acquisition consists of \$44 million, which was paid in cash on the Closing Date (subject to adjustment as set forth in the Purchase Agreement) and earn-out payments payable over a five-year period equal to fifty percent of Pumpco's earnings before taxes above a significant threshold, as set forth in the Purchase Agreement (the "Earn-Out"). The Earn-Out is payable in cash, MasTec common stock or a combination thereof as set forth in the Purchase Agreement. At closing, Pumpco had approximately \$17 million in indebtedness, including \$12.4 million of equipment financing.

The foregoing summary of the Purchase Agreement is not complete and is qualified in its entirety by reference to the Purchase Agreement, a copy of which is filed herewith as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

In connection with the Acquisition, on May 30, 2008, the Company entered into an equipment term loan in the aggregate principal amount of \$22.5 million (the "Equipment Term Loan") with General Electric Capital Corporation ("GE"), which proceeds were used to pay off \$8.7 million of Pumpco indebtedness with the balance used to pay in part the cash portion of the Acquisition. The Equipment Term Loan was pursuant to several promissory notes substantially in the form of Exhibit 10.2 to this Current Report on Form 8-K. The Equipment Term Loan is secured by most of Pumpco's existing equipment, as set forth in the Master Security Agreement dated May 30, 2008 between Pumpco and GE (the "Security Agreement"). The Equipment Term Loan is payable in 60 monthly installments and bears interest at a fixed rate of 7.05%. Any prepayments within the first three years of the Equipment Term Loan is subject to a prepayment penalty of 3%, 2% or 1% of the then outstanding principal balance, for any unscheduled prepayments made during year one, two or three, respectively, of the Equipment Term Loan. MasTec has guaranteed the Equipment Term Loan pursuant to a Guaranty dated May 30, 2008 between MasTec and GE (the "Guaranty"). The foregoing summary of the Equipment Term Loan, Security Agreement and Guaranty is not complete and is qualified in its entirety by reference to such documents a copy of each of which is filed herewith as Exhibits 10.2, 10.3 and 10.4, respectively, to this Current Report on Form 8-K and is incorporated herein by reference.

ITEM 9.01 Financial Statements and Exhibits.

(a) Financial Statements of Businesses Acquired.

The financial statements required by this Item 9.01(a) will be filed by amendment not later than 71 calendar days after the date that this Form 8-K must be filed.

(b) Pro Forma Financial Information.

The financial information required by this Item 9.01(b) will be filed by amendment not later than 71 calendar days after the date that this Form 8-K must be filed.

(c) Shell Company Transactions.

Not applicable.

(d) Exhibits.

- 10.1 Stock Purchase Agreement executed on May 30, 2008 and dated as of May 1, 2008, between MasTec North America, Inc., as buyer, and Alan B. Roberts, as seller.
 - 10.2 Form of Promissory Note for the Equipment Term Loan dated May 30, 2008 between Pumpco, Inc. and General Electric Capital Corporation.
 - 10.3 Master Security Agreement dated May 30, 2008 between Pumpco, Inc. and General Electric Capital Corporation.
 - 10.4 Corporate Guaranty dated May 30, 2008 from MasTec, Inc. to General Electric Capital Corporation.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

MASTEC, INC.

Date: June 5, 2008 By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial

Officer

EXHIBIT INDEX

Description

Exhibit No.

10.1	Stock Purchase Agreement executed on May 30, 2008 and dated as of May 1, 2008, between MasTec North America, Inc., as buyer, and Alan B. Roberts, as seller.
10.2	Form of Promissory Note for the Equipment Term Loan dated May 30, 2008 between Pumpco, Inc. and General Electric Capital Corporation.
10.3	Master Security Agreement dated May 30, 2008 between Pumpco, Inc. and General Electric Capital Corporation.
10.4	Corporate Guaranty dated May 30, 2008 from MasTec, Inc. to General Electric Capital Corporation.

STOCK PURCHASE AGREEMENT

Dated as of May 1, 2008

between

MasTec North America, Inc., the Buyer

and

Alan B. Roberts, the Seller

TABLE OF CONTENTS

1. PURCHASE AND SALE OF SHARES.

1.1	Purchase and Sale of Shares	1
1.2	Purchase Price	1
1.3	Purchase Price Adjustments	1
1.4	The Closing	3
1.5	Payments	3
1.6	Earn-out	4
REPI	RESENTATIONS AND WARRANTIES OF THE SELLER.	6
2.1	Organization; Predecessors	7
2.2	Capitalization of the Company; Title to Shares	7
2.3	Power and Authorization	8
2.4	Authorization of Governmental Authorities	8
2.5	Noncontravention	8
2.6	Financial Statements	9
	Absence of Undisclosed Liabilities	10
2.8	Absence of Certain Developments	10
2.9	Debt; Guarantees	12
2.10	Assets	12
2.11	Accounts Receivable	12
	Real Property	12
	Equipment	14
	Intellectual Property	14
	Legal Compliance; Illegal Payments; Permits	15
	Inventories	18
	Employee Benefit Plans	18
	Environmental Matters	20
	Contracts	21
	Affiliate Transactions	23
	Customer and Supplier	23
	Customer Warranties	24
	Capital Expenditures and Investments	24
	Employees	24
	Litigation; Governmental Orders	26
	Insurance	26
	Banking Facilities	27
	Powers of Attorney	27
	No Brokers	27
	Adequacy of Internal Controls	27
	Disclosure	27
	Securities Law Matters	27
2.33	Company Operations	28

3. REPRESENTATIONS AND WARRANTIES OF THE BUYER.		RESENTATIONS AND WARRANTIES OF THE BUYER.	28
	3.2 3.3 3.4 3.5	Organization Power and Authorization Authorization of Governmental Authorities Noncontravention No Brokers Securities Law Compliance	28 28 28 28 29 29
4. COVENANTS.		ENANTS.	29
	4.2 4.3 4.4 4.5 4.6	Release Confidentiality Publicity Covenant Against Competition Further Assurances Cooperation Regarding Financial Statements Corrective Action	29 30 30 32 33 33
5.	INDI	EMNIFICATION.	33
	5.2 5.3 5.4 5.5 5.6 5.7	Indemnification by the Seller Indemnity by the Buyer Time for Claims Third Party Claims Knowledge and Investigation Insurance Remedies Cumulative Right of Setoff	33 34 34 35 37 37 37
6.	TAX	MATTERS	37
	6.2	Representations and Obligations Regarding Taxes Covenants With Respect To Taxes Indemnification for Taxes	37 41 42
7.	MISO	CELLANEOUS	45
	7.2 7.3 7.4 7.5 7.6 7.7 7.8 7.9	Notices Succession and Assignment; No Third-Party Beneficiary Amendments and Waivers Entire Agreement Schedules; Listed Documents, etc. Counterparts; Execution Survival Severability Headings	45 46 46 47 47 47 47
	/.10	Construction	47

	/.11	Governing Law	48
	7.12	Jurisdiction; Venue; Service of Process	48
	7.13	Waiver of Jury Trial	48
	7.14	Expenses	49
8. DEFINITIONS; CERTAIN RULES OF CONSTRUCTION.		INITIONS; CERTAIN RULES OF CONSTRUCTION.	49
	8.1 8.2	Definitions Rules of Construction	49 58
	0.2	Nuics of Construction	50

SCHEDULES

2.5	Noncontravention
2.6	Financial Statements
2.6.1	Excluded Assets
2.8	Certain Developments
2.9	Debt
2.10	Assets
2.11	Accounts Receivable
2.12	Real Property
2.14	Intellectual Property
2.15.3	Permits
2.17(a)	Employee Benefit Plans
2.18	Environmental Matters
2.19	Contracts
2.19.4	Bids
2.20	Affiliate Transactions
2.21	Customers and Suppliers
2.22	Customer Warranty Claims
2.23	Capital Expenditures and Investments
2.24	Employees
2.25	Litigation
2.26	Insurance
2.27	Banking Facilities
2.28	Powers of Attorney
4.4.1(a)	Employees Who May be Solicited
6	Taxes
6.1(a)	Tax Returns
6.1(d)	Tax Clearances
6.1(e)	Tax Audits and Disputes
6.1(l)	Tax Claims or Elections for any Reliefs, Allowances or Credits
6.1(p)	Tax Residence

Predecessor Status, etc. Capitalization of the Company

2.1.2 2.2

STOCK PURCHASE AGREEMENT

This Stock Purchase Agreement dated as of May 1, 2008 (the "<u>Effective Date</u>"), as amended or otherwise modified (the "<u>Agreement</u>") is between MasTec North America, Inc., a Florida corporation (the "<u>Buyer</u>"), and Alan B. Roberts, an individual (the "<u>Seller</u>").

RECITALS

WHEREAS, the Seller is the record and beneficial owner of all of the issued and outstanding shares of capital stock (the "Shares") of Pumpco, Inc., a Texas corporation (the "Company"); and

WHEREAS, the Buyer desires to purchase from the Seller, and the Seller desires to sell to the Buyer, all of the Shares upon the terms and subject to the conditions set forth in this Agreement.

AGREEMENT

NOW THEREFORE, in consideration of the premises and mutual promises herein made, and in consideration of the representations, warranties and covenants herein contained, the Buyer and the Seller hereby agree as follows:

1. PURCHASE AND SALE OF SHARES.

- 1.1 <u>Purchase and Sale of Shares</u>. At the Closing, subject to the terms and conditions of this Agreement, the Seller will sell, transfer and deliver to the Buyer, and the Buyer will purchase from the Seller, the Shares free and clear of any and all Encumbrances.
- 1.2 <u>Purchase Price</u>. As consideration for the Shares, the Buyer shall pay the Seller an amount equal to Forty-Four Million Dollars (\$44,000,000) as adjusted pursuant to <u>Section 1.3</u> (the "<u>Cash Purchase Price</u>").

1.3 Purchase Price Adjustments.

1.3.1 The Seller acknowledges that the amount of the Purchase Price is based, in addition to the other terms and conditions of this Agreement, on the Effective Date (a) the Company having a tangible net worth ((x) assets of the Company, minus (y) liabilities of the Company, minus (z) intangible assets of the Company) determined in accordance with GAAP (except for the impact of using tax basis depreciation for fixed assets), equal to or exceeding Twelve Million Eight Hundred Thirty-Eight Thousand Dollars (\$12,838,000) (the "Minimum Tangible Net Worth"); (b) the Company having total assets determined in accordance with GAAP (except for the impact of using tax basis depreciation for fixed assets), equal to or exceeding Thirty-Two Million Eight Hundred Fifty-Nine Thousand Dollars (\$32,859,000) (the "Minimum Total Assets") of which (A) a minimum of Four Million Five Hundred Thousand Dollars (\$4,500,000) must be in the form of cash, and Accounts Receivable no older than ninety (90) days and (B) property, plant and equipment, including, but not limited to trucks, trailers and other construction equipment having a net book value of at least Twenty-One Million Dollars (\$21,000,000); (c) the Company having net working capital ((x) cash and Accounts Receivable

no older than ninety (90) days minus (y) accounts payable and accrued liabilities) determined in accordance with GAAP (except for the impact of using tax basis depreciation for fixed assets) of at least Three Million One Hundred Fifty-Three Thousand Dollars (\$3,153,000) (the "Minimum Net Working Capital"); and (d) the Company having Debt that is no greater than Thirteen Million Eight Hundred Twenty-Five Thousand Dollars (\$13,825,000) (the "Maximum Company Debt").

- 1.3.2 As soon as practicable but in no event more than ninety (90) days following the Closing, the Buyer shall prepare, or cause to be prepared, and deliver to the Seller (a) a balance sheet (the "Closing Balance Sheet") of the Company as of the Effective Date prepared in accordance with GAAP, except for the impact of using tax basis depreciation for fixed assets and no accrual shall be made with respect to Taxes for the period from the Effective Date to the Closing Date and with respect to the distribution and transfer of the Excluded Assets, (b) a closing statement as of the Effective Date (the "Closing Statement") setting forth the Company's actual tangible net worth (the "Actual Tangible Net Worth"), the Company's actual total assets (the "Actual Total Assets"), the Company's actual net working capital (the "Actual Net Working Capital") and the Company's actual Debt (the "Actual Company Debt"), and (c) supporting documentation for the foregoing as reasonably requested by the Seller.
- 1.3.3 The Seller and its accountants shall complete their review of the Closing Balance Sheet, Closing Statement, and supporting documentation within 30 days after delivery thereof by the Buyer. If the Seller objects to the Closing Balance Sheet or Closing Statement for any reason, the Seller shall, on or before the last day of such 30-day period, so inform the Buyer in writing (a "Seller's Objection"), setting forth a specific description of the basis of the Seller's determination and the adjustments to the Closing Balance Sheet or Closing Statement that the Seller believes should be made. If no Seller's Objection is received by the Buyer on or before the last day of such 30-day period, then the Closing Balance Sheet and Closing Statement delivered by the Buyer shall be final. The Buyer shall have 15 days from its receipt of the Seller's Objection to review and respond to the Seller's Objection.
- 1.3.4 If the Seller and the Buyer are unable to resolve all of their disagreements with respect to the proposed adjustments set forth in the Seller's Objection within 15 days following the completion of the Buyer's review of the Seller's Objection, they shall refer any remaining disagreements to the CPA Firm which, acting as experts and not as arbitrators, shall determine, on the basis set forth in and in accordance with this Section 1.3, and only with respect to the remaining differences so submitted, whether and to what extent, if any, the Closing Balance Sheet and Closing Statement require adjustment. The Buyer and the Seller shall instruct the CPA Firm to deliver its written determination to the Buyer and the Seller no later than 30 days after the remaining differences underlying the Seller's Objection are referred to the CPA Firm. The CPA Firm's determination shall be conclusive and binding upon the Buyer and the Seller. In resolving any disputed item, the CPA Firm may not assign a value to any disputed item that is greater than the greatest value claimed by either party or less than the smallest value claimed by either party for the item. The fees and disbursements of the CPA Firm shall be borne (i) by the Seller in the proportion that the aggregate dollar amount of the disputed items that are unsuccessfully disputed by the Seller (as finally determined by the CPA Firm) bears to the aggregate dollar amount of all disputed items and (ii) by the Buyer in the proportion that the aggregate dollar amount of the disputed items that are successfully disputed by the Seller (as

finally determined by the CPA Firm) bears to the aggregate dollar amount of all disputed items. For example, if the parties dispute \$100,000 of a proposed adjustment to be paid by the Seller, the CPA Firm determines that such adjustment should be \$40,000 and the CPA Firm's fees are \$50,000, then (i) the Seller shall pay \$20,000 (40%) of such fees and (ii) the Buyer shall pay \$30,000 (60%) of such fees. The Buyer and the Seller shall make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants, to the extent permitted by such accountants) relating to the Closing Balance Sheet, Closing Statement and the Seller's Objection and all other items reasonably requested by the CPA Firm in connection therewith.

- 1.3.5 Upon the final determination of Actual Tangible Net Worth, Actual Total Assets, Actual Net Working Capital and Actual Company Debt as set forth in this Section 1.3, the Cash Purchase Price will be reduced on a dollar-for-dollar basis by (a) the amount, if any, by which Minimum Tangible Net Worth exceeds Actual Tangible Net Worth; (b) the amount, if any, by which Minimum Total Assets exceeds Actual Total Assets; (c) the amount, if any, by which Minimum Net Working Capital exceeds Actual Net Working Capital; (d) the amount, if any, by which the Actual Company Debt exceeds the Maximum Company Debt; and (e) the amount, if any, by which (i) the total of actual cash and Accounts Receivable no older than ninety (90) days, is less than Four Million Five Hundred Thousand Dollars (\$4,500,000) and/or (ii) the net book value of property, plant and equipment, including, but not limited to trucks, trailers and other construction equipment is less than Twenty One Million Dollars (\$21,000,000); provided that there shall be no duplication of any reduction required by this Section 1.3. If the Actual Net Working Capital exceeds Six Million Five Hundred Thousand Dollars (\$6,500,000) (the "Maximum Net Working Capital"), the Cash Purchase Price shall be increased on a dollar-for-dollar basis by the amount by which the Actual Net Working Capital exceeds the Maximum Net Working Capital. If any adjustment to the Cash Purchase Price is required by this Section 1.3, the Seller or the Buyer, as applicable, shall promptly (and in any event within five Business Days) after the final determination thereof pay to the other party the purchase price adjustment amount by wire transfer of immediately available funds to an account designated by such party. For purposes of clarity, if the net adjustment pursuant to this Section 1.3 results in (i) an increase to the Purchase Price, the amount of the increase shall be paid by the Buyer to the Seller; or (ii) a decrease to the Purchase Price, the amount of the decrease shall be paid by the Selle
- 1.4 <u>The Closing</u>. Subject to the terms and conditions of this Agreement, the sale and purchase of the Shares contemplated by this Agreement shall take place at a closing (the "<u>Closing</u>") to be held on May 30, 2008 (the "<u>Closing Date</u>"), simultaneously at the offices of the Buyer and the Seller. Upon consummation, the Closing will be deemed for all purposes to have taken place as of 12:01 a.m. on the Effective Date, except for the filing of the Company's Tax Return to be prepared by the Seller as of the Closing Date and the distribution to the Seller of the Excluded Assets, which shall be deemed to have occurred as of the Closing Date.
- 1.5 <u>Payments</u>. At the Closing, the Buyer will deliver to the Seller the Cash Purchase Price (by wire transfer of immediately available federal funds to the account(s) furnished by the Seller) against delivery to the Buyer of certificates evidencing the Shares duly endorsed (or accompanied by duly executed stock transfer powers). The Seller will furnish its account

information to the Buyer in writing not fewer than two Business Days prior to the scheduled Closing Date.

1.6 Earn-out.

- 1.6.1 In addition to the Cash Purchase Price, and as additional consideration for the purchase of the Shares, the Buyer shall pay the Seller contingent cash payments (the "Earn-Out Payments", and together with the Cash Purchase Price, the "Purchase Price") as determined pursuant to this Section 1.6 at the times, in the manner and to the extent Earn-Out Payments are earned pursuant to the following terms:
 - (a) with respect to the Company's 2008 Earn-out Period EBT, 50% of the Company's 2008 Earn-out Period EBT above \$3,750,000 will be paid to the Seller.
 - (b) with respect to the Company's 2009 Earn-out Period EBT, 50% of the Company's 2009 Earn-out Period EBT above \$7,500,000 will be paid to the Seller.
 - (c) with respect to the Company's 2010 Earn-out Period EBT, 50% of the Company's 2010 Earn-out Period EBT above \$7,500,000 will be paid to the Seller.
 - (d) with respect to the Company's 2011 Earn-out Period EBT, 50% of the Company's 2011 Earn-out Period EBT above \$7,500,000 will be paid to the Seller.
 - (e) with respect to the Company's 2012 Earn-out Period EBT, 50% of the Company's 2012 Earn-out Period EBT above \$7,500,000 will be paid to the Seller.
 - (f) with respect to the Company's 2013 Earn-out Period EBT, 50% of the Company's 2013 Earn-out Period EBT above \$3,750,000 will be paid to the Seller.
 - (g) Notwithstanding the foregoing, if the Company's EBT for any Earn-out Period is negative taking into account the adjustment to such EBT resulting from a prior Negative EBT Earn-out Period as set forth in this <u>subsection 1.6.1(g)</u> (any such year, a "<u>Negative EBT Earn-out Period</u>"), the Company's EBT for the subsequent period shall be reduced by the amount by which EBT for the Negative EBT Earn-out Period was less than zero.
- 1.6.2 For purposes of this <u>Section 1.6</u>, the Company's "<u>Earn-out Period</u>" shall mean the period commencing on January 1 and ending of December 31 of the relevant year, provided, that (i) the "<u>Earn-out Period</u>" for 2008 shall commence on July 1, 2008 and end on December 31, 2008 and (ii) the "<u>Earn-out Period</u>" for 2013 shall commence on January 1, 2013 and end on June 30, 2013.

1.6.3 For purposes of this Section 1.6, "EBT" means for any Earn-out Period the sum of (a) the net income or loss generated by the Company's operations as they exist on the Effective Date and as they may grow or contract after the Effective Date in the normal course of business through internal growth and not through acquisitions or other extraordinary or non-recurring transactions (the "Existing Operations"), after deduction of all costs, expenses, interest, taxes, depreciation, amortization, and other proper charges (including without limitation (i) the cost of any bonuses or incentive payments earned with respect to the applicable period (excluding any Earn-Out Payment), (ii) the cost of any equity granted to employees, or other derivative securities granted during the applicable period, valued in accordance with the Black-Scholl model and amortized over the vesting period of any equity granted, (iii) the cost of capitalized leases, amortized over the lease term, calculated in accordance with GAAP, except for the impact of using tax basis depreciation for fixed assets and (iv) all interest and amortization with respect to the Company's Debt, including all Debt reflected on the Closing Date Balance Sheet) plus (b) (i) federal, state, or local taxes attributable to the Existing Operations for such period, in each case as determined in accordance with GAAP, except for the impact of using tax basis depreciation for fixed assets, and (ii) any amortization of goodwill solely attributable to the transactions contemplated hereunder, but only to the extent that such taxes or amortized goodwill were deducted in determining the Company's net income or loss. In determining the costs to be deducted in calculating net income under subparagraph (a), any costs for services or benefits provided to the Company by the Buyer or any Affiliate of the Buyer shall not exceed the actual cost of furnishing such services or benefits.

1.6.4 Except (i) if there is a disagreement as described in Section 1.6.6 or (ii) in the case of the Earn-Out Payment for the 2013 Earn-out Period, which shall be paid on October 15, 2013, the Buyer will pay to the Seller the Earn-Out Payment with respect to a specific Earn-out Period, if any Earn-Out Payment is owed, no later than April 15th of the next succeeding year, at the Buyer's option (i) in cash in immediately available funds or (ii) in MasTec, Inc. ("MasTec") common stock valued at the closing price of the common stock of MasTec on the New York Stock Exchange composite for the previous trading day or (iii) in any combination of cash or MasTec common stock. The maximum number of shares of MasTec common stock that may be issued hereunder is 672,088. If MasTec issues MasTec common stock for some, or any portion, of the Earn-Out Payment, such shares, at the time of their issuance, will be: (i) validly authorized shares and (ii) registered for resale pursuant to an effective registration statement that has been filed with the SEC.

1.6.5 The Seller will be entitled to the Earn-Out Payments, if earned, whether or not the Seller remains employed by the Buyer or the Company; provided that nothing in this Section 1.6 constitutes an agreement or understanding to employ the Seller for a term of years or otherwise to guarantee the Seller's employment with the Buyer or the Company.

1.6.6 On or prior to March 15th of each year after the expiration of each Earn-out Period (other than with respect to the Earn-Out Payment for the 2013 Earn-out Period, which will be calculated on or before September 15, 2013), the Buyer will calculate the Earn-Out Payment, if any, for such period and provide the Seller with the Buyer's calculation of the Earn-Out Payment together with reasonable supporting documentation. For thirty (30) days following the delivery of the Earn-Out Payment calculation to the Seller (the "Earn-Out Review Period"), the Seller may review the Earn-Out Payment calculation and other documentation. The Seller

must notify the Buyer, in writing, of any disagreement with the Earn-Out Payment calculation or documentation and the basis for the disagreement no later than the end of the Earn-Out Review Period. If the Seller does not notify the Buyer of a disagreement by the end of the Earn-Out Review Period, the Earn-Out Payment calculation prepared by the Buyer will be conclusive. If the Seller timely notifies the Buyer of a disagreement regarding the Earn-Out calculation and the parties are unable, through good faith negotiation, to resolve the disagreement within thirty (30) days after the end of the Earn-Out Review Period, the Buyer and the Seller shall submit the items in dispute to the CPA Firm (which shall act as experts and not as arbitrators) for resolution. The Buyer and the Seller shall instruct the CPA Firm to deliver its written determination to the Buyer and the Seller no later than 30 days after the dispute is referred to the CPA Firm. The CPA Firm's determination shall be conclusive and binding upon the Buyer and the Seller. In resolving any disputed item, the CPA Firm may not assign a value to the disputed item that is greater than the greatest value claimed by either party or less than the smallest value claimed by either party for the item. The fees and disbursements of the CPA Firm shall be borne (i) by the Seller in the proportion that the aggregate dollar amount of the disputed items that are unsuccessfully disputed by the Seller (as finally determined by the CPA Firm) bears to the aggregate dollar amount of all disputed items and (ii) by the Buyer in the proportion that the aggregate dollar amount of the disputed items that are successfully disputed by the Seller (as finally determined by the CPA Firm) bears to the aggregate dollar amount of all disputed items. For example, if the parties dispute \$1,000,000 of an Earn-Out Payment, the CPA Firm determines that \$400,000 should be included in the Earn-Out Payment and \$600,000 should be excluded and the CPA Firm's fees are \$50,000, then (i) the Seller shall pay \$30,000 (60%) of such fees and (ii) the Buyer shall pay \$20,000 (40%) of such fees. The Buyer and the Seller shall make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants, to the extent permitted by such accountants) relating to the determination of the Earn-Out Payment and all other items reasonably requested by the CPA Firm in connection therewith.

1.6.7 Following the Closing Date, the Buyer will continue to operate the Company during the period for which the Seller is entitled to Earn-Out Payments substantially as previously operated, subject to the business requirements of the Buyer and its Affiliates taken as a whole. The Buyer will be permitted, following the Closing Date, to make changes at its sole discretion to the operations, corporate organization, personnel, accounting practices, and other aspects of the Company and the Business so long as such changes are made in good faith, in the best interests of the Company, the Buyer, and their respective Affiliates or to conform to standard practices applicable generally to other MasTec Affiliates, and not with the specific intent of reducing amounts that otherwise would be payable to the Seller.

1.6.8 The Buyer will not be liable to the Seller for any Earn-Out Payments accruing for any time period commencing on the earliest date that the Seller violates any of the provisions of <u>Sections 4.2</u> or <u>4.4</u>.

2. REPRESENTATIONS AND WARRANTIES OF THE SELLER.

In order to induce the Buyer to enter into and perform this Agreement and to consummate the Contemplated Transactions, the Seller hereby represents and warrants to the Buyer as follows:

2.1 Organization; Predecessors.

- 2.1.1 Organization. The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Texas with full corporate power and authority to carry on its business as it is now being conducted and to own, operate and lease its properties and assets. The Company is duly qualified or licensed as a foreign corporation to do business and is in good standing in every jurisdiction in which the conduct of its business, and the ownership or lease of its properties, require it to be so qualified or licensed. The Company has no Subsidiaries and has no obligation to make any Investments in any Person. The Seller has delivered to the Buyer true, accurate and complete copies of (x) the Organizational Documents of the Company and (y) the minute books of the Company which contain records of all meetings held of, and other actions taken by, its stockholders, Board of Directors and any committees appointed by its Board of Directors.
- 2.1.2 Predecessors. <u>Schedule 2.1.2</u> sets forth a list of (a) any Person that has ever merged with or into the Company, (b) any Person a majority of whose capital stock (or similar outstanding ownership interests) has ever been acquired by the Company, (c) any Person all or substantially all of whose assets have ever been acquired by the Company and (d) any prior names of the Company or any Person described in clauses (a) through (c) (each such Person, a "<u>Predecessor</u>").

2.2 Capitalization of the Company; Title to Shares.

- 2.2.1 Outstanding Capital Stock. The authorized capital stock of the Company is as set forth on Schedule 2.2. The Company's Common Stock is its only class of capital stock. All of the outstanding shares of the capital stock of the Company have been duly authorized, validly issued, and are fully paid and non-assessable. None of the outstanding shares of Common Stock are subject to, nor were they issued in violation of, any purchase option, call option, right of first refusal or offer, preemptive right, subscription right or any similar right. The Company has not violated the 1933 Act, any state "blue sky" or securities laws, any other similar Legal Requirement or any preemptive or other similar rights of any Person in connection with the issuance or redemption of any of its capital stock. Except for the Shares, no shares of voting or non-voting capital stock, other equity interests, or other voting securities of the Company are issued, reserved for issuance or outstanding. There are no bonds, debentures, notes, other Debt or any other securities of the Company with voting rights (or convertible into, or exchangeable for, securities with voting rights) on any matters on which stockholders of the Company may vote.
- 2.2.2 Ownership. The shares of treasury stock held by the Company is set forth on Schedule 2.2. All of the Shares are held of record and beneficially owned by the Seller free and clear of all Encumbrances. The Seller has full right, power and authority to transfer and deliver to the Buyer valid title to the Shares, free and clear of all Encumbrances. Immediately following the Closing, the Buyer will be the record and beneficial owner of the Shares, and will have good and marketable title to the Shares, free and clear of all Encumbrances. The assignments, endorsements, stock powers and other instruments of transfer delivered by the Seller to the Buyer at Closing will be sufficient to transfer the Seller's entire interest, legal and beneficial, in the Shares to the Buyer. The Seller has delivered to the Buyer true, accurate and

complete copies of the stock ledger (or similar register) of the Company which reflects all issuances, transfers, repurchases and cancellations of shares of its capital stock.

- 2.2.3 Encumbrances, etc. Except as set forth set forth on Schedule 2.2, there are no outstanding securities, options, warrants, calls, rights, convertible or exchangeable securities or Contractual Obligations of any kind (contingent or otherwise) to which the Company is a party or by which it is bound obligating the Company to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other voting securities of the Company or obligating the Company to issue, grant, extend or enter into any such security, option, warrant, call, right or Contractual Obligation. There are no outstanding obligations of the Company (contingent or otherwise) to repurchase, redeem or otherwise acquire any shares of capital stock (or options or warrants to acquire any such shares) of the Company. There are no stock-appreciation rights, stock-based performance units, "phantom" stock rights or other Contractual Obligations or obligations of any character (contingent or otherwise) pursuant to which any Person is or may be entitled to receive any payment or other value based on the revenues, earnings or financial performance or other attribute of the Company or the Business or the Assets or calculated in accordance therewith or to cause the Company to file a registration statement under the 1933 Act, or which otherwise relate to the registration of any securities of the Company. Except as set forth on Schedule 2.2, there are no voting trusts, proxies or other Contractual Obligations of any character to which the Company or the Seller is a party or by which either of them is bound with respect to the issuance, holding, acquisition, voting or disposition of any shares of capital stock or similar interests of the Company. There are no existing Contractual Obligations between the Seller on the one hand, and any other Person, on the other hand, regarding the Shares.
- 2.3 <u>Power and Authorization</u>. The execution, delivery and performance by the Seller of this Agreement and the consummation of the Contemplated Transactions are within the power and authority of the Seller. This Agreement has been duly executed and delivered by the Seller and is a legal, valid and binding obligation of the Seller, Enforceable against the Seller in accordance with its terms.
- 2.4 <u>Authorization of Governmental Authorities</u>. No action by (including any authorization, consent or approval), or in respect of, or filing with, any Governmental Authority is required for, or in connection with, the valid and lawful (a) execution, delivery and performance by the Seller of this Agreement or (b) the consummation of the Contemplated Transactions by the Seller, provided, that, the Seller makes no representation or warranty regarding the applicability of or the need to comply with the Hart-Scott-Rodino Premerger Notification Act, 15 U.S.C. 18a.
- 2.5 <u>Noncontravention</u>. Except as disclosed on <u>Schedule 2.5</u>, neither the execution, delivery and performance by the Seller of this Agreement nor the consummation of the Contemplated Transactions will:
 - (a) violate any provision of any Legal Requirement applicable to the Seller or the Company;

- (b) result in a breach or violation of, default under, or give rise to a right for any third-party to terminate or any prepayment penalty under any Contractual Obligation of the Seller or the Company;
 - (c) result in the creation or imposition of an Encumbrance upon, or the forfeiture of, any Asset;
 - (d) result in a breach or violation of, or default under, the Organizational Documents of the Company; or
- (e) require any action by (including any authorization, consent or approval) or in respect of (including notice to), any Person under any Contractual Obligation of the Company or the Seller.

2.6 Financial Statements.

2.6.1 Financial Statements. Attached as <u>Schedule 2.6</u> is a copy of (a) the balance sheet of the Company as of April 30, 2008 (the "<u>Most Recent Balance Sheet Date</u>") and the related statement of income of the Company for the three months then ended (together with the Most Recent Balance Sheet, the "<u>Interim Financials</u>") and the reviewed balance sheets of the Company as of January 31, 2008, 2007, 2006 and 2005 and the related reviewed statements of income of the Company for the fiscal years then ended, accompanied in each case by any notes thereto and the report of the independent registered certified public accounting firm (collectively, the "<u>Reviewed Financials</u>", and together with the Interim Financials, the "<u>Financials</u>"). Since the Most Recent Balance Sheet Date, (i) the Company has not distributed, sold or otherwise disposed of any property or other assets other than in the Ordinary Course of Business, except for the distribution of the excluded assets set forth on <u>Schedule 2.6.1</u> (the "<u>Excluded Assets</u>") and (ii) the Company has not made or granted any cash distributions or dividends.

2.6.2 Compliance with GAAP, etc. The Financials (a) are complete and correct and were prepared in accordance with the books and records of the Company and (b) have been prepared in accordance with GAAP, except for the impact of using tax basis depreciation for fixed assets, subject, in the case of the Interim Financials, to normal month-end adjustments, the effect of which will not, individually or in the aggregate, be materially adverse to the Company. With respect to the Interim Financials, the absence of notes, if presented, would not differ materially from those included in the Reviewed Financials. The Financials fairly present the financial position of the Company as at the respective dates thereof and the results of the operations of the Company for the respective periods covered thereby, except that the Most Recent Balance Sheet reflects the distribution of the Excluded Assets but does not reflect any accrued Tax liabilities in connection therewith. The Company's utilization of the percentage of completion methodology is in conformity with GAAP. The gross profit and gross profit percentage of jobs currently in process have been properly recognized in accordance with GAAP. The Financials contain adequate reserves for the realization of all Assets and for all reasonably anticipated Liabilities in accordance with GAAP. The Company maintains adequate internal controls to assure the proper recording of all Assets, Liabilities and transactions in the Company's records and books of account and to safeguard the Company's Assets.

- 2.7 <u>Absence of Undisclosed Liabilities</u>. The Company has no Liabilities except for (a) Liabilities set forth on the face of the Most Recent Balance Sheet and (b) Liabilities incurred in the Ordinary Course of Business since the Most Recent Balance Sheet Date (none of which results from, arises out of, or relates to any breach or violation of, or default under, a Contractual Obligation or Legal Requirement and none of which constitutes a Material Adverse Effect) and to the Seller's Knowledge no facts or circumstances exist that could result in any such Liability.
- 2.8 <u>Absence of Certain Developments</u>. Except as set forth on <u>Schedule 2.8</u>, since the Most Recent Balance Sheet Date, the Business has been conducted in the Ordinary Course of Business and:
 - (a) the Company has not (i) except for an amendment dated May 27, 2008, amended its Organizational Documents, (ii) amended any term of its outstanding capital stock or (iii) issued, sold, granted, or otherwise disposed of, its capital stock;
 - (b) the Company has not become liable in respect of any Guarantee nor has it incurred, assumed or otherwise become liable in respect of any Debt, except for borrowings in the Ordinary Course of Business under credit facilities in existence on the Most Recent Balance Sheet Date;
 - (c) the Company has not sold, transferred or otherwise disposed of any of its Assets, except in the Ordinary Course of Business and the distribution of the Excluded Assets;
 - (d) the Company has not permitted any of its Assets to become subject to an Encumbrance other than a Permitted Encumbrance;
 - (e) the Company has not made or committed to make any capital expenditure except in the Ordinary Course of Business;
 - (f) the Company has not (i) made any declaration, setting aside or payment of any dividend or other distribution with respect to, or any repurchase, redemption or other acquisition of, any of its Common Stock or (ii) entered into, or performed, any transaction with, or for the benefit of, the Seller or any Affiliate of the Seller, except the distribution of the Excluded Assets;
 - (g) there has been no material loss, destruction, damage or eminent domain taking (in each case, whether or not insured) affecting the Business or any material Asset;
 - (h) the Company has not increased the Compensation payable or paid, whether conditionally or otherwise, to (i) any employee, consultant, independent contractor or agent other than in the Ordinary Course of Business, (ii) any director or officer of the Company or (iii) the Seller or any Affiliate of the Seller;
 - (i) the Company has not entered into any Contractual Obligation providing for the employment or consultancy of any Person on a full-time, part-time, consulting or other basis other than in the Ordinary Course of Business or otherwise providing Compensation or other benefits to any officer or director;

- (j) the Company has not made any change in its methods of accounting or accounting practices (including with respect to reserves) or its pricing policies, payment or credit practices or failed to pay any creditor any amount owed to such creditor when due or granted any extensions of credit other than in the Ordinary Course of Business;
- (k) the Company has not made, changed or revoked any material Tax election, elected or changed any method of accounting for Tax purposes, settled any Action in respect of Taxes or entered into any Contractual Obligation in respect of Taxes with any Governmental Authority;
 - (l) the Company has not terminated or closed any Facility, business or operation;
- (m) none of the customers or suppliers required to be disclosed on <u>Schedule 2.21</u> has canceled, terminated or, to the Seller's Knowledge, otherwise altered (including any reduction in the rate or amount of sales or purchases or change to the supply or credit terms, as the case may be) or notified the Company of any intention to do any of the foregoing or otherwise threatened in writing to cancel, terminate or materially alter (including any reduction in the rate or amount of sales or purchases, as the case may be) its relationship with the Company;
- (n) no insurer (i) has questioned, denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Liability Policy or (ii) has provided any notice of cancellation or any other indication that it plans to cancel any Liability Policy or raise the premiums or materially alter the coverage under any Liability Policy;
 - (o) the Company has not adopted any Employee Plan or increased any benefits under any Employee Plan;
- (p) the Company has not written off as uncollectible any Accounts Receivable or written up or written down any of its material Assets or revalued its Inventory;
- (q) the Company has not failed to make any scheduled capital expenditures or investments or failed to pay trade accounts payable or any other Liability when due;
 - (r) the Company has not failed to maintain or properly repair any of its Assets;
 - (s) the Company has not entered into any Contractual Obligation to do any of the things referred to elsewhere in this Section 2.8; and

- (t) to the Seller's Knowledge, no event or circumstance has occurred which has had, will have or may have, a Material Adverse Effect.
- 2.9 <u>Debt; Guarantees</u>. The Company has no Liabilities in respect of Debt except as set forth on <u>Schedule 2.9</u>. For each item of Debt, <u>Schedule 2.9</u> correctly sets forth the debtor, the principal amount of the Debt as of the Effective Date, the creditor, the maturity date and the collateral, if any, securing the Debt. The Company has no Liability in respect of a Guarantee of any Liability of any other Person.

2.10 Assets

- 2.10.1 Ownership of Assets. The Company has sole and exclusive, good and marketable title to, or, in the case of property held under a lease or other Contractual Obligation, a sole and exclusive, Enforceable leasehold interest in, or right to use, all of its properties, rights and assets, whether real or personal and whether tangible or intangible, including all Assets reflected in the Most Recent Balance Sheet or acquired after the Most Recent Balance Sheet Date (except for such Assets which have been sold or otherwise disposed of since the Most Recent Balance Sheet Date in the Ordinary Course of Business) (collectively, the "Assets"). Except as disclosed on Schedule 2.10, none of the Assets is subject to any Encumbrance other than Permitted Encumbrances.
- 2.10.2 Sufficiency of Assets. The Assets comprise all of the assets, properties and rights of every type and description, whether real or personal, tangible or intangible, used or necessary to conduct the Business and are adequate to conduct the Business.
- 2.11 Accounts Receivable. All Accounts Receivable, unbilled invoices, costs in excess of billings, work in process and other amounts ("Receivables") reflected on the Most Recent Balance Sheet and in the records and books of account of the Company since the Most Recent Balance Sheet Date through the Closing as being due to the Company have arisen in the Ordinary Course of Business, represent legal, valid, binding and enforceable obligations to the Company and, subject only to consistently recorded reserves for bad debts established as of a date prior to the Closing Date in a manner consistent with past practice, have been, or will be, collected or are, or will be, collectible in the aggregate recorded amounts thereof in accordance with their terms (and in no event later than the 90th day following the Effective Date, except for amounts representing retainage, which shall be collected when due) and, to the Seller's Knowledge, are not and will not be subject to any contests, claims, counterclaims or setoffs. There has been no material adverse change since the Most Recent Balance Sheet Date in the amount or collectability of the Receivables due the Company or the related provisions or reserves from that reflected in the Most Recent Balance Sheet. Except as set forth in Schedule 2.11, (i) no account debtor or note debtor is delinquent for payments in excess of \$50,000 or for more than ninety (90) days, (ii) no account debtor or note debtor has refused or, to the Seller's Knowledge, threatened to refuse to pay its obligations to the Company for any reason, or has otherwise made a claim to set-off or similar claim, and (iii) to the Seller's Knowledge no account debtor or note debtor is insolvent or bankrupt.

2.12 Real Property.

- 2.12.1 The Company owns no real property. Schedule 2.12 describes each leasehold interest in real property leased, subleased by, licensed or with respect to which a right to use or occupy has been granted to or by the Company (such leased real property is referred to as the "Real Property"), and specifies the lessor(s) of such leased property and identifies each lease or any other Contractual Obligation under which such property is leased (the "Real Property Leases"). Except as described on Schedule 2.12 there are no written or oral subleases, licenses, concessions, occupancy agreements or other Contractual Obligations granting to any other Person the right of use or occupancy of the Real Property and there is no Person (other than any lessee(s) of leased Real Property) in possession of the leased Real Property. With respect to each Real Property Lease that is a sublease, to the Seller's Knowledge, the representations and warranties set forth in Sections 2.19.2 and 2.19.3 are true and correct with respect to the underlying lease.
- 2.12.2 The Real Property Leases do not impose material restrictions on any portion of the Business that materially interfere with the Business. The Company is not obligated to pay any leasing or brokerage commission as a result of the Contemplated Transactions. There is no pending or, to the Seller's Knowledge, threatened eminent domain taking affecting any of the Real Property. The Seller has delivered to the Buyer true, correct and complete copies of the Real Property Leases including all amendments, modifications, notices or memoranda of lease thereto and all estoppel certificates or subordinations, non-disturbance and attornment agreements related thereto.
- 2.12.3 None of the Facilities currently existing on the Real Property encroaches upon, and any Facilities under construction on the Real Property will not encroach upon, the real property of any other Person. No facility of any other Person encroaches upon the Real Property. Each Facility is supplied with utilities and other services (including gas, electricity, water, drainage, sanitary sewer, storm sewer, fire protection and telephone) necessary for the operation of such Facility as the same is currently operated or proposed to be operated. Each parcel of Real Property abuts on, and has direct vehicular access to, a public road, or has access to a public road via a permanent, irrevocable appurtenant easement benefiting the parcel of Real Property, in each case, to the extent necessary for the conduct of the Business.
- 2.12.4 All Permits necessary in connection with the construction upon, and present use and operation of, the Real Property and the lawful occupancy thereof have been issued by the appropriate Governmental Authorities. The current use of the Real Property is in accordance with the certificates of occupancy relating thereto and the terms of any such Permits. All such Permits will continue in full force and effect immediately after giving effect to the Contemplated Transactions. The Real Property and its current use, occupancy and operation by the Company and the Facilities located thereon do not (a) constitute a nonconforming use under any applicable building, zoning, subdivision or other land use or similar Legal Requirements or (b) otherwise violate or conflict with any covenants, conditions, restrictions or other Contractual Obligations, including the requirements of any applicable Encumbrances thereto. Except as set forth on Schedule 2.12, neither the Company nor any Predecessor (a) is in violation of any Legal Requirement relating to Real Property, including setback requirements, zoning restrictions and ordinances, building, life, access, safety, health and fire codes and ordinances affecting the Real Property, or (b) has received notice of any eminent domain, condemnation or similar proceeding pending or, to the Seller's Knowledge, threatened, or any Government Order relating thereto.

The Real Property is not located within any flood plain or subject to any similar type of restriction for which any Permits necessary to the use thereof have not been obtained.

2.13 <u>Equipment</u>. All of the fixtures and other improvements to the Real Property included in the Assets (including any Facilities) and all of the tangible personal property other than inventory included in the Assets (the "<u>Equipment</u>") (a) are adequate and suitable for their present and intended uses, (b) are in good working order, operating condition and state of repair, reasonable wear and tear excepted, (c) have no defects (whether patent or latent) and (d) have been maintained in accordance with normal industry practice.

2.14 Intellectual Property.

- 2.14.1 Schedule 2.14 identifies all Intellectual Property owned or used by the Company in the Business, including, without limitation, all Intellectual Property with respect to Company Technology, and lists: (a) all registered Intellectual Property which has been issued to or is otherwise owned by the Company or that relates to or is otherwise used by the Company in the Business, and (b) each Contractual Obligation which the Company or the Seller has granted or has been granted rights to any Intellectual Property or to which any of them is otherwise bound to any third party. The Company has no pending application for registration which the Company has made or otherwise owns with respect to any Intellectual Property. True, accurate and complete copies of all such Contractual Obligations, as amended or otherwise modified and in effect, have been provided to the Buyer by the Seller, as well as true, accurate and complete copies of all other written documentation evidencing ownership and prosecution (if applicable) of each such item. All such Contractual Obligations are in full force and effect. To the Seller's Knowledge, there does not exist any claim, allegation, or basis for any claim or allegation, that any Intellectual Property owned or otherwise used by the Company is invalid or unenforceable or that the Company's rights with respect thereto are subject to claims or defenses that would impair or preclude enforcement of such rights, including, without limitation, misuse, laches, acquiescence, statute of limitations, abandonment or fraudulent registration. Schedule 2.14 also identifies each trade name, trade dress and unregistered trademark or service mark used by the Company or in connection with the Business or the Company Technology.
- 2.14.2 The Company is the sole owner of all rights, title and interest in and to all Intellectual Property with respect to, or has the right to use as specified in Schedule 2.14, all the Company Technology free and clear of any Encumbrances. The Company has the right to use all other Technology and Intellectual Property used in the Business as currently conducted and as currently contemplated to be conducted in the future. The Intellectual Property identified on Schedule 2.14 constitutes all Intellectual Property that is used in the Business and all Intellectual Property necessary for the conduct of the Business as currently conducted and as currently contemplated to be conducted in the future.
- 2.14.3 Neither the Company nor any Predecessor (a) has, nor has the conduct of the Business, interfered with, infringed upon, misappropriated, diluted or otherwise violated or come into conflict with any Intellectual Property rights of third parties or (b) has received any charge, complaint, claim, demand, or notice alleging any such interference, infringement, misappropriation, dilution or other violation or conflict (including, without limitation, any claim that it must obtain a license or refrain from using any Intellectual Property rights of any third

party in connection with the conduct of the Business or the use of the Company Technology, or, with respect to any other Technology used in the Business, that it must obtain a license that it does not already possess or refrain from using same). To the Seller's Knowledge, there does not exist any basis for any such claim or allegation. To the Seller's Knowledge, no third party has interfered with, infringed upon, misappropriated, diluted or otherwise violated or come into conflict with any Company Technology or any of the Company's other Intellectual Property. No claim or legal proceeding involving any Intellectual Property Right by or against the Company is pending or, to the Seller's Knowledge, has been threatened. Except as otherwise specified in Schedule 2.14, the Company is not bound by any Contractual Obligation containing any covenant or other provision relating to Intellectual Property that in any way limits or restricts the ability of the Company to use, exploit, assert, or enforce any of its Intellectual Property or conduct the Business anywhere in the world.

- 2.14.4 Except for that Software described in <u>Schedule 2.14</u>, the Company Software constitutes all the Software necessary to conduct the Business as currently conducted by the Company and as currently contemplated to be conducted in the future.
 - 2.14.5 With respect to each item of the Company Technology or other Technology or Intellectual Property used in the Business:
 - (a) the Company possesses all right, title, and interest in and to such item, or has the right to use same, free and clear of any Encumbrance and its use thereof;
 - (b) such item is not subject to any outstanding Government Order, and no Action is pending or threatened, which challenges the legality, validity, enforceability, use or ownership of such item; and
 - (c) the Company has not agreed to and does not otherwise have a Contractual Obligation to indemnify any Person for or against any interference, infringement, misappropriation or other conflict with respect to such item.
- 2.14.6 Schedule 2.14 identifies each item of Technology that any Person other than the Company owns and that is used by the Company or in connection with the Business pursuant to any license, sublicense or other Contractual Obligation (the "Licenses"). Except as disclosed on Schedule 2.14, there are no royalties for the use of any such Technology. The Company has made available to the Buyer true, accurate and complete copies of all of the Licenses, in each case, as amended or otherwise modified and in effect, and each of such Licenses is in full force and effect, is valid and enforceable in accordance with its terms and no party thereto is in breach of any of the terms thereof. With respect to each such item identified on Schedule 2.14: (a) such item is not subject to any outstanding Government Order, and no Action is pending or threatened which challenges the legality, validity or enforceability of such item and (b) neither the Seller nor the Company has granted any sublicense or similar right with respect to any License covering such item.
 - 2.15 Legal Compliance; Illegal Payments; Permits.

- 2.15.1 Compliance. The Company is not in breach or violation of, or default under, and since December 31, 2003 has not been in breach or violation of, or default under:
 - (a) its Organizational Documents nor, to the Seller's Knowledge, is there a basis which could constitute such a breach, violation or default; or
 - (b) any material Legal Requirement nor, to the Seller's Knowledge, is there a basis which could constitute such a breach, violation or default, except for breaches, violations or defaults which have not had, will not have and are not reasonably likely to have, a Material Adverse Effect.
- 2.15.2 Anti-Corruption. To the Seller's Knowledge, no agent, affiliate, employee or other Person associated with or acting on behalf of the Company directly or indirectly, has offered to pay or provide or has paid or provided anything of value in the form of any unlawful contribution, gift, entertainment or other unlawful expense to any Person for the purpose of gaining or retaining business or obtaining any unfair advantage, nor made any bribe, rebate, payoff, influence payment, kickback or other similar unlawful payment.
- 2.15.3 Permits. The Company has been duly granted all Permits under all Legal Requirements necessary for the conduct of the Business. Schedule 2.15.3 describes each Permit affecting, or relating to, the Assets or the Business together with the Governmental Authority or other Person responsible for issuing such Permit. Except as disclosed on Schedule 2.15.3, (a) the Permits are valid and in full force and effect, (b) the Company is not in breach or violation of, or default under, any such Permit, and, to the Seller's Knowledge, no basis exists which, with notice or lapse of time or both, would constitute any such breach, violation or default and (c) the Permits will continue to be valid and in full force and effect, on identical terms following the consummation of the Contemplated Transactions.
 - 2.15.4 Additional Compliance Representations.
 - (a) No petition under the federal bankruptcy laws or any state insolvency law has been filed by or against, or a receiver, fiscal agent or similar officer appointed by a court for the business or property of, (i) the Seller, or the Company, (ii) any partnership in which the Seller, or the Company was a general partner at or within two years before the date of this Agreement, or (iii) any corporation or business association of which the Seller was an executive officer at or within two years before the date of this Agreement.
 - (b) Neither the Seller, within the last five (5) years, nor the Company has been convicted in a criminal proceeding (excluding traffic violations and other minor offenses) nor is the Seller or the Company a named subject or target of a pending criminal proceeding or investigation.
 - (c) Neither the Seller nor the Company has been the subject of any order, judgment or decree, not subsequently reversed, suspended or vacated, of any court of competent jurisdiction, permanently or temporarily enjoining the Seller or the Company from, or otherwise limiting the Seller's or the Company's involvement in, any of the following activities:

- (i) acting as a futures commission merchant, introducing broker, commodity trading advisor, commodity pool operator, floor broker, leverage transaction merchant, any other Person regulated by the Commodity Futures Trading Commission, or as an associated Person of any of the foregoing, or as an investment adviser, underwriter, broker or dealer in securities, or as an affiliated Person, director or employee of any investment company, bank, savings and loan association or insurance company, or engaging in or continuing any conduct or practice in connection with such activity;
 - (ii) engaging in any type of business practice; or
- (iii) engaging in any activity in connection with the purchase or sale of any security or commodity, or in connection with any violation of federal or state securities laws or federal commodities laws.
- (d) Neither the Seller nor the Company has been the subject of any order, judgment or decree, not subsequently reversed, suspended or vacated, of any federal or state authority barring, suspending or otherwise limiting for more than sixty (60) days the Seller's or the Company's right to engage in any activity described in subparagraph (c) above, or to be associated with Persons engaged in any such activity.
- (e) Neither the Seller nor the Company has been found by a court of competent jurisdiction in a civil action or by the Securities and Exchange Commission ("SEC") to have violated any federal or state securities law, and the judgment in such civil action or finding by the SEC has not been subsequently reversed, suspended or vacated.
- (f) Neither the Seller nor the Company has been found by a court of competent jurisdiction in a civil action or by the Commodities Futures Trading Commission to have violated any federal commodities law, and the judgment in such civil action or finding by the Commodity Futures Trading Commission has not been subsequently reversed, suspended or vacated.
- (g) Neither the Seller nor the Company has been involved in any of the following matters: (a) making any political contributions that would be illegal under any law; (b) the disbursement or receipt of funds of the Company outside the normal system of accountability; (c) payments, whether direct or indirect, to or from foreign or domestic governments, officials, employees or agents for purposes other than the satisfaction of lawful obligations, or any transaction which has as its intended effect the transfer of assets of the Seller, or the Company for the purpose of effecting such payment; (d) the improper or inaccurate recording of payments and receipts on the books of the Company; or (e) any other matters of similar nature involving disbursements of assets of the Company.

2.16 <u>Inventories</u>. The Company maintains sufficient Inventory in the Ordinary Course of Business to conduct the Business consistent with past practices. The Inventory of the Company reflected on the Most Recent Balance Sheet and in the records and books of account of the Company since the Most Recent Balance Sheet Date are of a quality and a quantity usable or saleable, as the case may be, in the Ordinary Course of Business. The Company has sole custody and control of and maintains adequate insurance coverage on all materials, supplies, parts or other assets delivered to the Company by or on behalf of its customers for use in connection with projects the Company is undertaking for such customers (the "<u>Customer Assets</u>"); none of the Customer Assets have been damaged, lost, stolen, or otherwise suffered a material diminution in value from the time of receipt by the Company; and the Company has not received notice of any claim, loss, or damage related to the Customer Assets.

2.17 Employee Benefit Plans.

- (a) Schedule 2.17(a) contains a true and complete list of each "employee benefit plan" (within the meaning of Section 3(3) of ERISA, including, without limitation, multiemployer plans within the meaning of Section 3(37) of ERISA), and all stock purchase, stock option, severance, employment, change-in-control, fringe benefit, collective bargaining, bonus, incentive, deferred compensation, employee loan and all other employee benefit plans, agreements, programs, policies or other arrangements, whether or not subject to ERISA (including any funding mechanism therefor now in effect or required in the future as a result of the Contemplated Transactions or otherwise), whether formal or informal, oral or written, legally binding or not, under which (i) any current or former employee, director or consultant of the Company (the "Company Employees") has any present or future right to benefits and which are contributed to, sponsored by or maintained by the Company or (ii) the Company has had, has or may have any actual or contingent present or future liability or obligation. All such plans, agreements, programs, policies and arrangements shall be collectively referred to as the "Employee Plans".
- (b) With respect to each Employee Plan, the Seller has provided to the Buyer a current, accurate and complete copy (or, to the extent no such copy exists, an accurate description) thereof and, to the extent applicable: (i) any contracts or agreements relating to any Employee Plan, including, without limitation all trust agreements, insurance or annuity contracts, investment management agreements, record keeping agreements and other material documents or instruments related thereto; (ii) the most recent determination letter, if applicable; (iii) any summary plan description and other written communications (or a description of any oral communications) by the Company to the Company Employees concerning the extent of the benefits provided under a Employee Plan; (iv) a summary of any proposed amendments or changes anticipated to be made to the Employee Plans at any time within the twelve months immediately following the date hereof; and (v) for the three most recent years (A) the Form 5500 and attached schedules, (B) reviewed financial statements, (C) accuarial valuation reports and (D) any non-discrimination testing results.

- (c) Each Employee Plan has been established and administered in accordance with its terms, and in compliance with the applicable provisions of ERISA, the Code and other applicable laws, rules and regulations; (ii) each Employee Plan which is intended to be qualified within the meaning of Section 401(a) of the Code is so qualified and has received a favorable determination letter as to its qualification, and nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause the loss of such qualification; (iii) no event has occurred and no condition exists that would subject the Company, either directly or by reason of its affiliation with any member of its "Controlled Group" (defined as any organization which is a member of a controlled group of organizations within the meaning of Sections 414(b), (c), (m) or (o) of the Code), to any tax, fine, lien, penalty or other liability imposed by ERISA, the Code or other applicable laws, rules and regulations; (iv) for each Employee Plan with respect to which a Form 5500 has been filed, no material change has occurred with respect to the matters covered by the most recent Form since the date thereof; (v) no "reportable event" (as such term is defined in Section 4043 of the Code) that could reasonably be expected to result in any liability; (vi) no nonexempt "prohibited transaction" (as such term is defined in Section 406 of ERISA and Section 4975 of the Code) or "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA and Section 412 of the Code (whether or not waived)) has occurred with respect to any Employee Plan; (vii) there is no present intention that any Employee Plan be materially amended, suspended or terminated, or otherwise modified to adversely change benefits (or the levels thereof) under any Employee Plan at any time within the twelve months immediately following the date hereof; (viii) no Employee Plan is a split-dollar life insurance program or otherwise provides for loans to executive officers (within the meaning of the Sarbanes-Oxley Act of 2002); and (ix) the Company has not incurred any current or projected liability in respect of post-employment or post-retirement health, medical or life insurance benefits for current, former or retired employees of the Company, except as required to avoid an excise tax under Section 4980B of the Code or otherwise except as may be required pursuant to any other applicable law.
- (d) No Employee Plan is an "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) subject to Title IV of ERISA, and the Company has no obligation to contribute, and has not incurred any actual or contingent liability or obligation (including, without limitation, any obligation to make any contribution) to or in respect of any such plan. No Employee Plan is a "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA), and neither the Company nor any member of its Controlled Group has at any time sponsored or contributed to, or has or had any actual or contingent liability or obligation to or in respect of, any multiemployer plan.
- (e) With respect to each Employee Plan, (i) no actions, suits or claims (other than routine claims for benefits in the ordinary course) are pending or to the Seller's Knowledge threatened; (ii) to the Seller's Knowledge no facts or circumstances exist that could give rise to any such actions, suits or claims; (iii)

no written or oral communication has been received from the Pension Benefit Guaranty Corporation (the "<u>PBGC</u>") in respect of any Employee Plan subject to Title IV of ERISA concerning the funded status of any such plan or any transfer of assets and liabilities from any such plan in connection with the transactions contemplated herein; and (iv) no administrative investigation, audit or other administrative proceeding by the Department of Labor, the PBGC, the Internal Revenue Service (the "<u>IRS</u>") or other governmental agencies are pending, in progress or to the Seller's Knowledge threatened (including, without limitation, any routine requests for information from the PBGC).

- (f) No Employee Plan or Legal Requirement exists that, as a result of the execution of this Agreement, shareholder approval of this Agreement, or the Contemplated Transactions (whether alone or in connection with any subsequent event(s)), could (i) result in severance pay, termination indemnity or any similar payment or any increase in severance pay, termination indemnity or any similar payment, (ii) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Employee Plans, (iii) limit or restrict the right of the Company to merge, amend or terminate any of the Employee Plans, (iv) cause the Company to record additional compensation expense on its income statement with respect to any outstanding stock option or other equity-based award, or (v) result in payments under any of the Employee Plans which would not be deductible under Section 280G of the Code.
- (g) There has been no amendment to, written interpretation of or announcement (whether or not written) by the Company relating to, or any change in employee participation or coverage under, any Employee Plan that would increase the expense of maintaining such Employee Plan above the level of the expense incurred in respect thereof for the most recent fiscal year ended prior to the date hereof.
- (h) No compensation under any Employee Plan subject to Section 409A of the Code is or has been required to be includible in the gross income of any participant or beneficiary by reason of Section 409A(a)(i)(A) of the Code or is or has been subject to any additional tax under Section 409A(a)(i)(B) of the Code, and no amounts are or have been includible in any participants or beneficiaries by reason of Section 409A(b) of the Code.
- 2.18 Environmental Matters. Except as set forth in Schedule 2.18, the Company and its Predecessors are, and have been, in compliance with all Environmental Laws, (b) there has been no release or threatened release of any pollutant, petroleum or any fraction thereof, contaminant or toxic or hazardous material (including toxic mold), substance or waste (each a "Hazardous Substance") on, upon, into or from any site currently or, to the Seller's Knowledge, heretofore owned, leased or otherwise used by the Company or a Predecessor, (c) there have been no Hazardous Substances generated by the Company or a Predecessor that have been disposed of or come to rest at any site that has been included in any published U.S. federal, state

or local "superfund" site list or any other similar list of hazardous or toxic waste sites published by any Governmental Authority in the United States or in any other location throughout the world, (d) there are no underground storage tanks located on, no PCBs (polychlorinated biphenyls) or PCB-containing Equipment used or stored on, and no hazardous waste as defined by the Resource Conservation and Recovery Act stored on, any site owned or operated by the Company or a Predecessor, except for the storage of hazardous waste in compliance with Environmental Laws and (e) the Seller has made available to the Buyer true, accurate and complete copies of all environmental records, reports, notifications, certificates of need, permits, pending permit applications, correspondence, engineering studies, and environmental studies or assessments, in each case as amended and in effect. Set forth on Schedule 2.18 is a list of all locations at which the Company has stored or currently stores petroleum or any fraction thereof, petroleum products, or Hazardous Substances, with each such location being designated as either "Company Owned" or "Third Party Owned." There are no pending environmental investigations or enforcement actions being conducted with respect to the Company, the Business, or any properties at which the Business is conducted, by any Governmental Authority and no pending claims pertaining to environmental matters brought by third parties.

2.19 Contracts.

- 2.19.1 Contracts. Except as disclosed on Schedule 2.19, the Company is not bound by or a party to:
- (a) any Contractual Obligation (or group of related Contractual Obligations) for the purchase or sale of inventory, raw materials, commodities, supplies, goods, products, equipment or other personal property, or for the furnishing or receipt of services, in each case, the performance of which will extend over a period of more than one year or which provides for either an aggregate or annual payments to or by the Company in excess of twenty-five thousand dollars (\$100,000);
- (b) (i) any capital lease or (ii) any other lease or other Contractual Obligation relating to the Equipment providing for annual rental payments in excess of twenty-five thousand dollars (\$25,000), under which any Equipment is held or used by the Company;
- (c) any Contractual Obligation, other than Real Property Leases or leases relating to the Equipment, relating to the lease or license of any Asset, including Technology and Intellectual Property (and including all customer license and maintenance agreements) that is not included on <u>Schedule 2.14</u>;
- (d) any Contractual Obligation relating to the acquisition or disposition of (i) any business of the Company (whether by merger, consolidation or other business combination, sale of securities, sale of assets or otherwise) or (ii) any asset other than in the Ordinary Course of Business;
- (e) any Contractual Obligation under which the Company is, or may become, obligated to pay any amount in respect of indemnification obligations, purchase price adjustment or otherwise in connection with any (i) acquisition or disposition of assets or securities (other than the sale of inventory in the Ordinary Course of Business), (ii) merger, consolidation or other business combination or (iii) series or group of related transactions or events of the type specified in clauses (i) and (ii) above.

- (f) any Contractual Obligation concerning or consisting of a partnership, limited liability company or joint venture agreement or any other relationship involving the sharing of profits, losses or costs;
- (g) any Contractual Obligation (or group of related Contractual Obligations) (i) under which the Company has created, incurred, assumed or guaranteed any Debt in excess of twenty-five thousand dollars (\$25,000) or (ii) under which the Company has permitted any Asset to become subject to an Encumbrance:
 - (h) any Contractual Obligation under which any other Person has guaranteed any Debt of the Company;
- (i) any Contractual Obligation to purchase goods or services exclusively from a given Person or Persons or purchase a minimum amount of goods or services from a given Person or Persons, or all or a portion of the supply of certain goods or services utilized by the Company from a given Person or Persons;
 - (j) any Contractual Obligation relating to non-competition (whether the Company is subject to or the beneficiary of such obligations);
- (k) any Contractual Obligation under which the Company is, or may become, obligated to incur any severance pay or special Compensation obligations which would become payable by reason of, this Agreement or the Contemplated Transactions;
- (l) any Contractual Obligation under which the Company is, or may, have any Liability to any investment bank, broker, financial advisor, finder or other similar Person (including an obligation to pay any legal, accounting, brokerage, finder's, or similar fees or expenses in connection with this agreement or the Contemplated Transactions);
- (m) any profit sharing, stock option, stock purchase, stock appreciation, deferred compensation, severance, or other plan or arrangement for the benefit of the Company's current or former directors, officers, and employees;
- (n) any Contractual Obligation providing for the employment or consultancy (including on an independent contractor basis) with an individual (or in the case of a consultant or independent contractor, an entity) on a full-time, part-time, consulting or other basis or otherwise providing Compensation or other benefits to any officer, director, employee or consultant (other than an Employee Plan);

- (o) any agency, dealer, distributor, sales representative, marketing, handler, third party service provider, or other similar agreement;
- (p) any Contractual Obligation under which the Company has advanced or loaned an amount to any of its Affiliates or employees other than in the Ordinary Course of Business; or
- (q) any other Contractual Obligation (or group of related Contractual Obligations) the performance of which involves consideration in excess of twenty-five thousand dollars (\$25,000) over the life of such Contractual Obligation.

The Seller has delivered to the Buyer true, accurate and complete copies of each written Contractual Obligation listed on <u>Schedule 2.19</u>, in each case, as amended or otherwise modified and in effect.

- 2.19.2 Enforceability, etc. To the Seller's Knowledge, each Contractual Obligation required to be disclosed on Schedule 2.9 (Debt), Schedule 2.12 (Real Property Leases), Schedule 2.14 (Intellectual Property), Schedule 2.17(a) (Employee Benefit Plans), Schedule 2.19 (Contracts), Schedule 2.21 (Customers and Suppliers) or Schedule 2.26 (Insurance) (each, a "Disclosed Contract") is Enforceable against each party to such Contractual Obligation, and is in full force and effect, and will continue to be so Enforceable and in full force and effect on identical terms following the consummation of the Contemplated Transactions.
- 2.19.3 Breach, etc. Neither the Company nor, to the Seller's Knowledge, any other party to any Disclosed Contract is in breach or violation of, or default under, or has repudiated any provision of, any Disclosed Contract (including all performance bonds, warranty obligations or otherwise). The Company has not received notice from any other party to any Disclosed Contract or to the Seller's Knowledge has any reason to believe that such party intends to terminate such Disclosed Contract or alter in any way the relationship of the parties under such Disclosed Contract.
 - 2.19.4 Bids. Schedule 2.19.4 sets forth a complete and accurate list of each outstanding bid or proposal for business submitted by the Company.
- 2.20 <u>Affiliate Transactions</u>. Except for the matters disclosed on <u>Schedule 2.20</u>, neither the Seller nor any Affiliate of the Seller is an officer, director, employee, consultant, competitor, creditor, debtor, customer, distributor, supplier or vendor of, or is a party to any Contractual Obligation with, the Company. Except as disclosed on <u>Schedule 2.20</u>, neither the Seller nor any Affiliate of the Seller owns any Asset used in, or necessary to, the Business.
- 2.21 <u>Customer and Supplier</u>. <u>Schedule 2.21</u> sets forth a complete and accurate list of (a) the ten largest customers of the Company (measured by aggregate billings) during the fiscal year ended January 31, 2008, indicating the existing Contractual Obligations with each such customer by product or service provided and (b) the ten largest suppliers of materials, products

or services to the Company (measured by the aggregate amount purchased by the Company) during the fiscal year ended January 31, 2008, indicating the Contractual Obligations for continued supply from each such supplier. The relationships of the Company with the customers and the suppliers required to be listed on Schedule 2.21 are good commercial working relationships and none of such customers or the suppliers has canceled, terminated or otherwise altered (including any reduction in the rate or amount of sales or purchases, material increase in the prices charged or paid, or change to the supply or credit terms, as the case may be) or notified the Company of any intention to do any of the foregoing or otherwise threatened in writing to cancel, terminate or alter (including any reduction in the rate or amount of sales or purchases, material increase in the prices charged or paid, or change to the supply or credit terms, as the case may be) its relationship with the Company. As of the Closing Date, to the Seller's Knowledge, there is no reason to believe that there could be any change in the relationships of the Company with any of its customers or suppliers as a result of the Contemplated Transactions. The Seller has no notice of any facts or circumstances that has or could result in a change in the relationship that Company has with any of its customers and suppliers, other than facts or circumstances that relate primarily to economic conditions in general and the economic performance of the entire industry in which the Company is engaged.

2.22 <u>Customer Warranties</u>. There have been no pending, nor to the Seller's Knowledge, threatened, claims under or pursuant to any warranty, whether expressed or implied, on the products or services sold prior to the Closing Date by the Company that are not disclosed or referred to in the Reviewed Financials and that are not fully reserved against in accordance with GAAP. All of the services rendered by the Company (whether directly or indirectly through independent contractors) have been performed in conformity with all express warranties and, in all material respects, with all applicable contractual commitments, and the Company does not have nor shall it have any Liability for replacement or repair or for other damages relating to or arising from any such services, except for amounts incurred in the Ordinary Course of Business which are immaterial individually and in the aggregate and not required by GAAP to be disclosed in the Reviewed Financials. Except as set forth on <u>Schedule 2.22</u>, none of the Company's Contractual Obligations with any customer contains any unusual warranty provisions that would impose material liability on the Company. Set forth on <u>Schedule 2.22</u> is the aggregate amount of warranty claims incurred by the Company during each of the last three completed fiscal years. To the Seller's Knowledge, there is no reason to expect an increase in the amount of warranty claims in the future.

2.23 <u>Capital Expenditures and Investments</u>. The Company's outstanding Contractual Obligations and budget for capital expenditures and Investments are set forth on <u>Schedule 2.23</u>, which includes a schedule of all monies disbursed on account of capital expenditures and Investments made by the Company since the Most Recent Balance Sheet Date.

2.24 Employees.

2.24.1 All present employees of the Company are listed on <u>Schedule 2.24</u>. <u>Schedule 2.24</u> also contains a list of the ranch employees whose employment with the Company will be terminated at or prior to the Closing. To the Seller's knowledge, no present or past employee of the Company is bound by a non-competition agreement. Except as set forth

Schedule 2.24, the Company is not a party to any written or oral agreement with any employee or former employee.

2.24.2 Except as disclosed on Schedule 2.24, there are no labor troubles (including any work slowdown, lockout, stoppage, picketing or strike) pending, or to the Seller's Knowledge, threatened between the Company, on the one hand, and its employees, on the other hand. Except as disclosed on Schedule 2.24, (a) no employee of the Company is represented by a labor union, (b) the Company is not a party to, or otherwise subject to, any collective bargaining agreement or other labor union contract, (c) during the past five years there have been no strikes, slowdowns, work stoppages, disputes, lockouts, or to the Seller's Knowledge threats thereof, by or with respect to any employees of the Company, (d) no petition has been filed or proceedings instituted by an employee or group of employees of the Company with any labor relations board seeking recognition of a bargaining representative and (e) there is no organizational effort currently being made or threatened by, or on behalf of, any labor union to organize employees of the Company and no demand for recognition of employees of the Company has been made by, or on behalf of, any labor union. The Company is not a party to, or otherwise bound by, any consent decree with, or citation or other Order by, any Governmental Authority relating to employees or employment practices. The Company is in compliance with applicable Legal Requirements, Contractual Obligations, and policies relating to employment, employment practices, wages, hours, and terms and conditions of employment, including, but not limited to, the obligations of the Fair Labor Standards Act ("FLSA") and the Worker Adjustment and Retraining Notification Act of 1988 ("WARN"), and all other notification and bargaining obligations arising under any collective bargaining agreement, by Legal Requirement or otherwise. The Company has not effectuated a "plant closing" or "mass layoff" as those terms are defined in WARN, affecting in whole or in part any site of employment, facility, operating unit or employee of the Company without complying with all provisions of WARN, or implemented any early retirement, separation or window program within the past five years, nor has the Company planned or announced any such action or program for the future. No executive officer's or other key employee's employment with the Company has been terminated for any reason nor has any such officer or employee notified the Company of his or her intention to resign or retire since the Most Recent Balance Sheet Date.

2.24.3 The Company is not delinquent in payments to any of its employees or consultants for any wages, salaries, overtime pay, commissions, bonuses, benefits or other compensation for any services or otherwise arising under any policy, practice, Contractual Obligation, plan, program or Legal Requirement. None of the Company's employment policies or practices is currently being audited or investigated by any Governmental Authority. There is no pending or, to the Seller's Knowledge, threatened Action, unfair labor practice charge, or other charge or inquiry against the Company brought by or on behalf of any employee, prospective employee, former employee, retiree, labor organization or other representative of the Company's employees, or other individual or any Governmental Authority with respect to employment practices brought by or before any Governmental Authority.

2.24.4 To the Seller's Knowledge, no employee, officer, contractor or consultant of the Company is obligated under any applicable law or under any Contractual Obligation of any nature, or is subject to any judgment, decree or order of any court or administrative agency, that would interfere with the use of such employee's, officer's, contractor's or consultant's best

efforts to promote the interests of the Company or that would conflict with the Business. To the Seller's Knowledge, the conduct of the Business has not and will not, and the consummation of the Contemplated Transactions will not, conflict with or result in a breach of the terms, conditions or provisions, or constitute a default under any Contractual Obligation under which any employee, officer, contractor or consultant of the Company is obligated.

2.25 Litigation; Governmental Orders.

2.25.1 Litigation. Except as disclosed on Schedule 2.25, there is no Action to which the Company is a party (either as plaintiff or defendant) or to which its Assets are subject pending, or to the Seller's Knowledge, threatened. There is no Action to which the Company is a party (either as plaintiff or defendant) or to which its Assets are subject pending, or to the Seller's Knowledge, threatened, which (a) in any manner challenges or seeks the rescission of, or seeks to prevent, enjoin, alter or delay the consummation of, or otherwise relates to, this Agreement and the Contemplated Transactions, or (b) may result in any change in the current equity ownership of the Company, nor, to the Seller's Knowledge, is there any basis for any of the foregoing. Except as disclosed on Schedule 2.25, there is no Action which the Company presently intends to initiate. The Seller has no notice of any facts or circumstances that could result in any Action to which the Company is a party (either as plaintiff or defendant) or to which its Assets are subject. All oral and written statements, information and documentation provided to the Buyer with respect to the litigation disclosed on Schedule 2.25 regarding the accidental death on Jose Arteaga are true, complete and accurate and do not omit to state any material fact necessary in order to make such statements, information and documentation not misleading.

2.25.2 Governmental Orders. Except as disclosed on <u>Schedule 2.25</u>, no Governmental Order has been issued which is applicable to, or otherwise affects, the Company or its Assets or the Business.

2.26 <u>Insurance</u>. <u>Schedule 2.26</u> sets forth a list of insurance policies, including policies by which the Company, or any of its Assets, employees, officers or directors or the Business has been insured since December 31, 2005 (the "<u>Liability Policies</u>") and, with respect to such Liability Policies under which the Company, or any of its Assets, employees, officers or directors or the Business is currently insured, their respective expiration dates. The list includes for each Liability Policy the type of policy, form of coverage, policy number and name of insurer. The Seller has made available to the Buyer true, accurate and complete copies of all Liability Policies, in each case, as amended or otherwise modified and in effect. <u>Schedule 2.26</u> describes any self-insurance arrangements affecting the Company. The Company has since December 31, 2005 maintained in full force and effect with financially sound and reputable insurers insurance with respect to its Assets and the Business, in such amounts and against such losses and risks as is customarily carried by Persons engaged in the same or similar business and as is required under the terms of any applicable Real Property Leases or other Contractual Obligations. Except as disclosed on <u>Schedule 2.26</u>, no insurer (a) has questioned, denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Liability Policy or (b) has provided any notice of cancellation or any other indication and neither the Seller nor the Company has any reason to believe that any insurer plans to cancel any Liability Policy or raise the premiums or alter the coverage under any Liability Policy.

- 2.27 <u>Banking Facilities</u>. <u>Schedule 2.27</u> sets forth a true, correct and complete list of: (a) each bank, savings and loan or similar financial institution with which the Company has an account or safety deposit box or other arrangement, and any numbers or other identifying codes of such accounts, safety deposit boxes or such other arrangements maintained by the Company thereat; (b) the names of all Persons authorized to draw on any such account or to have access to any such safety deposit box facility or such other arrangement; and (c) any outstanding powers of attorney executed by or on behalf of the Company.
- 2.28 <u>Powers of Attorney</u>. Except as set forth on <u>Schedule 2.28</u>, the Company does not have any general or special powers of attorney outstanding (whether as grantor or grantee thereof).
- 2.29 <u>No Brokers</u>. The Company has no Liability of any kind to, and is not subject to any claim of, any broker, finder or agent in connection with the Contemplated Transactions other than those which will be borne by the Seller.
- 2.30 <u>Adequacy of Internal Controls</u>. The Company has established and maintains effective internal control over financial reporting, which internal control over financial reporting includes those policies and procedures that (a) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the Company; (b) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP; and (c) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's assets that could affect the Company's financial statements.
- 2.31 <u>Disclosure</u>. The representations and warranties contained in this <u>Section 2</u> and in the documents, instruments and certificates and all diligence material and information, furnished by the Company or the Seller to the Buyer and its Representatives pursuant to this Agreement do not contain any untrue statement of fact or omit to state any material fact necessary in order to make the statements and information contained therein not misleading. To the Seller's Knowledge there is no material fact (other than any facts or circumstances that relate primarily to economic conditions in general and the economic performance of the entire industry in which the Company is engaged) which has not been disclosed to the Buyer which materially and adversely affects the Seller's ability to consummate the transactions contemplated hereby.

2.32 Securities Law Matters.

- 2.32.1 The MasTec common stock that may be acquired by the Seller pursuant to this Agreement is being acquired for the Seller's own account and not with a view to, or intention of, distribution thereof in violation of the 1933 Act, or any applicable state securities laws, and the MasTec common stock will not be disposed of by the Seller in contravention of the 1933 Act or any applicable state securities laws.
- 2.32.2 The Seller is an "accredited investor" as defined in Rule 501(a) under the 1933 Act, is sophisticated in financial matters and is able to evaluate the risks and benefits of the investment in the MasTec common stock and make an informed investment decision.

- 2.32.3 The Seller has had an opportunity to ask questions and receive answers concerning MasTec and the MasTec common stock and has had full access to such other information concerning MasTec and the MasTec common stock as the Seller has requested or which has been filed by MasTec with the SEC.
- 2.32.4 The Seller has discussed with and relied upon the advice of its independent legal counsel, tax and financial advisors with regard to the meaning and legal consequences of the Seller's representations and warranties contained herein and the considerations involved in making an investment in MasTec common stock, and the Seller understands that MasTec is relying on the information set forth herein.
- 2.33 <u>Company Operations</u>. Since the Effective Date, the Seller has operated the Company in the Ordinary Course of Business consistent with past practices and has not made any material decisions regarding the operation of the Business and has not incurred any material Liabilities outside of the Ordinary Course of Business without the Buyer's prior knowledge and written consent.

3. REPRESENTATIONS AND WARRANTIES OF THE BUYER.

In order to induce the Seller to enter into and perform this Agreement and to consummate the Contemplated Transactions, the Buyer hereby represents and warrants to the Seller as follows:

- 3.1 <u>Organization</u>. The Buyer is duly organized, validly existing and in good standing under the laws of the State of Florida and has the corporate power and authority to own its properties and to carry on its business as it is now being conducted.
- 3.2 <u>Power and Authorization</u>. The execution, delivery and performance by the Buyer of this Agreement and the consummation of the Contemplated Transactions are within the power and authority of the Buyer and have been duly authorized by all necessary action on the part of the Buyer and no other corporate or other action on the part of the Buyer or any other Person is necessary to authorize the execution and delivery of this Agreement by the Buyer or the consummation by the Buyer of the transactions contemplated by this Agreement. This Agreement (a) has been duly executed and delivered by the Buyer and (b) is a legal, valid and binding obligation of the Buyer, Enforceable against the Buyer in accordance with its terms.
- 3.3 <u>Authorization of Governmental Authorities</u>. No action by (including any authorization, consent or approval), or in respect of, or filing with, any Governmental Authority is required for, or in connection with, the valid and lawful (a) authorization, execution, delivery and performance by the Buyer of this Agreement or (b) the consummation of the Contemplated Transactions by the Buyer.
- 3.4 <u>Noncontravention</u>. Neither the execution, delivery and performance by the Buyer of this Agreement nor the consummation of the Contemplated Transactions will:
 - (a) violate any provision of any Legal Requirement applicable to the Buyer;

- (b) result in a breach or violation of, or default under, or give rise to a right for any third-party to terminate or any prepayment penalty under any Contractual Obligation of the Buyer;
- (c) require any action by (including any authorization, consent or approval) or in respect of (including notice to), any Person under any Contractual Obligation; or
 - (d) result in a breach or violation of, or default under, the Buyer's Organizational Documents.
- 3.5 No Brokers. The Buyer has no Liability of any kind to any broker, finder or agent with respect to the Contemplated Transactions for which the Seller could be liable.
- 3.6 <u>Securities Law Compliance</u>. MasTec has made all filings with the SEC that it is required to make under the 1933 Act and the Exchange Act (collectively the "<u>Public Reports</u>"). No material adverse change has occurred in the financial condition or business operations or opportunities of MasTec, or any of its Affiliates, since the filing of MasTec's most recent Public Reports.

4. COVENANTS.

4.1 <u>Release</u>. Effective as of the Effective Date, the Seller hereby releases, remises and forever discharges any and all rights and claims that it has had, now has or might now have against the Company, including with respect to the Excluded Assets and the warranty deeds transferring the real property included therein, except for rights and claims arising from or in connection with this Agreement.

4.2 Confidentiality.

4.2.1 The Seller acknowledges that the success of the Company after the Effective Date depends upon the continued preservation of the confidentiality of certain information possessed by the Seller, that the preservation of the confidentiality of such information by the Seller is an essential premise of the bargain between the Seller and the Buyer, and that the Buyer would be unwilling to enter into this Agreement in the absence of this Section 4.2.1. Accordingly, the Seller hereby agrees with the Buyer that the Seller and its Representatives will not, and that the Seller will cause its Affiliates not to, at any time on or after the Effective Date, directly or indirectly, without the prior written consent of the Buyer, disclose or use, any confidential or proprietary information involving or relating to the Business or the Company, including without limitation (1) customer and supplier information, including lists of names and addresses of customers and suppliers of the Company or its Affiliates, (2) business plans and strategies, compensation plans, compensation information, sales plans and strategies, pricing and other terms applicable to transactions between existing and prospective customers, suppliers or business associates, (3) market research and databases, sources of leads and methods of obtaining new business, and methods of purchasing, marketing, selling, performing and pricing products and services employed by the Company, (4) information concerning the configuration and architecture, technical data, networks, methods, practices, standards and capacities of the Company's information systems, Company Software and Company

Technology, and (5) information identified as confidential and/or proprietary in internal documents of the Company; provided, however, that the information subject to the foregoing provisions of this sentence will not include any information generally available to, or known by, the public (other than as a result of disclosure in violation hereof); and provided, further, that the provisions of this Section 4.2.1 will not prohibit any retention of copies of records or disclosure (a) required by any applicable Legal Requirement so long as reasonable prior notice is given to the Buyer of such disclosure and a reasonable opportunity is afforded to the Buyer to contest the same or (b) made in connection with the enforcement of any right or remedy relating to, or the performance of any obligation arising under, this Agreement or the Contemplated Transactions. The Seller agrees that it will be responsible for any breach or violation of the provisions of this Section 4.2.1 by any of its Representatives.

- 4.2.2 Notwithstanding the foregoing, the Seller, and each of its Representatives may disclose to any and all Persons, without limitation of any kind, the tax treatment and tax structure of the Contemplated Transactions and all materials of any kind (including opinions or other tax analyses) that are provided to the Seller relating to such tax treatment and tax structure.
- 4.3 <u>Publicity</u>. No public announcement or disclosure will be made by any party with respect to the subject matter of this Agreement or the Contemplated Transactions without the prior written consent of the Buyer and the Seller; provided, however, that the provisions of this <u>Section 4.3</u> will not prohibit (a) any disclosure required by any applicable Legal Requirements, including any disclosure necessary or desirable to provide proper disclosure under the securities laws or under any rules or regulations of any securities exchange on which the securities of such party may be listed or traded or (b) any disclosure made in connection with the enforcement of any right or remedy relating to, or the performance of any obligation arising under, this Agreement or the Contemplated Transactions.

4.4 Covenant Against Competition.

- 4.4.1 <u>Noncompetition</u>. The Seller on his own behalf and on behalf of each of his Affiliates, covenants that from the Effective Date through October 15, 2018 (the five year anniversary of the due date for the final Earn-Out Payment), neither he nor his Affiliates will for his or their own account or jointly with another, directly or indirectly, for or on behalf of any Person, as principal, agent or otherwise:
 - (a) (A) own, manage or control, or become engaged or serve as a shareholder, bondholder, creditor, officer, director, partner, member, employee, agent, consultant, advisor, contractor with, employer or representative of, or in any similar capacity, or (B) give financial, technical or other assistance to, otherwise invest or have a financial interest in, or in exchange for compensation otherwise associate with any Person, business or enterprise that competes directly or indirectly with the Company or the Business (a "Competitive Business") anywhere within the continental Unites States (the "Territory"); provided that the Seller and its Affiliates may passively hold up to 1% of the outstanding publicly-traded securities of a Person engaged in the Business for investment purposes only;

- (b) recruit, induce, solicit for employment, or employ, or in any manner attempt to recruit, induce, solicit or employ, any person employed by the Company as of the Effective Date, or any other person employed (or who had been so employed during the preceding twelve (12) months) by the Company, whether or not such employment is (or was) pursuant to a written contract and whether or not such employment is (or was) at will; provided, that the Seller may solicit for employment or employ those individuals set forth on Schedule 4.4.1(a) on a part-time basis so long as such solicitation or employment does not interfere or compete with or otherwise adversely affect the person's normal duties with the Company;
- (c) solicit, contact or deal with: (i) any Person that is at such time, or in the case of the Company during the five-year period preceding the Effective Date was, a customer, supplier or business associate of the Company, or (ii) any Person from whom the Company solicited business or with whom the Company discussed a potential business relationship at any time during the five-year period preceding the Effective Date, in each case, for the purpose of offering or providing services or products which are competitive with services or products provided by the Company;
- (d) cause or seek to cause to be terminated or adversely affected, or otherwise interfere with, any agreement or arrangement of any kind to which the Company is a party or from which it benefits; or
- (e) seek to interfere with or adversely affect the ongoing relationships between the Company, on the one hand, and its suppliers, customers and professional and business contacts, on the other.
- 4.4.2 Acknowledgement. The Seller acknowledges that the Company conducts business throughout the Territory. Accordingly, the Seller agrees that the Territory is reasonable to protect the legitimate business interests of the Buyer. The Seller agrees that any reduction to the Territory would seriously undermine the efficacy of this Section 4.4 and the protections that it is intended to provide. The Seller acknowledges and agrees that the covenants contained in this Section 4.4 are essential elements of this Agreement and that but for these covenants the Buyer would not have agreed to purchase the Shares. The Seller further expressly agrees and acknowledges that (a) the confidentiality, nonsolicitation and non-competition covenants contained in this Agreement (i) are reasonable and necessary for the protection of the Buyer with respect to the covenants' respective stated purposes, time, scope and geographic area, and are supported by adequate consideration; (ii) are necessary for the protection of the Buyer's legitimate business interests, including without limitation, the trade secrets, goodwill, and relationships with customers and suppliers purchased by the Buyer pursuant to this Agreement; (iii) are not unduly restrictive of any rights of the Seller; and (iv) for purposes of Texas law, including, without limitation, Texas Business & Commerce Code § 15.50(a) et seq.; are "ancillary to or part of an otherwise enforceable agreement" as of the time this Agreement was made, which includes, among others, the confidentiality and nonsolicitation covenants contained in this Agreement, each of which shall be deemed to be a separate "otherwise enforceable agreement" for purposes thereof; and (b) the Seller has sufficient employment alternatives and

sufficient assets such that the Seller does not have to compete with Company or impermissibly use Company's confidential, proprietary, trade secret information described in this Agreement during the term of this Section 4.4 in order to earn a living. The existence of any claim or cause of action against the Buyer by the Seller, whether predicated on the Buyer's breach of this Agreement or otherwise, shall not constitute a defense to the enforcement by the Buyer of the covenants contained in this Section 4.4.

- 4.4.3 <u>Injunction</u>. The Seller agrees that a violation of the terms, provisions, covered obligations, duties and conditions described in this Agreement will give rise to the Buyer's causes of action (including breach of this Agreement) against Seller for, among other relief, issuance of injunctive relief, issuance of a temporary restraining order, recovery of damages and recovery of the Buyer's costs, attorneys' fees, and expert witness fees. The Seller further agrees that it is difficult to calculate the amount and extent of any damages caused by such a violation and such a violation threatens to injure or actually does injure the Buyer and the Company's property. The Seller agrees that the Buyer shall have the non-exclusive right to apply for a temporary restraining order, a temporary or preliminary and a permanent injunction to enforce the terms, provisions, covenants, obligations, duties and conditions described in this <u>Section 4.4</u>. The Seller further agrees that the Buyer, in applying for or receiving any restraining order or injunctive relief, need only post, and shall only be required to post, a bond of not more than \$1,000.
- 4.4.4 <u>Specific Performance</u>. If the Seller commits a breach of any of the provisions of this <u>Section 4.4</u>, the Buyer shall have the right and remedy, in addition to any others that may be available, at law or in equity, to: (i) have the provisions of this <u>Section 4.4</u> specifically enforced by any court in any country having jurisdiction, through injunctive or other relief, it being acknowledged that any such breach will cause irreparable injury to the Buyer, the amount of which will be difficult to determine, and that money damages will not provide an adequate remedy to the Buyer; and (ii) collect all reasonable costs of enforcement of this Agreement, including reasonable attorneys' fees and costs, from the Seller.
- 4.4.5 <u>Severability</u>. If any covenant or restriction contained in this <u>Section 4.4</u>, or any part thereof, is hereafter construed or found to be invalid or unenforceable in part or in whole, this shall not effect the remainder of the covenants or restrictions, which shall be given full effect, without regard to the invalid portions, and any court having jurisdiction shall enforce such invalid covenant or restriction to the maximum extent possible under applicable law including, without limitation, the maximum permissible time, scope and geographic area for such covenant or restriction.
- 4.5 <u>Further Assurances</u>. From and after the Effective Date, upon the request of the Seller or the Buyer, each party will do, execute, acknowledge and deliver all such further acts, assurances, deeds, assignments, transfers, conveyances and other instruments and papers as may be reasonably required or appropriate to carry out the Contemplated Transactions. The Seller will not take any action that is designed or intended to have the effect of discouraging any lessor, licensor, supplier, distributor or customer of the Company or other Person with whom the Company has a relationship from maintaining the same relationship with the Company after the Closing as it maintained prior to the Closing.

- 4.6 <u>Cooperation Regarding Financial Statements</u>. The Seller shall use its best efforts to assist and cooperate with the efforts of the Buyer and its accountants and auditors to prepare and audit any financial statements (including but not limited to pro forma financial statements) that the Buyer will be required to prepare, file or furnish pursuant to any applicable securities laws or exchange requirements, including the 1933 Act, the Exchange Act, the rules and regulations of the SEC promulgated thereunder, and any rules or regulations of the New York Stock Exchange or other stock exchange, or under any other applicable laws.
- 4.7 <u>Corrective Action</u>. The Seller has engaged Enprotec/Hibbs & Todd (the "<u>Environmental Consultant</u>") to conduct a Phase II Environmental Site Assessment (the "<u>Phase II</u>") for the Real Property located in Giddings, Texas (the "<u>Site</u>"). The Seller acknowledges that the Phase II may identify environmental contamination at the Site, including the presence of Contaminants in the soil and groundwater at quantities above the State of Texas's acceptable protective concentration levels, as defined in Chapter 350, T.A.C. ("<u>Acceptable Levels</u>"). Furthermore, the Seller acknowledges that the Environmental Consultant may recommend Corrective Action in order to attain Acceptable Levels for the soil and groundwater and comply with Environmental Laws. Accordingly, the Seller hereby agrees to take all actions necessary or advisable to complete the Corrective Action as recommended in the Phase II or as may otherwise be required by Environmental Laws.

5. INDEMNIFICATION.

- 5.1 <u>Indemnification by the Seller</u>. Subject to the limitations set forth in this <u>Section 5</u>, the Seller will indemnify and hold harmless the Buyer and each of its Affiliates (including, following the Closing, the Company), and the Representatives and Affiliates of each of the foregoing Persons (each, a "<u>Buyer Indemnified Person</u>"), from, against and in respect of any and all Actions, Liabilities, Governmental Orders, Encumbrances, losses, damages, bonds, dues, assessments, fines, penalties, Taxes, fees, costs (including costs of investigation, defense and enforcement of this Agreement), expenses or amounts paid in settlement (in each case, including attorneys' and experts fees and expenses), whether or not involving a Third Party Claim (collectively, "<u>Losses</u>"), incurred or suffered by Buyer Indemnified Persons or any of them as a result of, arising out of or directly or indirectly relating to:
 - 5.1.1 any fraud of the Seller or any breach of, or inaccuracy in, any representation or warranty made by the Seller in Section 2;
- 5.1.2 any breach or violation of any covenant or agreement of the Seller to the extent required to be performed or complied with by the Seller (including under this <u>Section 5</u>) in or pursuant to this Agreement;
- 5.1.3 any Liabilities of the Company, the Seller, of any nature whatsoever, whether accrued, absolute, contingent or otherwise, arising from or relating to the operations or activities of the Company prior to the Effective Date, except to the extent the liabilities of the Company are reflected in the Closing Balance Sheet; or
- 5.1.4 (i) the Excluded Assets, (ii) the ranch employees set forth on <u>Schedule 2.24</u> whose employment with the Company will be terminated on or prior to the Closing Date, (iii) the matters set forth on <u>Schedule 2.18</u> and <u>Schedule 2.25</u> and (iv) the Seller's employment of the Company's employees following the Closing Date as permitted by <u>Section 4.4.1(b)</u>.

- 5.1.5 In the event that the Seller is required to indemnify any Buyer Indemnified Person for any Uncollected Receivables, the Buyer agrees to assign all rights and title to Uncollected Receivables to the Seller immediately following the payment of such Losses (in an amount equal to the Losses actually paid from Seller to Buyer).
- 5.1.6 The Seller shall not be liable for the indemnification obligations pursuant to <u>Section 5.1.1</u> or <u>Section 5.1.3</u> until the aggregate amount of Losses with respect to matters referred to in <u>Section 5.1.1</u> and <u>Section 5.1.3</u> exceeds One Hundred Thousand and 00/100 Dollars (\$100,000) (the "<u>Deductible</u>"), after which the Seller will be responsible for all Losses in excess of the Deductible. Notwithstanding anything herein to the contrary, the Deductible will not apply (i) in the case of fraud or intentional misrepresentation of the Seller or (ii) with respect to any breach of <u>Sections 2.1</u> (Organization and Predecessors), <u>2.2</u> (Capitalization and Title), <u>2.3</u> (Power and Authorization), <u>2.5(d)</u> (Breach of Organizational Documents), the last sentence of <u>2.6.1</u> (Financial Statements), <u>2.10.1</u> (Ownership of Assets), <u>2.15</u> (Legal Compliance; Illegal Payments; Permits), <u>2.17</u> (Employee Benefit Plans), <u>2.18</u> (Environmental Matters), <u>2.24</u> (Employees), the last sentence of <u>2.25.1</u> (Litigation) or <u>2.29</u> (No Brokers).
- 5.2 <u>Indemnity by the Buyer</u>. Subject to the limitations set forth in this <u>Section 5</u>, the Buyer will indemnify and hold harmless the Seller and the Seller's Affiliates, and the Representatives and Affiliates of each of the foregoing Persons (each, a "<u>Seller Indemnified Person</u>"), from, against and in respect of any and all Losses incurred or suffered by Seller Indemnified Persons or any of them as a result of, arising out of or relating to, directly or indirectly:
 - 5.2.1 any fraud of the Buyer or any breach of, or inaccuracy in, any representation or warranty made by the Buyer in Section 3; or
- 5.2.2 any breach or violation of any covenant or agreement of the Buyer (including under this <u>Section 5</u>) or any covenant or agreement of the Company to the extent required to be performed or complied with by the Company after the Closing Date, in either case in or pursuant to this Agreement; or
- 5.2.3 subject to the satisfaction in full of all of the Seller's indemnification obligations set forth in <u>Section 5.1</u>, the Buyer's failure to cause the Company to timely pay all Liabilities of the Company either shown on the Closing Balance Sheet or disclosed in the Schedules attached hereto, in either such case with a specific reference that the Seller has personally guaranteed or has personal liability with regard thereto.
- 5.3 <u>Time for Claims</u>. No claim may be made or suit instituted seeking indemnification pursuant to <u>Section 5.1.1, 5.1.3 or 5.2.1</u> for any breach of, or inaccuracy in, any representation or warranty unless a written notice describing such breach or inaccuracy in reasonable detail in light of the circumstances then known to the Indemnified Person, is provided to the Indemnifying Person:

- 5.3.1 at any time, in the case of any breach of, or inaccuracy in, the representations and warranties set forth in <u>Sections 2.1</u> (Organization and Predecessors), <u>2.2</u> (Capitalization and Title), <u>2.3</u> (Power and Authorization), <u>2.5(d)</u> (Breach of Organizational Documents), the last sentence of <u>2.6.1</u> (Financial Statements) <u>2.10.1</u> (Ownership of Assets), the last sentence of <u>2.25.1</u> (Litigation), <u>2.29</u> (No Brokers), <u>3.1</u> (Organization), <u>3.2</u> (Power and Authorization), <u>3.4(d)</u> (Breach of Organizational Documents) or <u>3.5</u> (No Brokers);
 - 5.3.2 at any time, in the case of any claim or suit based upon violations of law, fraud or intentional misrepresentation;
- 5.3.3 at any time prior to the thirtieth day after the expiration of the applicable statute of limitations (taking into account any tolling periods and other extensions) in the case of any breach of, or inaccuracy in, the representations and warranties set forth in Sections 2.15 (Legal Compliance; Illegal Payments; Permits), 2.17 (Employee Benefit Plans), 2.18 (Environmental Matters) or 2.24 (Employees); and
- 5.3.4 at any time prior to last day of the 36th month following the Effective Date, in the case of any breach of, or inaccuracy in, any other representation and warranty or any Loss arising from Liabilities described in Section 5.1.3.

Claims for indemnification pursuant to any other provision of Sections 5.1 and 5.2 are not subject to the limitations set forth in this Section 5.3.

5.4 Third Party Claims.

- 5.4.1 <u>Notice of Claim</u>. If any third party notifies an Indemnified Person with respect to any matter (a "<u>Third Party Claim</u>") which may give rise to an Indemnified Claim against an Indemnifying Person under this <u>Section 5</u>, then the Indemnified Person will promptly give written notice to the Indemnifying Person; provided, however, that no delay on the part of the Indemnified Person in notifying the Indemnifying Person will relieve the Indemnifying Person from any obligation under this <u>Section 5</u>.
- 5.4.2 <u>Assumption of Defense, etc.</u> The Indemnifying Person will be entitled to participate in the defense of any Third Party Claim that is the subject of a notice given by the Indemnified Person pursuant to <u>Section 5.4.1</u>. In addition, the Indemnifying Person will have the right to assume the defense of the Indemnified Person against the Third Party Claim with counsel satisfactory to the Indemnified Person so long as (a) the Indemnifying Person gives written notice to the Indemnified Person within fifteen days after the Indemnified Person has given notice of the Third Party Claim that the Indemnifying Person will indemnify the Indemnified Person from and against the entirety of any and all Losses the Indemnified Person may suffer resulting from, arising out of, relating to, in the nature of, or caused by the Third Party Claim, (b) the Indemnifying Person provides the Indemnified Person with evidence reasonably acceptable to the Indemnified Person that the Indemnifying Person will have adequate financial resources to defend against the Third Party Claim and fulfill its indemnification obligations hereunder, (c) the Third Party Claim involves only money damages and does not seek an injunction or other equitable relief against the Indemnified Person, (d) the Indemnified Person has not been advised by counsel that an actual or potential conflict exists between the Indemnified Person and the

Indemnifying Person in connection with the defense of the Third Party Claim, (e) the Third Party Claim does not relate to or otherwise arise in connection with Taxes (provided, that with respect to Taxes, this Section 5.4.2 shall not affect the Seller's right to defend any claim of a Governmental Authority as permitted by Section 6.3(c)) or any criminal or regulatory enforcement Action and (f) the Indemnifying Person conducts the defense of the Third Party Claim actively and diligently. The Indemnified Person may retain separate co-counsel at its sole cost and expense and participate in the defense of the Third Party Claim; provided, however, that the Indemnifying Person will pay the fees and expenses of separate co-counsel retained by the Indemnified Person that are incurred prior to the Indemnifying Person's assumption of control of the defense of the Third Party Claim.

5.4.3 <u>Limitations on Indemnifying Person</u>. The Indemnifying Person will not consent to the entry of any judgment or enter into any compromise or settlement with respect to the Third Party Claim without the prior written consent of the Indemnified Person, which consent will not be unreasonably withheld, unless such judgment, compromise or settlement (a) provides for the payment by the Indemnifying Person of money as sole relief for the claimant, (b) results in the full and general release of Buyer Indemnified Persons or Seller Indemnified Persons, as applicable, from all liabilities arising or relating to, or in connection with, the Third Party Claim and (c) involves no finding or admission of any violation of Legal Requirements or the rights of any Person and no effect on any other claims that may be made against the Indemnified Person.

5.4.4 Indemnified Person's Control. If the Indemnifying Person does not deliver the notice contemplated by clause (a), or the evidence contemplated by clause (b), of Section 5.4.2 within 15 days after the Indemnified Person has given notice of the Third Party Claim, or otherwise at any time fails to conduct the defense of the Third Party Claim actively and diligently, the Indemnified Person may defend, and may consent to the entry of any judgment or enter into any compromise or settlement with respect to, the Third Party Claim in any manner it may deem appropriate (and the Indemnified Person need not consult with, or obtain any consent from, the Indemnifying Person in connection therewith). If such notice and evidence is given on a timely basis and the Indemnifying Person conducts the defense of the Third Party Claim actively and diligently but any of the other conditions in Section 5.4.2 is or becomes unsatisfied, the Indemnified Person may defend, and may consent to the entry of any judgment or enter into any compromise or settlement with respect to, the Third Party Claim with the consent of the Indemnifying Person, which consent shall not be unreasonably delayed, conditioned or withheld. In the event that the Indemnified Person conducts the defense of the Third Party Claim pursuant to this Section 5.4.4, the Indemnifying Person will (a) advance the Indemnified Person promptly and periodically for the costs of defending against the Third Party Claim (including attorneys' fees and expenses) and (b) remain responsible for any and all other Losses that the Indemnified Person may incur or suffer resulting from, arising out of, relating to, in the nature of or caused by the Third Party Claim to the fullest extent provided in this Section 5.

5.4.5 <u>Consent to Jurisdiction Regarding Third Party Claim</u>. The Buyer and the Seller, in their capacities as Indemnifying Persons, hereby consent to the non-exclusive jurisdiction of any court in which any Third Party Claim may be properly brought against any Indemnified Person for purposes of any claim which such Indemnified Person may have against such Indemnifying Person pursuant to this Agreement in connection with such Third Party Claim, and in furtherance thereof, the provisions of <u>Section 7.12</u> are incorporated herein by reference, *mutatis mutandis*.

- 5.5 <u>Knowledge and Investigation</u>. The right of any Buyer Indemnified Person or Seller Indemnified Person to indemnification pursuant to this <u>Section 5</u> will not be affected by any investigation conducted or knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy of any representation or warranty, or performance of or compliance with any covenant or agreement, referred to in <u>Sections 5.1 and 5.2</u>.
- 5.6 <u>Insurance</u>. Any Losses otherwise payable to an Indemnified Person shall be reduced to the extent of any insurance proceeds received by the Indemnified Person with respect to the event giving rise to the Loss, net of the costs of collection of such insurance proceeds or claim. Any Person entitled to be indemnified hereunder agrees to use commercially reasonable efforts to collect all applicable insurance coverage and all Losses sought by any Person shall be net of the proceeds received by such Indemnified Person, net of the costs of collection.
- 5.7 <u>Remedies Cumulative</u>. The rights of each Buyer Indemnified Person and Seller Indemnified Person under this <u>Section 5</u> are cumulative, and each Buyer Indemnified Person and Seller Indemnified Person, as the case may be, will have the right in any particular circumstance, in its sole discretion, to enforce any provision of this <u>Section 5</u> without regard to the availability of a remedy under any other provision of this <u>Section 5</u>.
- 5.8 <u>Right of Setoff</u>. Upon notice to the Seller specifying in reasonable detail the basis therefor, the Buyer may set off any amount to which it may be entitled from the Seller against amounts otherwise payable, if any, under <u>Section 1.6</u>. The exercise of such right of setoff by the Buyer in good faith, whether or not ultimately determined to be justified, will not constitute an event of default under this Agreement or any other agreement between the Buyer or any of its Affiliates and the Seller. Neither the exercise or failure to exercise such right of setoff will constitute an election of remedies or limit the Buyer in any manner in the enforcement of any other remedies that may be available to it. Notwithstanding the foregoing, if the Buyer sets off an amount pursuant to this <u>Section 5.8</u> and it is later finally mutually determined by the parties or pursuant to <u>Section 7.12</u> that such setoff amount was in excess of the amount which the Seller was required to indemnify the Buyer, then the Buyer shall pay interest to the Seller on such excess amount at the Prime Rate as published by the Wall Street Journal on the date such setoff is made.

6. TAX MATTERS

- 6.1 <u>Representations and Obligations Regarding Taxes</u>. The Seller represents and warrants to and agrees with the Buyer as follows, in each case except to the extent set forth on <u>Schedule 6</u> (for purposes of this <u>Section 6</u>, the "<u>Company</u>" includes the Company and/or any corporation that at any time has been a subsidiary of the Company):
 - (a) The Company has timely filed all Tax Returns that it was required to file. All of those Tax Returns were correct and complete. Schedule 6.1(a) lists all Federal, state, local and foreign Tax Returns filed with respect to the Company for taxable periods ending after December 31, 2000. The Company has delivered to the Buyer correct and complete copies of all Federal income Tax Returns and all state and local income or franchise Tax returns for each taxable period ending after December 31, 2000. All material elections with respect to Taxes affecting the Company are disclosed on or attached to a Tax Return of the Company.

- (b) Except as set forth in the second succeeding sentence, all Taxes owed by the Company (whether or not shown on any Tax Return and whether or not any Tax Return was required) have been paid, and the Company has made adequate accrual in its Reviewed Financials for all Taxes through the date hereof not yet due and payable. The Company has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party. The unpaid Taxes of the Company (i) did not, as of the most recent fiscal month end, exceed the reserve for Tax liability (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the Most Recent Balance Sheet (rather than in any notes thereto) and (ii) will not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Company in filing its Tax Returns.
- (c) The Company has not made any change in accounting methods, received a ruling from any taxing authority or signed an agreement with respect thereto or signed any closing agreement with respect to any Tax year. The Company is not required to include in income any adjustment pursuant to section 481(a) of the Code by reason of any voluntary change in accounting method (nor has any Governmental Authority proposed in writing any such adjustment or change of accounting method).
- (d) The Company is not currently the beneficiary of any extension of time within which to file any Tax Return. All private letter rulings issued by the IRS to the Company (and any corresponding ruling or determination of any state, local or foreign Governmental Authority, including, without limitation, any clearance for taxation purposes in relation to any transaction or arrangement involving the Company) have been disclosed on Schedule 6.1(d), and there are no pending requests for any rulings (or corresponding determinations). There is no power of attorney with respect to any Tax executed or filed with any Governmental Authority.

- (e) There are no liens on any of the Assets that arose in connection with any failure (or alleged failure) to pay any Tax, except for liens for Taxes not yet due. The Company has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency which waiver or extension has not since expired. Schedule 6.1(e) lists those Tax Returns filed with respect to the Company for taxable periods ended on or after December 31, 2000 that have been audited or that have been the subject of any written or unwritten notice of any audit, examination, investigation or other proceeding. The Company has delivered to the Buyer correct and complete copies of all audit, examination, revenue agent's and other reports of, and any closing agreement with, any taxing jurisdiction with respect to taxable periods ended on or after December 31, 2000. No issue relating to Taxes has been raised in writing by a Governmental Authority during any pending audit or examination, and no issue relating to Taxes was raised in writing by a Governmental Authority in any completed audit or examination, that reasonably can be expected to recur in a later taxable period. Except as noted on Schedule 6.1(e), there is no dispute or claim concerning any Tax liability of the Company either (i) claimed or raised by any Governmental Authority in writing or (ii) as to which any of the directors or officers (or employees responsible for Tax matters) of the Company has knowledge. No claim has ever been made by a Governmental Authority in a jurisdiction where the Company does not file Tax Returns that it is or may be subject to taxation by that jurisdiction or that it satisfies or may satisfy the nexus or permanent establishment requirements to be subject to taxation by that jurisdiction. Except to the extent set forth on Schedule 6.1(e), no director or officer (or employee responsible for Tax matters) of the Company has any reason to believe that any Governmental Authority will assert liability
- (f) The Company has not made any payments, is not obligated to make any payments and is not a party to any agreement that under certain circumstances could obligate it to make any payments that will not be deductible under section 280G of the Code or that would give rise to any obligation to indemnify any Person for any excise tax payable pursuant to section 4999 of the Code. The Company has not entered into any sale leaseback or leveraged lease transaction that fails to satisfy the requirements of Revenue Procedure 75-21 or Revenue Procedure 2001-28 (or similar provisions of foreign law) or any safe harbor lease transaction, and the Company is not a lessor or lessee of any tangible property located outside the United States. None of the property of the Company is tax-exempt use property within the meaning of section 168(h) of the Code, is property that is required to be treated as being owned by any other Person pursuant to the safe harbor lease provisions of former section 168(f)(8) of the Code or is property that directly or indirectly secures any debt the interest on which is exempt from federal income tax pursuant to section 103 of the Code. The Company is not a controlled corporation or a distributing corporation in respect of a distribution to which section 355(e) of the Code could apply by reason of the acquisition of the Company's shares pursuant to this Agreement. The Company has not distributed stock of another entity, or had its stock distributed by another entity, in a transaction that was purported or intended to be governed in whole or in part by section 355 or section 361 of the Code. No Debt of the Company consists of "corporate acquisition indebtedness" within the meaning of section 279 of the Code.

- (g) The Company (i) has not made an election, or is required to treat any of its assets as owned by another Person pursuant to the provisions of Section 168(f) of the Internal Revenue Code of 1954 or as tax-exempt bond financed property or tax-exempt use of property within the meaning of Section 168 of the Code; (ii) owns no property that is subject to a "section 467 rental agreement" as defined in Section 467 of the Code; (iii) has not elected at any time to be treated as an S corporation within the meaning of Sections 1361 or 1362 of the Code; or (iv) made any of the foregoing elections or is required to apply any of the foregoing rules under any comparable state or local Tax provision.
 - (h) The Company does not have a non-accountable expense reimbursement arrangement with the meaning of Treasury Regulation Section 1.62-2(c).
- (i) No taxing authority is asserting or to the Seller's Knowledge threatening to assert a claim against the Company under or as a result of Section 482 of the Code or any similar provision of any foreign, state or local Tax law.
- (j) The Company will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion of any taxable period) after the Closing Date as a result of (i) a closing agreement as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or non-U.S. Tax law); (ii) an installment sale or open transaction disposition occurring on or prior to the Closing Date; (iii) the use of the cash basis method of accounting; or (iv) a prepaid amount received on or prior to the Closing Date.
- (k) The Company has not been a United States real property holding corporation within the meaning of section 897(c)(2) of the Code during the applicable period specified in section 897(c)(1)(A)(ii) of the Code. The Company does not have, and has not had, a permanent establishment in any foreign country, as defined in any applicable income tax treaty to which the United Sates and the foreign country are parties or under the law of the foreign country. The Company does not have an overall foreign loss within the meaning of section 904(f) of the Code.
- (l) The Company is not a party to any Tax allocation or sharing agreement. The Company (i) has not been a member of an Affiliated Group filing a consolidated Federal income Tax Return and (ii) has no liability for the Taxes of any Person under Treasury regulation section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by contract or otherwise. The Company is not a party to any joint venture, partnership or other arrangement or contract that could be treated as a partnership for Federal income tax purposes. Details of all claims or elections for any reliefs, allowances or credits, the making or claiming of which was taken into account in computing the provision or reserve for taxation in the Reviewed Financials and which have not been made by the Company are set out in Schedule 6.1(1). Except as provided in the Reviewed Financials, the Company is not and will not be under any obligation to make and has no entitlement to receive in respect of any period ending on or before the Closing any payment under any taxation sharing arrangement in respect of profits, gains or losses of another Person.

- (m) The Company has disclosed on its Federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of Federal income Tax within the meaning of section 6662 of the Code. The Company has not engaged in any "listed transaction" or "reportable transaction" within the meaning of section 6707A(c) of the Code or Treasury regulation section 1.6011-4(b) or any transaction (i) that was marketed to the Company in writing as a transaction that is intended to generate substantial tax benefits and (ii) with respect to which the Company has paid a promoter of the transaction total fees in excess of \$100.000.
- (n) The transfer of the Shares to the Buyer pursuant to the terms of this Agreement will not result in any Tax liability to the Company or result in a reduction of the amount of any net operating loss, net operating loss carryover, net capital loss, net capital loss carryover, Tax credit, Tax credit carryover, excess charitable contribution or basis of property that otherwise would be available to the Company by reason or as a result of deferred intercompany transactions, excess loss accounts, or otherwise.
- (o) No charge to taxation will arise on the Company and no arrangement, concession, dispensation or agreement which the Company has with or from any Tax Authority will be prejudiced by virtue of the entering into of this Agreement or the consummation of the transactions contemplated hereby.
- (p) Except as described in <u>Schedule 6.1(p)</u>, the Company is not treated for any taxation purpose as resident in a country other than the country of its incorporation and the Company has not, nor has it within the past six (6) years had, a branch, agency or permanent establishment in a country other than the country of its incorporation.
- (q) No transactions or arrangements involving the Company have taken place or are in existence which are such that any provision relating to transfer pricing might be invoked by a Governmental Authority.

6.2 Covenants With Respect To Taxes

(a) the Buyer shall (i) grant to the Seller reasonable access to the Company's books and records (including tax workpapers and returns and correspondence with tax authorities), including the right to take extracts therefrom and make copies thereof, to the extent that the books and records relate to the operations of the Company during taxable periods ending on or prior to or that include the Closing Date as may be necessary for Tax return preparation, audit, or other financial reporting matters, and (ii) otherwise cooperate with the Seller in connection with any audit of Taxes that relates to the business of the Company prior to Closing, in each case to the extent relevant to the Seller's indemnification obligations under Section 6.3.

(b) the Buyer shall be responsible for preparing and filing, or causing the Company to prepare and file, all Tax Returns of the Company for the taxable periods ending after the Closing Date, and the Seller shall be responsible for filing all Tax Returns of the Company for taxable periods ending on or before the Closing Date. The Seller shall furnish such Tax Returns to the Buyer for its review, comment and approval at least 30 days prior to the due date (or extended due date) for filing such Tax Returns and such Tax Returns shall be prepared in accordance with past practice, except as required by law or regulation. The Seller shall pay all Taxes required to be paid with respect to such Tax Returns, reduced by the amount of the accrual for those Taxes in the Closing Balance Sheet to the extent included in computing net working capital and reduced by the amount of such Taxes attributable to transactions occurring after the Effective Date and on or before the Closing. The Seller shall pay to the Buyer within five (5) days after the date on which Taxes are paid with respect to periods beginning before the Effective Date and ending on or after the Effective Date an amount equal to the portion of those Taxes that relates to the portion of the taxable period ending on the Effective Date. Other than the Taxes referred to in the preceding sentence that the Seller shall pay, for purposes of this Agreement, in the case of any period that begins before the Effective Date and ends after the Effective Date, any tax based directly or indirectly on gross or net income or receipts or imposed in respect of specific transactions, and any credits available with respect to any Tax, shall be allocated by assuming that the taxable period ended on the Effective Date, and any other Tax shall be allocated based on the number of days in the taxable period ending on the Effective Date divided by the total number of days in the taxable period.

6.3 Indemnification for Taxes.

- (a) The Seller shall indemnify the Buyer and its Affiliates, including, after the Closing, the Company (each herein sometimes referred to as an "Indemnified Taxpayer") against, and agrees to protect, save and hold harmless each Indemnified Taxpayer from, any and all claims, damages, deficiencies and losses and all expenses, including, without limitation, attorneys', accountants' and experts' fees and disbursements (all herein referred to as "Tax Losses"), resulting from:
 - (i) (A) any Taxes of the Company or the Seller allocable to any period ending on or prior to the Effective Date or allocable to any period that begins before and ends after the Effective Date (but only to the extent that such Taxes relate to the portion of the taxable period ending on the Effective Date in accordance with Section 6.2(b)), except to the extent the liability for such Taxes is included in computing net working capital and (B) any Taxes of the Company or any corporation that is or was a member of an Affiliated Group of which the Company was or is a member, or any liability of any of the foregoing for the Taxes of any Person, whether as a transferee or successor, by contract or otherwise;

- (ii) Any Taxes arising from or occasioned by the sale of the Company's capital stock pursuant to this Agreement; or
- (iii) Any misrepresentation or breach of any representation, warranty or obligation set forth in this Section 6.
- (b) Subject to the resolution of any Tax contest pursuant to <u>Section 6.3(c)</u>, upon notice from the Buyer to the Seller that an Indemnified Taxpayer is entitled to an indemnification payment for a Tax Loss pursuant to <u>Section 6.3(a)</u>, the Seller thereupon shall pay to the Indemnified Taxpayer an amount that, net of any Taxes imposed on the Indemnified Taxpayer with respect to the payment, will indemnify and hold the Indemnified Taxpayer harmless from the Tax Loss.
- (c) (i) If a claim shall be made by any Governmental Authority that, if successful, would result in the indemnification of an Indemnified Taxpayer, the Indemnified Taxpayer shall promptly notify the Seller in writing of that fact; provided, however, that any failure to give the notice will not waive any rights of the Indemnified Taxpayer.
 - (ii) The Seller shall have the right to defend the Indemnified Taxpayer against the claim with counsel satisfactory to the Indemnified Taxpayer so long as (A) the Seller notifies the Indemnified Taxpayer in writing within 30 days after the Indemnified Taxpayer has given notice of the claim that the Seller will indemnify the Indemnified Taxpayer from and against the entirety of any Tax Losses the Indemnified Taxpayer may suffer resulting from, arising out of, relating to, in the nature of, or caused by the claim, (B) if the Indemnified Taxpayer is requested to pay the Tax claimed and sue for a refund, the Seller shall have advanced to the Indemnified Taxpayer, on an interest free basis, the full amount the Indemnified Taxpayer is required to pay, and (C) the Seller conducts the defense of the claim actively and diligently.
 - (iii) Subject to the provisions of paragraph (ii) above, the Seller shall be entitled to prosecute the contest to a determination in a court of initial jurisdiction, and if the Seller shall reasonably request, to a determination in an appellate court provided that, if requested by the Indemnified Taxpayer, the Seller shall provide to the Indemnified Taxpayer an opinion, in form and substance satisfactory to the Indemnified Taxpayer, of counsel satisfactory to the Indemnified Taxpayer, that there exists a reasonable basis for the Company to prevail on that appeal.

- (iv) The Seller shall not be entitled to settle or to contest any claim relating to Taxes if the settlement of, or an adverse judgment with respect to, the claim would be likely, in the good faith judgment of the Indemnified Taxpayer, to cause the liability for any Tax of the Indemnified Taxpayer or of any Affiliate of the Indemnified Taxpayer for any taxable period ending after the Closing Date to increase (including, without limitation, by making any election or taking any action having the effect of making any election, by deferring the inclusion of any amount in income or by accelerating the deduction of any amount or the claiming of any credit) or to take a position that, if applied to any taxable period ending after the Closing Date, would be adverse to the interest of the Indemnified Taxpayer or any Affiliate of the Indemnified Taxpayer.
- (v) If, after actual receipt by the Indemnified Taxpayer of an amount advanced by the Seller pursuant to paragraph (ii)(D) above, the extent of the liability of the Indemnified Taxpayer with respect to the indemnified matter shall be established by the judgment or decree of a court that has become final or a binding settlement with an administrative agency having jurisdiction thereof that has become final, the Indemnified Taxpayer shall promptly pay to the Seller any refund received by or credited to the Indemnified Taxpayer with respect to the indemnified matter (together with any interest paid or credited thereon by the Governmental Authority and any recovery of legal fees from the Governmental Authority); provided, however, that the Indemnified Taxpayer shall have been indemnified and held harmless from all Tax Losses by reason of any indemnification payments retained by the Indemnified Taxpayer net of any Taxes imposed on the Indemnified Taxpayer with respect to indemnification payments received by the Indemnified Taxpayer or with respect to the receipt of any payment from the Governmental Authority. Notwithstanding the foregoing, the Indemnified Taxpayer shall not be required to make any payment hereunder before the time that the Seller shall have made all payments or indemnities then due with respect to Indemnified Taxpayer pursuant to this Section 6.
- (vi) If any of the conditions in Section 6.3(c)(i) above are or become unsatisfied, (A) the Indemnified Taxpayer may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, the claim in any manner it may deem appropriate (and the Indemnified Taxpayer need not consult with, or obtain any consent from, the Seller in connection therewith), (B) the Seller will reimburse the Indemnified Taxpayer promptly and periodically for the costs of defending against the claim (including, without limitation, attorneys', accountants' and experts' fees and disbursements) and (C) the Seller will remain responsible for any Losses the Indemnified Taxpayer may suffer to the fullest extent provided in this Section 6.3.

- (d) Anything to the contrary in this Agreement notwithstanding, the indemnification obligations of the Seller under this Section 6 shall survive the Closing until the end of the applicable statutes of limitations. With respect to any indemnification obligation for any Tax for which a Governmental Authority asserts a claim within 90 days before the end of the applicable statute of limitations, an Indemnified Taxpayer shall be treated as having provided timely notice to the Seller by providing written notice to the Seller on or before the 90th day after the Indemnified Taxpayer's receipt of a written assertion of the claim by the Governmental Authority.
- (e) All transfer, documentary, sales, use, stamp, registration and other similar Taxes and fees (including any penalties and interest) incurred in connection with this Agreement shall be paid by the Seller when due, and the Seller will, at its own expense, file all necessary Tax Returns and other documentation with respect to all the transfer, documentary, sales, use, stamp, registration and other Taxes and fees, and, if required by applicable law, the Buyer will, and will cause its Affiliates to, join in the execution of any of those Tax Returns and other documentation.

7. MISCELLANEOUS

- 7.1 <u>Notices</u>. All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be delivered, given or otherwise provided:
 - (a) by hand (in which case, it will be effective upon delivery);
 - (b) by facsimile (in which case, it will be effective upon receipt of confirmation of good transmission); or
 - (c) by overnight delivery by a nationally recognized courier service (in which case, it will be effective on the Business Day after being deposited with such courier service);

in each case, to the address (or facsimile number) listed below:

If to the Seller, to:

Alan B. Roberts 1001 County Road 230 Giddings, Texas 78942 Telephone: (979) 542-0272 Facsimile: (979) 542-9402

with a copy (which shall not constitute notice) to:

Decker, Jones, McMackin, McClane, Hall & Bates, P. C. 801 Cherry Street, Unit #46 Fort Worth, Texas 76102 Telephone: (817) 336-2400 Facsimile: (817) 336-2181 Attention: James L. Stripling, Esq. If to the Buyer, to:

MasTec North America, Inc.

c/o MasTec. Inc.

800 S. Douglas Road, 12th Floor

Coral Gables, FL 33134 Telephone: (305) 406-1892 Facsimile: (305) 406-1900

Attention: Pablo A. Alvarez, Executive Vice

President, Mergers and Acquisitions

with a copy (which shall not constitute notice) to:

MasTec, Inc.

800 S. Douglas Road, 12th Floor Coral Gables, FL 33134 Telephone: (305) 406-1849 Facsimile: (305) 406-1947

Attention: Albert de Cardenas, Executive Vice President and General Counsel

with a copy (which shall not constitute notice) to:

Greenberg Traurig, P.A. 1221 Brickell Avenue Miami, FL 33131

Telephone number: (305) 579-0724 Facsimile number: (305) 961-5724 Attention: David Barkus, Esq.

Each of the parties to this Agreement may specify a different address or facsimile number by giving notice in accordance with this <u>Section 7.1</u> to each of the other parties hereto.

7.2 <u>Succession and Assignment; No Third-Party Beneficiary</u>. Subject to the immediately following sentence, this Agreement will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, each of which such successors and permitted assigns will be deemed to be a party hereto for all purposes hereof. No party may assign, delegate or otherwise transfer either this Agreement or any of its rights, interests, or obligations hereunder (other than transfers by operation of law) without the prior written approval of the other parties; provided, however, that the Buyer may (a) assign any or all of its rights and interests hereunder to one or more of its Affiliates and (b) designate one or more of its Affiliates to perform its obligations hereunder, in each case, so long as the Buyer is not relieved of any Liability hereunder. Except as expressly set forth in <u>Section 5</u> with respect to Indemnified Persons who are not parties to this Agreement, this Agreement is for the sole benefit of the parties and their permitted successors and assignees and nothing herein expressed or implied will give or be construed to give any Person, other than the parties and such successors and assignees, any legal or equitable rights hereunder.

7.3 <u>Amendments and Waivers</u>. No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by the Buyer and the Seller, or in the case of a waiver, by the party against whom the waiver is to be effective. No waiver by any party of any breach or violation or, default under or inaccuracy in any representation, warranty or covenant hereunder, whether intentional or not,

will be deemed to extend to any prior or subsequent breach, violation, default of, or inaccuracy in, any such representation, warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No delay or omission on the part of any party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.

- 7.4 Entire Agreement. This Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and supersedes any and all prior and contemporaneous discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, with respect thereto.
- 7.5 <u>Schedules; Listed Documents, etc.</u> Neither the listing nor description of any item, matter or document in any Schedule hereto nor the furnishing or availability for review of any document will be construed to modify, qualify or disclose an exception to any representation or warranty of any party made herein or in connection herewith, except to the extent that such representation or warranty specifically refers to such Schedule and such modification, qualification or exception is clearly described in such Schedule.
- 7.6 <u>Counterparts; Execution</u>. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument. This Agreement will become effective when duly executed by each party hereto. The facsimile transmission of a signed signature page, by any party to the other(s), shall constitute valid execution and acceptance of this Agreement by the signing/transmitting party.
- 7.7 <u>Survival</u>. The covenants and agreements and, subject to <u>Section 5.3</u>, the representations and warranties set forth in this Agreement shall survive and remain in effect after the Closing.
- 7.8 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. In the event that any provision hereof would, under applicable law, be invalid or unenforceable in any respect, each party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable law.
- 7.9 <u>Headings</u>. The headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation hereof
- 7.10 <u>Construction</u>. The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties and no presumption or burden of proof will arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement. The parties intend that each representation, warranty and covenant contained herein will have independent significance. If any party has breached or violated, or if there is an inaccuracy in, any representation, warranty or covenant contained herein in any

respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which the party has not breached or violated, or in respect of which there is not an inaccuracy, will not detract from or mitigate the fact that the party has breached or violated, or there is an inaccuracy in, the first representation, warranty or covenant.

7.11 <u>Governing Law</u>. This Agreement, the rights of the parties and all Actions arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the domestic substantive laws of the State of Florida, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.

7.12 Jurisdiction; Venue; Service of Process.

- 7.12.1 <u>Jurisdiction</u>. Subject to the provisions of <u>Section 4.4.4</u> and <u>5.4.5</u>, each party, by its execution hereof, (a) hereby irrevocably submits to the exclusive jurisdiction of the state courts of the State of Florida or the United States District Courts located in the Southern District of Florida for the purpose of any Action between the parties arising in whole or in part under or in connection with this Agreement, (b) hereby waives to the extent not prohibited by applicable law, and agrees not to assert, by way of motion, as a defense or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that any such Action brought in one of the above-named courts should be dismissed on grounds of *forum non conveniens*, should be transferred or removed to any court other than one of the above-named courts, or should be stayed by reason of the pendency of some other proceeding in any other court other than one of the above-named courts, or that this Agreement or the subject matter hereof may not be enforced in or by such court and (c) hereby agrees not to commence any such Action other than before one of the above-named courts. Notwithstanding the previous sentence a party may commence any Action in a court other than the above-named courts solely for the purpose of enforcing an order or judgment issued by one of the above-named courts.
- 7.12.2 <u>Venue</u>. Each party agrees that for any Action between the parties arising in whole or in part under or in connection with this Agreement, such party bring Actions only in Miami-Dade County, Florida. Each party further waives any claim and will not assert that venue should properly lie in any other location within the selected jurisdiction.
- 7.12.3 <u>Service of Process</u>. Each party hereby (a) consents to service of process in any Action between the parties arising in whole or in part under or in connection with this Agreement in any manner permitted by Florida law, (b) agrees that service of process made in accordance with clause (a) or made by registered or certified mail, return receipt requested, at its address specified pursuant to <u>Section 7.1</u>, will constitute good and valid service of process in any such Action and (c) waives and agrees not to assert (by way of motion, as a defense, or otherwise) in any such Action any claim that service of process made in accordance with clause (a) or (b) does not constitute good and valid service of process.

- 7.13 <u>Waiver of Jury Trial</u>. TO THE EXTENT NOT PROHIBITED BY APPLICABLE LAW THAT CANNOT BE WAIVED, THE PARTIES HEREBY WAIVE, AND COVENANT THAT THEY WILL NOT ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE), ANY RIGHT TO TRIAL BY JURY IN ANY ACTION ARISING IN WHOLE OR IN PART UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY OF THE CONTEMPLATED TRANSACTIONS, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES IRREVOCABLY TO WAIVE THEIR RIGHTS TO TRIAL BY JURY IN ANY ACTION WHATSOEVER BETWEEN OR AMONG THEM RELATING TO THIS AGREEMENT OR ANY OF THE CONTEMPLATED TRANSACTIONS, WHICH ACTION WILL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.
- 7.14 <u>Expenses</u>. Except as provided in <u>Sections 5 or 6</u>, each party shall bear its own expenses incurred in connection with this Agreement and the transactions contemplated hereby.
- 8. DEFINITIONS: CERTAIN RULES OF CONSTRUCTION.
 - 8.1 <u>Definitions</u>. As used herein, the following terms will have the following meanings:
 - "1933 Act" means the Securities Act of 1933, as amended.
 - "Acceptable Levels" is defined in Section 4.7.
- "Accounts Receivable" means the aggregate amount of accounts, commissions and debts payable to the Company. For all purposes hereunder, the Accounts Receivable shall be valued at their net realizable value, net of an allowance for bad debts.
- "Action" means any claim, action, cause of action or suit (whether in contract, tort or otherwise), litigation (whether at law or in equity, whether civil or criminal), controversy, assessment, arbitration, investigation, hearing, charge, complaint, demand, notice or proceeding to, from, by or before any Governmental Authority.
 - "Actual Company Debt" is defined in Section 1.3.2.
 - "Actual Net Working Capital" defined in Section 1.3.2.
 - "Actual Tangible Net Worth" is defined in Section 1.3.2.
 - "Actual Total Assets" is defined in Section 1.3.2.

"Affiliate" means with respect to any specified Person, (a) each Person directly or indirectly controlling, controlled by or under direct or indirect common control with such specified Person at such time, (b) each Person who is at such time an officer or director of, or direct or indirect beneficial holder of at least 20% of any class of the capital stock of, such specified Person, (c) each Person that is managed by a common group of executive officers and/or directors as such specified Person, (d) the Members of the Immediate Family (i) of each officer, director or holder described in clause (b) and (ii) if such specified Person is an individual, of such specified Person, provided that solely with respect to Section 4.4, James Roberts and David Roberts, shall not be considered a Member of the Immediate Family and (e) each Person of which such specified Person or an Affiliate (as defined in clauses (a) through (d)) thereof will, directly or indirectly, beneficially own at least 20% of any class of equity interests at such time.

"Affiliated Group" means any affiliated group within the meaning of section 1504(a) of the Code or any similar group defined under a similar provision of any Legal Requirement.

"Agreement" is defined in the Preamble.

"Assets" is defined in Section 2.10.1.

"Business" means the Company's pipeline construction and right-of-way business, including but not limited to: (i) cross country construction of pipelines from two (2) to forty-two (42) inches; (ii) industrial metalizing, sandblasting and tank lining painting; (iii) field and shop fabrication; (iv) hydrostatic testing; (v) pipeline repair, retirement and removal; (vi) right-of-way maintenance and mowing; (vii) automated and manual welding; and (viii) compressor station pipelines, retirement, meter stations and demolition, all as it relates to the pipeline construction business.

"Business Day" means any weekday other than a weekday on which banks in New York City, New York, Coral Gables, Florida or Giddings, Texas are authorized or required to be closed.

"Buyer" is defined in the Preamble.

"Buyer Indemnified Person" is defined in Section 5.1.

"Cash Purchase Price" is defined in Section 1.2.

"Closing" is defined in Section 1.4.

"Closing Balance Sheet" is defined in Section 1.3.2.

"Closing Date" is defined in Section 1.4.

"Closing Statement" is defined in Section 1.3.2.

"Code" means the U.S. Internal Revenue Code of 1986, as amended.

"Common Stock" means the common stock of the Company, par value \$1.00.

"Company" is defined in the recitals to this Agreement.

"Company Employees" is defined in Section 2.17(a).

"Company Software" means Owned Software and Licensed Software.

"Company Technology." means any and all Technology used or useful in connection with the Business and any and all Intellectual Property in any and all such Technology.

"Compensation" means, with respect to any Person, all salaries, compensation, remuneration, bonuses or benefits of any kind or character whatever, paid or provided directly or indirectly by the Company to such Person or Affiliates of such Person.

"Competitive Business" is defined in Section 4.4.1(a).

"Contaminant" means any pollutant, hazardous substance, radioactive substance, toxic substance, hazardous waste, medical waste, radioactive waste, special waste, petroleum or petroleum-derived substance or waste, asbestos, polychlorinated biphenyls, or any hazardous or toxic constituent thereof and includes, without limitation, any substance defined in or regulated under any Environmental Law.

"Contemplated Transactions" means the transactions contemplated by this Agreement, including the sale and purchase of the Shares.

"Contractual Obligation" means, with respect to any Person, any contract, agreement, deed, mortgage, lease, license, commitment, promise, undertaking, arrangement, performance bond, warranty obligation or understanding, whether written or oral and whether express or implied, or other document or instrument (including any document or instrument evidencing or otherwise relating to any Debt), to which or by which such Person is a party or otherwise subject or bound or to which or by which any property, business, operation or right of such Person is subject or bound.

"Controlled Group" is defined in Section 2.17(c).

"Corrective Action" means all monitoring well installations, tests, inspections, borings, engineering studies, surveys, appraisals, environmental studies, remediation operations or other activities, including any subsequent monitoring, as may be required, in compliance with Environmental Laws, in order to remediate any environmental contamination located at the Site and identified by the Phase II.

"CPA Firm" means PricewaterhouseCoopers.

"Customer Assets" is defined in Section 2.16.

"<u>Debt</u>" means, with respect to any Person, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees and premiums) of such Person (a) for borrowed money (including overdraft facilities), (b) evidenced by notes, bonds, debentures or similar Contractual Obligations, (c) for the deferred purchase price of property, goods or services (other than trade payables or accruals incurred in the Ordinary Course of Business), (d) under capital leases (in accordance with GAAP), (e) in respect of letters of credit and bankers' acceptances, (f) for Contractual Obligations relating to interest rate protection, swap agreements and collar agreements, (g) in the nature of prepayment penalties in connection with the obligations described in clauses (a) through (f) above, and (h) in the nature of Guarantees of the obligations described in clauses (a) through (f) above of any other Person.

- "Deductible" is defined in Section 5.1.6.
- "Disclosed Contract" is defined in Section 2.19.2.
- "Earn-Out Payments" is defined in Section 1.6.1.
- "Earn-out Period" is defined in Section 1.6.2.
- "Earn-Out Review Period" is defined in Section 1.6.6.
- "EBT" is defined in Section 1.6.3.
- "Effective Date" is defined in the Preamble.
- "Employee Plans" is defined in Section 2.17(a).
- "Encumbrance" means any charge, claim, community or other marital property interest, condition, equitable interest, lien, license, option, pledge, security interest, mortgage, right of way, easement, encroachment, servitude, right of first offer or first refusal, buy/sell agreement and any other restriction or covenant with respect to, or condition governing the use, construction, voting (in the case of any security or equity interest), transfer, receipt of income or exercise of any other attribute of ownership.

"Enforceable" means, with respect to any Contractual Obligation stated to be "Enforceable" by or against any Person, that such Contractual Obligation is a legal, valid and binding obligation of such Person enforceable by or against such Person in accordance with its terms, subject to the effect of applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws now or hereafter in effect affecting the rights of creditors generally, including without limitation, court decisions, general equity principles, and the statutory provisions of the federal bankruptcy code and applicable state law pertaining to the voidability of preferential or fraudulent transfers or conveyances.

"Environmental Consultant" is defined in Section 4.7.

"Environmental Laws" means any Legal Requirement relating to (a) releases or threatened releases of Hazardous Substances, (b) pollution or protection of public health or the environment or worker safety or health or (c) the manufacture, handling, transport, use, treatment, storage, or disposal of Hazardous Substances

- "Equipment" is defined in Section 2.13.
- "ERISA" means the federal Employee Retirement Income Security Act of 1974.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Excluded Assets" is defined in Section 2.6.1.

- "Existing Operations" is defined in Section 1.6.3.
- "Facilities" means any buildings, plants, improvements or structures located on the Real Property.
- "FLSA" is defined in Section 2.24.2.
- "Financials" is defined in Section 2.6.1.
- "GAAP" means generally accepted accounting principles in the United States as in effect from time to time.
- "Government Order" means any order, writ, judgment, injunction, decree, stipulation, ruling, determination or award entered by or with any Governmental Authority in a judicial or administrative proceeding.
- "Governmental Authority" means any United States federal, state or local or any foreign government, or political subdivision thereof, or any multinational organization or authority or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any arbitrator or arbitral body.
- "Guarantee" means, with respect to any Person, (a) any guarantee of the payment or performance of, or any contingent obligation in respect of, any Debt or other Liability of any other Person; (b) any other arrangement whereby credit is extended to any obligor (other than such Person) on the basis of any promise or undertaking of such Person (i) to pay the Debt or other Liability of such obligor, (ii) to purchase any obligation owed by such obligor, (iii) to purchase or lease assets under circumstances that are designed to enable such obligor to discharge one or more of its obligations or (iv) to maintain the capital, working capital, solvency or general financial condition of such obligor; and (c) any liability as a general partner of a partnership or as a venturer in a joint venture in respect of Debt or other obligations of such partnership or venture.
 - "Hazardous Substance" is defined in Section 2.18.
 - "Indemnified Person" means, with respect to any Indemnity Claim, the Person asserting such claim under Section 5.1 or 5.2, as the case may be.
 - "Indemnified Taxpayer" is defined in Section 6.3(a).
- "Indemnifying Person" means, with respect to any Indemnity Claims, (i) the Seller, or (ii) the Buyer under Section 5.1 or 5.2, respectively, against whom such claim is asserted.
 - "Indemnity Claim" means a claim for indemnity under Section 5.1 or 5.2.
- "Intellectual Property" means the entire right, title and interest in and to all proprietary rights of every kind and nature, including all rights and interests pertaining to or deriving from:

- (a) patents, copyrights, mask work rights, technology, know-how, processes, trade secrets, algorithms, inventions, works, proprietary data, databases, formulae, research and development data and computer software or firmware;
 - (b) trademarks, trade names, service marks, service names, brands, trade dress and logos, and the goodwill and activities associated therewith;
- (c) domain names, rights of privacy and publicity, moral rights, and proprietary rights of any kind or nature, however denominated, throughout the world in all media now known or hereafter created;
 - (d) any and all registrations, applications, recordings, licenses, common-law rights and Contractual Obligations relating to any of the foregoing; and
- (e) all Actions and rights to sue at law or in equity for any past, present or future infringement or other impairment of any of the foregoing, including the right to receive all proceeds and damages therefrom, and all rights to obtain renewals, continuations, divisions or other extensions of legal protections pertaining thereto.

"Interim Financials" is defined in Section 2.6.1.

"Inventory" means all inventory related to the Business, wherever located, including all finished goods whether held at any location or facility of the Company or in transit to the Company.

"Investment" means (a) any direct or indirect ownership, purchase or other acquisition by a Person of any notes, obligations, instruments, capital stock, options, securities or ownership interests (including partnership interests and joint venture interests) of any other Person; and (b) any capital contribution or similar obligation by a Person to any other Person.

"IRS" is defined in Section 2.17(e).

"<u>Law</u>" means any federal, national, supranational, state, provincial, local or similar statute, law, ordinance, regulation, rule, code, order, requirement or rule of law (including common law.

"<u>Legal Requirement</u>" means any United States federal, state or local or foreign law, statute, standard, ordinance, code, rule, regulation, resolution or promulgation, or any Government Order, or any license, franchise, permit or similar right granted under any of the foregoing, or any similar provision having the force or effect of law.

"<u>Liability</u>" means, with respect to any Person, any liability or obligation of such Person whether known or unknown, whether asserted or unasserted, whether determined, determinable or otherwise, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, whether incurred or consequential, whether due or to become due and whether or not required under GAAP to be accrued on the financial statements of such Person.

- "Liability Policies" is defined in Section 2.26.
- "<u>Licensed Software</u>" means all Software that is owned by any third party and that is licensed to and used by the Company in the conduct of the Businesses.
 - "Licenses" is defined in Section 2.14.6.
 - "Losses" is defined in Section 5.1.
 - "MasTec" is defined in Section 1.6.4.
- "Material Adverse Effect" means any change in, or effect on, the Business, operations, Assets, prospects or condition (financial or otherwise) of the Company which, when considered either individually or in the aggregate together with all other adverse changes or effects with respect to which such phrase is used in this Agreement, is, or is reasonably likely to be, materially adverse to the Business, operations, Assets, prospects or condition (financial or otherwise) of the Company, taken as a whole.
 - "Maximum Company Debt" is defined in Section 1.3.1.
 - "Maximum Net Working Capital" is defined in Section 1.3.5.
- "Members of the Immediate Family" means, with respect to any individual, (a) such Person's spouse, (b) each parent, brother, sister or child of such Person or such Person's spouse, (c) the spouse of any Person described in clause (b) above, (d) each child of any Person described in clauses (a), (b) or (c) above, (e) each trust created solely for the benefit of one or more of the Persons described in clauses (a) through (d) above and (f) each custodian or guardian of any property of one or more of the Persons described in clauses (a) through (e) above in his capacity as such custodian or guardian.
 - "Minimum Net Working Capital" is defined in Section 1.3.1.
 - "Minimum Tangible Net Worth" is defined in Section 1.3.1.
 - "Minimum Total Assets" is defined in Section 1.3.1.
 - "Most Recent Balance Sheet" is defined in Section 2.6.1.
 - "Most Recent Balance Sheet Date" is defined in Section 2.6.1.
 - "Negative EBT Earn-out Period" is defined in Section 1.6.1(g).
- "Ordinary Course of Business" means an action taken by any Person in the ordinary course of such Person's business which is consistent with the past customs and practices of such Person (including past practice with respect to quantity, amount, magnitude and frequency, standard employment and payroll policies and past practice with respect to management of working capital) which is taken in the ordinary course of the normal day-to-day operations of such Person.

"Organizational Documents" means, with respect to any Person (other than an individual), (a) the certificate or articles of incorporation or organization and any joint venture, limited liability company, operating or partnership agreement and other similar documents adopted or filed in connection with the creation, formation or organization of such Person and (b) all by-laws, voting agreements and similar documents, instruments or agreements relating to the organization or governance of such Person, in each case, as amended or supplemented.

"Owned Software" means all Software used by the Company in the conduct of the Businesses that is owned or purported to be owned by the Company.

"PBGC" is defined in Section 2.17(e).

"<u>Permits</u>" means, with respect to any Person, any license, franchise, permit, consent, approval, right, privilege, certificate or other similar authorization issued by, or otherwise granted by, any Governmental Authority or any other Person to which or by which such Person is subject or bound or to which or by which any property, business, operation or right of such Person is subject or bound.

"Permitted Encumbrance" means (a) statutory liens for current Taxes, special assessments or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings and for which appropriate reserves have been established in accordance with GAAP, (b) mechanics', materialmen's, carriers', workers', repairers' and similar statutory liens arising or incurred in the Ordinary Course of Business which liens have not had and are not reasonably likely to have a Material Adverse Effect, (c) zoning, entitlement, building and other land use regulations imposed by governmental agencies having jurisdiction over any Real Property which are not violated in any material respect by the current or contemplated use and operation of the Real Property, (d) deposits or pledges made in connection with, or to secure payment of, worker's compensation, unemployment insurance, old age pension programs mandated under applicable Legal Requirements or other social security, (e) covenants, conditions, restrictions, easements, encumbrances and other similar matters of record affecting title to but not adversely affecting current or contemplated occupancy or use of the Real Property in any material respect and (f) restrictions on the transfer of securities arising under federal and state securities laws.

"Person" means any individual or corporation, association, partnership, limited liability company, joint venture, joint stock or other company, business trust, trust, organization, Governmental Authority or other entity of any kind.

"Phase II" is defined in Section 4.7.

"Predecessor" is defined in Section 2.1.2.

"Public Reports" is defined in Section 3.6.

"Purchase Price" is defined in Section 1.6.1.

- "Real Property" is defined in Section 2.12.1.
- "Real Property Leases" is defined in Section 2.12.1.
- "Receivables" is defined in Section 2.11.
- "Representative" means, with respect to any Person, any director, officer, employee, agent, consultant, advisor, or other representative of such Person, including legal counsel, accountants, and financial advisors.
 - "Reviewed Financials" is defined in Section 2.6.1.
 - "SEC" is defined in Section 2.15.4(e).
 - "Seller" is defined in the Preamble.
 - "Seller Indemnified Person" is defined in Section 5.2.
- "Seller's Knowledge" means the knowledge of the Seller, Charles Myers, Jewel Pierce, Jarvis Boriack, Robyn Roberts, Scott Crumley, James Roberts, each of whom will be deemed to have knowledge of all such matters as he or she would have discovered, had he or she made reasonable inquiries.
 - "Seller's Objection" is defined in Section 1.3.3.
 - "Shares" is defined in the recitals to this Agreement.
 - "Site" is defined in Section 4.7.
- "<u>Software</u>" means computer software or firmware in any form, including but not limited to computer instructions, commands, programs, modules, routines, procedures, rules, libraries, macros, algorithms, tools, and scripts, and all documentation of or for any of the foregoing.
- "Subsidiary" means, with respect to any specified Person, any other Person of which such specified Person will, at the time, directly or indirectly through one or more Subsidiaries, (a) own at least 50% of the outstanding capital stock (or other shares of beneficial interest) entitled to vote generally, (b) hold at least 50% of the partnership, limited liability company, joint venture or similar interests or (c) be a general partner, managing member or joint venturer.
- "Tax" or "Taxes" means (a) any and all federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, capital stock, franchise, profits, withholding, social security (or similar, including FICA), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, goods and services, mineral oil, alternative or add-on minimum, estimated, or other tax of any kind or any charge of any kind in the nature of (or similar to) taxes whatsoever, including any interest, penalty, or addition thereto, whether disputed or not and (b) any liability for the payment of any amounts of the type described in clause (a) of this definition as a result of being a member of an affiliated, consolidated, combined or unitary group for any period, as a result of any tax sharing or tax allocation agreement, arrangement or understanding, or as a result of being liable for another person's taxes as a transferee or successor, by contract or otherwise.

"Tax Losses" is defined in Section 6.3(a).

"<u>Tax Return</u>" means any return, declaration, report, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Technology" means all inventions, works, discoveries, innovations, know-how, information (including ideas, research and development, know-how, formulas, compositions, processes and techniques, data, designs, drawings, specifications, customer and supplier lists, pricing and cost information, business and marketing plans and proposals, documentation and manuals), computer software, firmware, computer hardware, integrated circuits and integrated circuit masks, electronic, electrical and mechanical equipment and all other forms of technology, including improvements, modifications, works in process, derivatives or changes, whether tangible or intangible, embodied in any form, whether or not protectable or protected by patent, copyright, mask work right, trade secret law or otherwise, and all documents and other materials recording any of the foregoing.

"Territory" is defined in Section 4.4.1(a).

"Third Party Claim" is defined in Section 5.4.1.

"Treasury Regulations" means the regulations promulgated under the Code.

"<u>Uncollected Receivables</u>" means any Accounts Receivable reflected on the Closing Balance Sheet that are not collected within one hundred twenty (120) days of the Effective Date.

"WARN" is defined in Section 2.24.2.

8.2 <u>Rules of Construction</u>. Except as otherwise explicitly specified to the contrary, (a) each reference to a Section, Exhibit or Schedule means a Section of, or Schedule or Exhibit to this Agreement, unless another agreement is specified, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time, (d) words in the singular or plural form include the plural and singular form, respectively, (e) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement and (f) all pronouns and any variations thereof refer to the masculine, feminine or neuter singular or plural as the identity of the Person or Persons may require. The terms "hereof", "herein", "hereunder", "hereto" and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement and not to any particular provision of this Agreement. The word "or" shall not be exclusive. All references herein to "dollars" or "\$" are to United States dollars. Any accounting term used in this Agreement shall have, unless otherwise specifically provided herein, the meaning customarily given such term in accordance with GAAP and all financial computations hereunder will be computed, unless otherwise specifically provided herein, in accordance with GAAP consistently applied. All references herein to any period of days shall mean the relevant number of calendar days unless otherwise specified. All references herein to a "party" or "parties" are to a party or parties to this Agreement unless otherwise specified.

Buyer:	MASTEC NORTH AMERICA, INC.		
	By: /s/ Jose R. Mas		
	Name: Jose R. Mas		
	Title: President and CEO		
Seller:	/s/ Alan B. Roberts		
	Alan B. Roberts		

IN WITNESS WHEREOF, each of the undersigned has executed and delivered this Agreement as of the date first above written.

Form of PROMISSORY NOTE

PROMISSORY NOTE			
	(Date)		

FOR VALUE RECEIVED, PUMPCO, INC. , a corporation located at the address stated below ("Maker") promises, jointly and severally if more than one to pay to the order of General Electric Capital Corporation or any subsequent holder hereof (each, a "Payee") at its office located at 11175 Cicero Drive Suite 600, Alpharetta, GA 30022 or at such other place as Payee may designate as follows:						
(a) the principal sum of, and						
(b) interest on the unpaid principal balance from the date hereof through and inc 05/100 percent (7.05%) per annum (the "Contract Rate") in Sixty (60) consecu-						
Periodic Installment	Amount					

(each, a "Periodic Installment") and a final installment which shall be in the amount of the total outstanding and unpaid principal, accrued interest and any and all amounts due hereunder and under the other Debt Documents (as defined below). The first Periodic Installment shall be due and payable on ______ and the following Periodic Installments and the final installment shall be due and payable on the same day of each succeeding period (each, a "Payment Date"). All payments shall be applied: first, to interest due and unpaid hereunder and under the other Debt Documents, all other amounts due and unpaid hereunder and under the other Debt Documents. The acceptance by Payee of any payment which is less than payment in full of all amounts due and owing at such time shall not constitute a waiver of Payee's right to receive payment in full at such time or at any prior or subsequent time. Interest shall be calculated on the basis of a 365-day year (or a 366-day leap year, as applicable) and will be charged at the Contract Rate for each calendar day on which any principal is outstanding. The payment of any Periodic Installment after its due date shall result in a corresponding decrease in the portion of the Periodic Installment credited to the remaining unpaid principal balance. The payment of any Periodic Installment prior to its due date shall result in a corresponding increase in the portion of the Periodic Installment credited to the remaining unpaid principal balance.

All amounts due hereunder and under the other Debt Documents are payable in the lawful currency of the United States of America. Maker hereby expressly authorizes Payee to insert the date value is actually given in the blank space on the face hereof and on all related documents pertaining hereto.

This Note may be secured by a security agreement, chattel mortgage, pledge agreement or like instrument (each of which is hereinafter called a "Security Agreement", and collectively with any other document or agreement related thereto or to this Note, the "Debt Documents").

Time is of the essence hereof. If Payee does not receive from Maker payment in full of any Periodic Installment or any other sum due under this Note or any other Debt Document is not received within fifteen (15) days after its due date, Maker agrees to pay a late fee equal to five percent (5%) on such late Periodic Installment or other sum, but not exceeding any lawful maximum. Such late fee will be immediately due and payable, and is in addition to any other reasonable costs, fees and expenses that Maker may owe as a result of such late payment. Additionally, if an Event of Default or default (as such terms are defined and/or used in the Security Agreement) has occurred, then the entire principal sum remaining unpaid, together with all accrued interest thereon and any other sum payable under this Note or any other Debt Document, at the election of Payee, shall immediately become due and payable, with interest thereon at the lesser of five percent (5%) per annum or the highest rate not prohibited by applicable law from the date of such accelerated maturity until paid (both before and after any judgment). The application of such 5% interest rate shall not be interpreted or deemed to extend any cure period set forth in this Note or any other Debt Document, cure any default or otherwise limit Payee's right or remedies hereunder or under any Debt Document.

Maker may prepay in full, but not in part, all outstanding amounts hereunder before they are due on any scheduled Payment Date upon at least thirty (30) days' prior written notice to Payee. Payee is authorized and entitled to apply any amounts paid by Maker as a prepayment of indebtedness to delinquent interest or other amounts due and owing from Maker to Payee hereunder and under any other Debt Documents before application of such funds to principal outstanding hereunder.

If Maker makes a prepayment of this Note for any reason, Maker shall pay irrevocably and in full to Payee (i) all outstanding principal amounts, (ii) all accrued interest, (iii) the Prepayment Fee (as defined below) and (iv) any and all other amounts due hereunder or under the other Debt Documents. Maker specifically acknowledges that, to the fullest extent allowed by applicable law, it shall be liable for the Prepayment Fee on any acceleration hereof or under the other Debt

Documents. In the event of an acceleration hereof or under the other Debt Documents, the Prepayment Fee shall be determined as if (a) Maker prepaid this Note in full immediately before such acceleration and (b) the prepayment notice referred to above was received by Payee thirty (30) days prior to such date.

For purposes hereof, "**Prepayment Fee**" shall be an amount equal to the following percentage of remaining principal balance for prepayments occurring in the indicated period: three percent (3%) (for prepayments occurring prior to the first anniversary of the date hereof), two percent (2%) (for prepayments occurring on and after the first anniversary of the date hereof but prior to the second anniversary of the date hereof), one percent (1%) (for prepayments occurring on and after the second anniversary of the date hereof but prior to the third anniversary of the date hereof) and zero percent (0%) (for prepayments occurring any time thereafter).

It is the intention of the parties hereto to comply with the applicable usury laws; accordingly, it is agreed that, notwithstanding any provision to the contrary in this Note or any other Debt Document, in no event shall this Note or any other Debt Document require the payment or permit the collection of interest in excess of the maximum amount permitted by applicable law. If any such excess interest is contracted for, charged or received under this Note or any other Debt Document, or if all of the principal balance shall be prepaid, so that under any of such circumstances the amount of interest contracted for, charged or received under this Note or any other Debt Document on the principal balance shall exceed the maximum amount of interest permitted by applicable law, then in such event: (a) the provisions of this paragraph shall govern and control, (b) neither Maker nor any other person or entity now or hereafter liable for the payment hereof shall be obligated to pay the amount of such interest to the extent that it is in excess of the maximum amount of interest permitted by applicable law, (c) any such excess which may have been collected shall be either applied as a credit against the then unpaid principal balance or refunded to Maker, at the option of Payee, and (d) the effective rate of interest shall be automatically reduced to the maximum lawful contract rate allowed under applicable law as now or hereafter construed by the courts having jurisdiction thereof. It is further agreed that without limitation of the foregoing, all calculations of the rate of interest contracted for, charged or received under this Note or any Debt Document which are made for the purpose of determining whether such rate exceeds the maximum lawful contract rate, shall be made, to the extent permitted by applicable law, by amortizing, prorating, allocating and spreading in equal parts during the period of the full stated term of the indebtedness evidenced hereby, all interest at any time contracted for, charged or received from Maker or otherwise by Payee in connection with such indebtedness; provided, however, that if any applicable state law is amended or the law of the United States of America preempts any applicable state law, so that it becomes lawful for Payee to receive a greater interest per annum rate than is presently allowed. Maker agrees that, on the effective date of such amendment or preemption, as the case may be, the lawful maximum hereunder shall be increased to the maximum interest per annum rate allowed by the amended state law or the law of the United States of America.

Maker hereby consents to any and all extensions of time, renewals, waivers or modifications of, and all substitutions or releases of, security or of any party primarily or secondarily liable on this Note or any other Debt Document or any term and provision of either, which may be made, granted or consented to by Payee, and agrees that suit may be brought and maintained against Maker and/or any and all sureties, endorsers, guarantors or any others who may at any time become liable for payments and performance under this Note and any other Debt Documents (each such person, other than Maker, an "Obligor"), at the election of Payee without joinder of any other as a party thereto, and that Payee shall not be required first to foreclose, proceed against, or exhaust any security hereof in order to enforce payment of this Note. Maker hereby waives presentment, demand for payment, notice of nonpayment, protest, notice of protest, notice of dishonor, and all other notices in connection herewith, as well as filing of suit (if permitted by law) and diligence in collecting this Note or enforcing any of the security hereof, and agrees to pay (if permitted by law) all expenses incurred in collection, including Payee's reasonable attorneys' fees.

THIS NOTE SHALL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE LAWS OF THE STATE OF CONNECTICUT.

MAKER IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE STATE OF CONNECTICUT TO HEAR AND DETERMINE ANY SUIT, ACTION OR PROCEEDING AND TO SETTLE ANY DISPUTES, WHICH MAY ARISE OUT OF OR IN CONNECTION HEREWITH AND WITH THE DEBT DOCUMENTS (COLLECTIVELY, THE "PROCEEDINGS"), AND MAKER FURTHER IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE TO REMOVE ANY SUCH PROCEEDINGS FROM ANY SUCH COURT (EVEN IF REMOVAL IS SOUGHT TO ANOTHER OF THE ABOVE-NAMED COURTS). MAKER IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MIGHT NOW OR HEREAFTER HAVE TO THE ABOVE-NAMED COURTS BEING NOMINATED AS THE EXCLUSIVE FORUM TO HEAR AND DETERMINE ANY SUCH PROCEEDINGS AND AGREES NOT TO CLAIM THAT IT IS NOT PERSONALLY SUBJECT TO THE JURISDICTION OF THE ABOVE-NAMED COURTS FOR ANY REASON WHATSOEVER, THAT IT OR ITS PROPERTY IS IMMUNE FROM LEGAL PROCESS FOR ANY REASON WHATSOEVER, THAT ANY SUCH COURT IS NOT A CONVENIENT OR APPROPRIATE FORUM IN EACH CASE WHETHER ON THE GROUNDS OF VENUE OR FORUM NON-CONVENIENS OR OTHERWISE. MAKER ACKNOWLEDGES THAT BRINGING ANY SUCH SUIT, ACTION OR PROCEEDING IN ANY COURT OTHER THAN THE COURTS SET FORTH ABOVE WILL CAUSE IRREPARABLE HARM TO PAYEE WHICH COULD NOT ADEQUATELY BE COMPENSATED BY MONETARY DAMAGES, AND, AS SUCH, MAKER AGREES THAT, IN ADDITION TO ANY OF THE REMEDIES TO WHICH PAYEE MAY BE ENTITLED AT LAW OR IN EQUITY, PAYEE WILL BE ENTITLED TO AN INJUNCTION OR INJUNCTIONS (WITHOUT THE POSTING OF ANY BOND AND WITHOUT PROOF OF ACTUAL DAMAGES) TO ENJOIN THE PROSECUTION OF ANY SUCH PROCEEDINGS IN ANY OTHER COURT. Notwithstanding the foregoing, each of Maker and Payee shall have the right to apply to a court of competent jurisdiction in the United States of America or abroad for equitable relief as is necessary to preserve, protect and enforce its respective rights under this Note and any other Debt Document, including, but not limited to orders of attachment or injunction necessary to maintain the status quo pending litigation or to enforce judgments against Maker, any Obligor or the collateral pledged to Payee pursuant to any Debt Document or to gain possession of such collateral. MAKER HEREBY UNCONDITIONALLY WAIVES ITS RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF, DIRECTLY OR INDIRECTLY, THIS NOTE, ANY DEBT DOCUMENTS, ANY DEALINGS BETWEEN MAKER AND PAYEE RELATING TO THE SUBJECT MATTER OF THIS TRANSACTION OR ANY RELATED TRANSACTIONS, AND/OR THE RELATIONSHIP THAT IS BEING ESTABLISHED BETWEEN MAKER AND PAYEE. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT (INCLUDING, WITHOUT LIMITATION, CONTRACT CLAIMS, TORT

CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS.) THIS WAIVER IS IRREVOCABLE MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND THE WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS NOTE, ANY DEBT DOCUMENTS. OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THIS TRANSACTION OR ANY RELATED TRANSACTION. IN THE EVENT OF LITIGATION. THIS NOTE MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

This Note and the other Debt Documents constitute the entire agreement of Maker and Payee with respect to the subject matter hereof and supersede all prior understandings, agreements and representations, express or implied.

No variation or modification of this Note, or any waiver of any of its provisions or conditions, shall be valid unless in writing and signed by an authorized representative of Maker and Payee. Any such waiver, consent, modification or change shall be effective only in the specific instance and for the specific purpose given.

Payment Authorization

Company Name

Payee is hereby directed and authorized by I	Maker to advance and/or apply the	proceeds of the loan as evide	nced by this Note to the	following parties in the
stipulated amounts as set forth below:				

Address Any provision in this Note or any of the other Debt Documents which is in conflict with any statute, law or applicable rule shall be deemed omitted, modified

or altered to conform thereto.

PUMPCO, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: VP and Treasurer Federal Tax ID #: 742196341

Address: 1209 South Main Street, Giddings, Lee

County, TX 78942

MASTER SECURITY AGREEMENT

dated as of May 30, 2008

THIS MASTER SECURITY AGREEMENT (as amended, supplemented or otherwise modified from time to time, this "Agreement") is between General Electric Capital Corporation (together with its successors and assigns, if any, "Secured Party") and PUMPCO, INC. ("Debtor"). Secured Party has an office at 11175 Cicero Drive, Suite 600, Alpharetta, GA 30022. Debtor is a corporation organized and existing under the laws of the state of Texas. Debtor's mailing address and chief executive office is 1209 South Main Street, Giddings, TX 78942.

1. CREATION AND GRANT OF SECURITY INTEREST.

Debtor grants to Secured Party, its successors and assigns, a security interest in and against (A) all property listed on any collateral schedule now or in the future annexed hereto and made a part hereof (each a "Collateral Schedule", and collectively, the "Collateral Schedule"), and (B) all other equipment, vehicles, and titled vehicles owned by Debtor at the date of this Master Security Agreement as such date is shown above, and in the case of (A) and (B), in and against any and all additions, attachments, accessories and accessions to such property, all substitutions, replacements or exchanges therefor, and all insurance and/or other proceeds thereof (all of the foregoing being hereinafter individually and collectively referred to as the "Collateral"). This security interest is given to secure the payment and performance of all debts, obligations and liabilities of any kind whatsoever of Debtor to Secured Party, now existing or arising in the future, including but not limited to the payment and performance of certain Promissory Notes from time to time identified on any Collateral Schedule (collectively "Notes" and each a "Note"), and any renewals, extensions and modifications of such debts, obligations and liabilities (such Notes, debts, obligations and liabilities are called the "Indebtedness").

2. REPRESENTATIONS, WARRANTIES AND COVENANTS OF DEBTOR.

Debtor represents, warrants and covenants as of the date of this Agreement and as of the date of each Collateral Schedule that:

- (a) Debtor's exact legal name is as set forth in the preamble of this Agreement and Debtor is, and will remain, validly existing and in good standing under the laws of the state of its formation (specified in the preamble of this Agreement). Debtor has its chief executive office at the location specified in the preamble, and will not change its chief executive office without the prior written notice to the Secured Party, and is, and will remain, duly qualified and licensed in every jurisdiction wherever necessary to carry on its business and operations, including the jurisdiction(s) where the Collateral is or is to be located, except where the failure to be so qualified would not have a materially adverse effect;
- (b) Debtor has adequate power and capacity to enter into, and to perform its obligations under this Agreement, each Note and any other documents evidencing, or given in connection with, any of the Indebtedness (all of the foregoing are called the "Debt Documents");
- (c) This Agreement and the other Debt Documents have been duly authorized, executed and delivered by Debtor and constitute legal, valid and binding agreements enforceable in accordance with their terms, except to the extent that the enforcement of remedies may be limited under applicable bankruptcy and insolvency laws;
- (d) No approval, consent or withholding of objections is required from, and no notice is required to be given to, any governmental authority or instrumentality, or any other person or entity, with respect to the entry into, or performance by Debtor of any of the Debt Documents, except any already duly obtained, other than (A) those that would not (i) have a material adverse effect in the property, assets, business, operations or financial conditions of the Debtor, or (ii) materially impair the right or ability of the Debtor to carry on its operations substantially as now conducted or anticipated to be conducted in the future, or its ability to perform its obligations hereunder or under any Note, or (iii) materially impair the rights of Secured Party hereunder or under any Debt Document or in and to the Collateral and (B) those that have been, or will be prior to the closing date hereof or under any Collateral Schedule or Note duly obtained or made;
- (e) The entry into, and performance by, Debtor of the Debt Documents will not (i) violate any of the organizational documents of Debtor or any judgment, order, law or regulation applicable to Debtor, or (ii) result in any breach of or constitute a default under any contract to which Debtor is a party, or result in the creation of any lien, claim or encumbrance on any of Debtor's property (except for liens in favor of Secured Party) pursuant to any indenture, mortgage, deed of trust, bank loan, credit agreement, or other agreement or instrument to which Debtor is a party, other than (A) those that would not (i) have a

material adverse effect in the property, assets, business, operations or financial conditions of the Debtor, or (ii) materially impair the right or ability of the Debtor to carry on its operations substantially as now conducted or anticipated to be conducted in the future, or its ability to perform its obligations hereunder or under any Note, or (iii) materially impair the rights of Secured Party hereunder or under any Debt Document or in and to the Collateral and (B) those that have been duly waived, as applicable, prior to the closing date hereof or under any Collateral Schedule or Note;

- (f) There are no suits or proceedings pending in court or before any commission, board or other administrative agency against or affecting Debtor which could, in the aggregate, have a material adverse effect on Debtor, its business or operations, or its ability to perform its obligations under the Debt Documents, nor does Debtor have reason to believe that any such suits or proceedings are threatened;
- (g) All financial statements delivered to Secured Party in connection with the Indebtedness have been prepared in accordance with generally accepted accounting principles, and since the date of the most recent financial statement, there has been no material adverse change in Debtor's financial condition;
 - (h) The Collateral is, and will remain personal property, and is not, and will not be, used by Debtor for personal, family or household purposes;
 - (i) The Collateral is, and will remain, in good condition and repair (normal wear and tear excepted), and Debtor will not be negligent in its care and use;
- (j) Debtor is, and will remain, the sole and lawful owner, and in possession of, the Collateral, and has the sole right and lawful authority to grant the security interest described in this Agreement;
- (k) The Collateral is, and will remain, free and clear of all liens, claims and encumbrances of any kind whatsoever, except for (i) liens in favor of Secured Party, (ii) liens for taxes assessments and other governmental charges or levies not yet due or for taxes assessments and other governmental charges or levies being contested in good faith and which do not involve, in the reasonable judgment of Secured Party, any risk of the sale, forfeiture or loss of any of the Collateral, and (iii) inchoate materialmen's, mechanic's, repairmen's, carrier's, warehousemen, and similar liens arising by operation of law in the normal course of business for amounts which are not delinquent (all of such liens are called "Permitted Liens"); and
- (l) Debtor is and will remain in (A) material compliance (solely to the extent applicable to it) with all laws and regulations applicable to it and (B) in full compliance (solely to the extent applicable to it) with all laws and regulations ensuring that no person who owns a controlling interest in or otherwise controls Debtor is or shall be (Y) listed on the Specially Designated Nationals and Blocked Person List maintained by the Office of Foreign Assets Control ("OFAC"), Department of the Treasury, and/or any other similar lists maintained by OFAC pursuant to any authorizing statute, Executive Order or regulation or (Z) a person designated under Section 1(b), (c) or (d) of Executive Order No. 13224 (September 23, 2001), any related enabling legislation or any other similar Executive Orders, and (ii) compliance with all applicable Bank Secrecy Act ("BSA") laws, regulations and government guidance on BSA compliance and on the prevention and detection of money laundering violations.

3. COLLATERAL.

- (a) Until the declaration of any Event of Default, Debtor shall remain in possession of the Collateral; except that Secured Party shall have the right to possess (i) any chattel paper or instrument that constitutes a part of the Collateral, and (ii) any other Collateral in which Secured Party's security interest may be perfected only by possession. Secured Party may inspect any of the Collateral during normal business hours after giving Debtor reasonable prior notice. If Secured Party asks, Debtor will promptly notify Secured Party in writing of the location of any Collateral.
- (b) Debtor shall (i) use the Collateral only in its trade or business, (ii) maintain all of the Collateral in good operating order and repair, normal wear and tear excepted, (iii) use and maintain the Collateral only in compliance with manufacturers recommendations and all applicable laws, and (iv) keep all of the Collateral free and clear of all liens, claims and encumbrances (except for Permitted Liens).
- (c) Secured Party does not authorize and Debtor agrees it shall not (i) part with possession of any of the Collateral (except to Secured Party or for maintenance and repair), (ii) remove any of the Collateral from the continental United States, or (iii) sell, rent, lease, mortgage, license, grant a security interest in or otherwise transfer or encumber (except for Permitted Liens) any of the Collateral.
- (d) Debtor shall report and pay promptly when due all taxes, license fees, assessments and public and private charges levied or assessed on any of the Collateral, on its use, operation, purchase, ownership, delivery, leasing or possession thereof, or on this Agreement or any of the other Debt Documents (or any receipts hereunder and thereunder), by any governmental entity or taxing authority during or related to the term of this Agreement, or to any other period during which Debtor had use or possession of the Collateral, including, without limitation, all license and registration fees, and all sales, use, personal property, excise, gross receipts, franchise, stamp or other taxes, imposts, duties and charges, together with any penalties, fines or interest thereon (collectively "Taxes"). Debtor shall have no liability for Taxes imposed by the United States of America or any state or political subdivision thereof which are on or measured by the net

income of Secured Party. Debtor shall promptly reimburse Secured Party (on an after-tax basis) for any Taxes charged to or assessed against or paid by Secured Party. Debtor shall send Secured Party a copy of each report or return and evidence of Debtor's payment of Taxes upon request by Secured Party. At its option, Secured Party may discharge taxes, liens, security interests or other encumbrances at any time levied or placed on the Collateral and may pay for the maintenance, insurance and preservation of the Collateral and effect compliance with the terms of this Agreement or any of the other Debt Documents. Debtor agrees to reimburse Secured Party, on demand, all reasonable costs and expenses incurred by Secured Party in connection with such payment or performance and agrees that such reimbursement obligation shall constitute indebtedness.

- (e) Debtor shall, at all times, keep accurate and complete records of the Collateral, and Secured Party shall have the right to inspect and make copies of all of Debtor's books and records relating to the Collateral during normal business hours, after giving Debtor reasonable prior notice. Notwithstanding anything herein to the contrary, the Secured Party shall conduct no more than 3 inspections per year of such books, records and of the Collateral itself and the Debtor shall not be responsible for costs and expenses all in excess of \$1000.00 per annum (in each case, so long as there is not a continuing Event of Default, in which case such limitations shall not apply).
- (f) Debtor agrees and acknowledges that any third person who may at any time possess all or any portion of the Collateral shall be deemed to hold, and shall hold, the Collateral as the agent of, and as pledge holder for, Secured Party. Secured Party may at any time give notice to any third person described in the preceding sentence that such third person is holding the Collateral as the agent of, and as pledge holder for, Secured Party.

4. INSURANCE.

- (a) Debtor shall at all times bear the entire risk of any loss, theft, damage to, or destruction of, any of the Collateral from any cause whatsoever.
- (b) Debtor agrees, at its own expense, to keep the Collateral insured with companies reasonably acceptable to Secured Parties for such amounts and against such hazards as Secured Party may require, including, but not limited to, all risks physical damage insurance for the Collateral itself, including, but not limited to, loss or damage by fire and extended coverage perils, theft, burglary, and for any or all Collateral which are vehicles, for risk of loss by collision. The physical insurance coverage shall be in an amount no less than the full replacement value of the Collateral, and deductible amounts, insurers and policies shall be acceptable to Secured Party. Debtor shall deliver to Secured Party policies or certificates of insurance evidencing such coverage. Each policy shall name Secured Party as loss payee, shall provide for coverage to Secured Party regardless of the breach by Debtor of any warranty or representation made therein, shall not be subject to co-insurance, and shall provide that coverage may not be canceled or altered by the insurer except upon thirty (30) days' prior written notice to Secured Party. Upon an Event of Default that has occurred and is continuing, Debtor irrevocably appoints Secured Party as its attorney-infact to make proof of loss, claim for insurance and adjustments with insurers, and to receive payment of and execute or endorse all documents, checks or drafts in connection with insurance payments. Debtor may not make adjustments with insurers except with Secured Party's prior written consent. Proceeds of insurance shall be applied, at the option of Secured Party, to repair or replace the Collateral or to reduce any of the Indebtedness under the Debt Documents.

5. REPORTS.

- (a) Debtor shall promptly notify Secured Party (i) at least thirty (30) days' prior to any change in the name of Debtor, (ii) at least sixty (60) days' prior to any change in the state of its incorporation, organization or registration, (iii) at least thirty (30) days' prior to any relocation of its chief executive offices, (iv) at least thirty (30) days' prior to any permanent or indefinite relocation of any of the Collateral from the location(s) specified in its applicable Collateral Schedule, (v) promptly upon any of the Collateral being lost, stolen, missing, destroyed, materially damaged or worn out, or (vi) promptly upon Debtor becoming aware of any lien, claim or encumbrance other than Permitted Liens attaching to or being made against any of the Collateral.
- (b) Debtor will deliver to Secured Party Debtor's complete financial statements, certified by a recognized firm of certified public accountants, within one hundred and twenty (120) days of the close of each fiscal year of Debtor. If Secured Party requests, Debtor will deliver to Secured Party copies of Debtor's quarterly financial reports certified by Debtor's chief financial officer, within ninety (90) days after the close of each of Debtor's fiscal quarter. Debtor will deliver to Secured Party copies of all Forms 10-K and I0-Q, if any, within 30 days after the dates on which they are filed with the Securities and Exchange Commission.

6. FURTHER ASSURANCES.

(a) Debtor shall, upon request of Secured Party, furnish to Secured Party such further information, execute and deliver to Secured Party such documents and instruments (including, without limitation, Uniform Commercial Code financing statements) and shall do such other acts and things as Secured Party may at any time reasonably request relating to the perfection or protection of the security interest created by this Agreement or for the purpose of carrying out the intent

of this Agreement. Without limiting the foregoing, Debtor shall cooperate and do all acts deemed reasonably necessary or advisable by Secured Party to continue in Secured Party a perfected first security interest in the Collateral, and shall obtain and furnish to Secured Party any subordinations, releases, landlord waivers, lessor waivers, mortgagee waivers, or control agreements, and similar documents as may be from time to time reasonably requested by, and in form and substance reasonably satisfactory to, Secured Party.

- (b) Debtor authorizes Secured Party to file a financing statement and amendments thereto describing the Collateral and containing any other information required by the applicable Uniform Commercial Code. Debtor irrevocably grants to Secured Party the power to sign Debtor's name and generally to act on behalf of Debtor to execute and file applications for title, transfers of title, financing statements, notices of lien and other documents pertaining to any or all of the Collateral; this power is coupled with Secured Party's interest in the Collateral. Debtor shall, if any certificate of title be required or permitted by law for any of the Collateral, obtain and promptly deliver to Secured Party such certificate showing the lien of this Agreement with respect to the Collateral. Debtor ratifies its prior authorization for Secured Party to file financing statements and amendments thereto describing the Collateral and containing any other information required by the Uniform Commercial Code if filed prior to the date hereof.
- (c) Debtor shall indemnify and save on a net after-tax basis Secured Party and its affiliates and all of Secured Party's and such affiliates' respective directors, shareholders, officers, employees, agents, predecessors, attorneys-in-fact, lawyers, successors and assigns (each an "Indemnitee"), harmless from and against all claims, costs, expenses (including reasonable legal fees), demands, suits, damages and liabilities of any kind and nature whatsoever, including without limitation personal injury, death and property damage claims arising in tort or otherwise, under any legal theory including but not limited to strict liability (including claims involving or alleging environmental damage, criminal acts, hijacking, acts of terrorism or similar acts, product liability or strict or absolute liability in tort, latent and other defects (whether or not discoverable), or for patent, trademark or copyright infringement, collectively, "Claims"), that may be imposed on, incurred by or asserted against any Indemnitee whether or not such Indemnitee shall also be indemnified as to any such Claim by any other Person in any way relating to, arising out of or in connection with (i) the Debt Documents, including, without limitation, the execution, delivery, breach (including any Event of Default), enforcement, performance or administration of the Debt Documents and (ii) the Collateral, including, without limitation, the perfection, maintenance, protection or realization upon the Collateral or any other security for the Indebtedness, and the manufacture, inspection, construction, purchase, acceptance, rejection, ownership, management, pooling, interchange, chartering, titling or re-titling, delivery, lease, sublease, possession, use, operation, maintenance, condition, registration or re-registration, sale, removal, repossession, storage or other disposition of the Collateral or any part thereof or any accident in connection therewith. Notwithstanding the foregoing, Debtor shall not be required to indemnify an Indemnitee for an

7. DEFAULT AND REMEDIES.

- (a) Debtor shall be in default under this Agreement and each of the other Debt Documents upon the occurrence of any of the following (each an "Event of Default"):
 - (i) Debtor fails to pay within seven (7) business days after its due date any installment or other amount due under any of the Debt Documents;
- (ii) Debtor, without the prior written consent of Secured Party, attempts to or does sell all or any fractional interest in, rent, lease, license, charter, mortgage, assign, grant a lien on or security interest in, or otherwise transfer or encumber (except for Permitted Liens) any of the Collateral or any part thereof:
 - (iii) Debtor breaches any of its insurance obligations under Section 4, without the prior written consent of Secured Party in its sole discretion;
- (iv) Debtor breaches any of its other obligations under any of the Debt Documents (other than those described under Section 7(a)(i) through (iii) above) and fails to cure that breach within thirty (30) days after written notice from Secured Party;
- (v) Any warranty, representation or statement made by Debtor or any guarantor or surety for the Indebtedness (each a "Guarantor", and collectively, the "Guarantors") in any of the Debt Documents or otherwise in connection with any of the Indebtedness shall be false or misleading in any material respect;
- (vi) Any of the Collateral is subjected to attachment, execution, levy, seizure or confiscation in any legal proceeding or otherwise, or if any legal or administrative proceeding is commenced against Debtor or any of the Collateral, which in the good faith judgment of Secured Party subjects any of the Collateral to a material risk of attachment, execution, levy, seizure or confiscation and no bond is posted or protective order obtained to negate such risk;
 - (vii) Debtor or any Guarantor breaches or is in default under any other agreement between Debtor or such Guarantor and Secured Party;

- (viii) Debtor or any Guarantor dissolves, terminates its existence, becomes insolvent or ceases to do business as a going concern;
- (ix) Intentionally Omitted.
- (x) A receiver, custodian or trustee is appointed for all or of any part of the property of Debtor or any Guarantor, or Debtor or any Guarantor makes any assignment for the benefit of creditors, or Debtor or any Guarantor by any act or omission shall indicate its consent to, approval of or acquiescence in any such appointment of a custodian, receiver or trustee;
- (xi) Debtor or any Guarantor files a petition under any bankruptcy, insolvency or similar law, or in the event an involuntary petition is filed against Debtor or any Guarantor and such petition is not dismissed within sixty (60) days, or Debtor or any Guarantor by any act or omission shall indicate its consent to, approval of or acquiescence in any such petition, application, proceeding, order for relief;
- (xii) Debtor or any Guarantor improperly files, or cause to be filed, an amendment or termination statement relating to a filed financing statement describing the Collateral, without the prior written consent of Secured Party in its sole discretion;
- (xiii) Any Guarantor revokes or attempts to revoke its obligations under any Debt Documents to which it is a party or fails to observe or perform any covenant, condition or agreement to be performed under such Debt Document to which it is a party;
- (xiv) Debtor or any Guarantor defaults under any other material obligations (other than set forth in this Section 7) for (A) borrowed money, (B) the deferred purchase price of property or (C) payments due under any lease agreement;
- (xv) Debtor or any Guarantor is declared in default under any contract or obligation requiring the payment of money in an original principal amount greater than \$50,000; or
 - (xvi) There is any dissolution, termination of existence, merger, consolidation or change in controlling ownership of Debtor or any Guarantor.
- (b) Upon the occurrence of any Event of Default, Secured Party, at its option, may declare any or all of the Indebtedness to be immediately due and payable, without demand or notice to Debtor or any Guarantor. The accelerated obligations and liabilities shall bear interest from the occurrence of the Event of Default (both before and after any judgment) until paid in full at a per annum rate equal to the lower of eighteen percent (18%) or the maximum rate not prohibited by applicable law (the "Per Diem Interest Rate"). The application of such Per Diem Interest Rate shall not be interpreted or deemed to extend any cure period set forth herein, cure any default or otherwise limit Secured Party's right or remedies hereunder. Notwithstanding anything to the contrary contained herein, in no event shall this Agreement require the payment or permit the collection of amounts in excess of the maximum permitted by applicable law.
- (c) After default, Secured Party shall have all of the rights and remedies of a secured party under the Uniform Commercial Code, and under any other applicable law. Without limiting the foregoing, Secured Party shall have the right to (i) notify any account debtor of Debtor or any obligor on any instrument which constitutes part of the Collateral to make payment to Secured Party, (ii) with or without legal process, enter any premises where the Collateral may be and take possession of and remove the Collateral from the premises or store it on the premises, (iii) sell the Collateral at public or private sale, in whole or in part, and have the right to bid and purchase at said sale, or (iv) lease or otherwise dispose of all or part of the Collateral, applying proceeds from such disposition to the obligations then in default. If requested by Secured Party, Debtor shall promptly assemble the Collateral and make it available to Secured Party at a place to be designated by Secured Party which is reasonably convenient to both parties. Secured Party may also render any or all of the Collateral unusable at Debtor's premises and may dispose of such Collateral on such premises without liability for rent or costs. Any notice that Secured Party is required to give to Debtor under the Uniform Commercial Code of the time and place of any public sale or the time after which any private sale or other intended disposition of the Collateral is to be made shall be deemed to constitute reasonable notice if such notice is given to the last known address of Debtor at least seven (7) business days prior to such action.
- (d) Secured Party shall have the right to apply any amounts collected from Debtor or Guarantor pursuant to this Section 7 or under any other Debt Document in the following order of priorities: (i) to pay all of Secured Party's costs, charges and expenses incurred in enforcing its rights under this Agreement or in taking, removing, holding, repairing, refurbishing, selling, leasing or otherwise disposing of the Collateral; then, (ii) to pay any and all late fees, per diem fees, other such charges due hereunder, any and all interest due hereunder and all amounts owing pursuant to any indemnity claims; then (iii) to pay all principal due hereunder; then (iv) to pay all other amounts due and owing to Secured Party under any of the Debt Documents.
- (e) Secured Party shall have the right to any proceeds of sale, lease or other disposition of the Collateral, if any, and shall have the right to apply same in the following order of priorities: (i) to pay all of Secured Party's costs, charges and expenses incurred in enforcing its rights under this Agreement or in taking.

removing, holding, repairing, refurbishing, selling, leasing or otherwise disposing of the Collateral; then, (ii) to pay any and all late fees, per diem fees, other such charges due hereunder, any and all interest due hereunder and any amounts owing pursuant to any indemnity claims; then (iii) to pay all principal due hereunder; then (iv) to pay all other amounts due and owing to Secured Party under any of the Debt Documents; then (v) any surplus shall be refunded to Debtor. Debtor shall pay any deficiency in (i), (iii) and (iv) immediately upon demand.

- (f) Debtor agrees to pay all reasonable attorneys' fees and other costs incurred by Secured Party in connection with the enforcement, assertion, defense or preservation of Secured Party's rights and remedies under this Agreement, or if prohibited by law, such lesser sum as may be permitted. Debtor further agrees that such fees and costs shall constitute Indebtedness.
- (g) Secured Party's rights and remedies under this Agreement or otherwise arising are cumulative and nonexclusive of any other rights and remedies that Secured Party may have under any other agreement or at law or in equity and may be exercised individually or concurrently, any or all thereof may be exercised instead of or in addition to each other or any remedies at law, in equity, or under statute. Neither the failure nor any delay on the part of Secured Party to exercise any right, power or privilege under this Agreement shall operate as a waiver, nor shall any single or partial exercise of any right, power or privilege preclude any other or further exercise of that or any other right, power or privilege. SECURED PARTY SHALL NOT BE DEEMED TO HAVE WAIVED ANY OF ITS RIGHTS UNDER THIS AGREEMENT OR UNDER ANY OTHER AGREEMENT, INSTRUMENT OR PAPER SIGNED BY DEBTOR UNLESS SUCH WAIVER IS EXPRESSED IN WRITING AND SIGNED BY SECURED PARTY. A waiver on any one occasion shall not be construed as a bar to or waiver of any right or remedy on any future occasion. Except as provided in Section 7(c) above, Debtor waives notice of sale or other disposition (and the time and place thereof), and the manner and place of any advertising, and any other notice required to be given under the Uniform Commercial Code. Secured Party shall have no obligation to marshal any of the Collateral.
- (h) Secured Party hereby acknowledges that Debtor shall not be in default hereunder, nor shall an Event of Default be deemed to exist unless any applicable grace periods have expired pursuant hereto.

8. MISCELLANEOUS.

- (a) This Agreement, any Note and/or any of the other Debt Documents may be assigned, in whole or in part, by Secured Party without notice to Debtor, and Debtor hereby waives and agrees not to assert against any such assignee, or assignee's assigns, any defense, set-off, recoupment, claim or counterclaim which Debtor has or may at any time have against Secured Party for any reason whatsoever. Debtor agrees that if Debtor receives written notice of an assignment from Secured Party, Debtor will pay all amounts payable under any assigned Debt Documents to such assignee or as instructed by Secured Party. Debtor also agrees to confirm in writing receipt of the notice of assignment as may be reasonably requested by Secured Party or assignee.
- (b) All notices to be given in connection with this Agreement shall be in writing, shall be addressed to the parties at their respective addresses set forth in this Agreement (unless and until a different address may be specified in a written notice to the other party), and shall be deemed given (i) on the date of receipt if delivered in hand or by facsimile transmission, (ii) on the next business day after being sent by express mail, and (iii) on the fourth business day after being sent by regular, registered or certified mail. As used herein, the term "business day" shall mean and include any day other than Saturdays, Sundays, or other days on which commercial banks in New York, New York are required or authorized to be closed.
- (c) Secured Party may correct patent errors and fill in all blanks in this Agreement or in any Collateral Schedule consistent with the agreement of the parties.
- (d) Time is of the essence of this Agreement. This Agreement shall be binding, jointly and severally, upon all parties described as the "Debtor" and their respective heirs, executors, representatives, successors and assigns, and shall inure to the benefit of Secured Party, its successors and assigns.
 - (e) The unenforceability of any provisions hereof or of the Debt Documents shall not affect the validity of any other provision hereof or thereof.
- (f) Debtor hereby acknowledges and agrees that Secured Party reserves the right to impose reasonable fees or charges for returned checks and certain optional services that Secured Party may offer or provide to Debtor during the term of this Agreement. Secured Party will notify Debtor the amount of the applicable fee or charge if Debtor requests such optional services. In addition, Secured Party may make available to Debtor a schedule of fees or charges for such optional services from time to time or upon demand, provided, however, that such fees and charges are subject to change in Secured Party's sole discretion without notice to Debtor.
- (g) This Agreement and its Collateral Schedules constitute the entire agreement between the parties with respect to the subject matter of this Agreement and supersede all prior understandings (whether written, verbal or implied) with respect to such subject matter. THIS AGREEMENT AND ITS COLLATERAL SCHEDULES SHALL NOT BE CHANGED OR TERMINATED ORALLY OR BY COURSE OF CONDUCT, BUT ONLY BY A WRITING SIGNED BY

BOTH PARTIES. Section headings contained in this Agreement have been included for convenience only, and shall not affect the construction or interpretation of this Agreement.

- (h) This Agreement shall continue in full force and effect until all of the Indebtedness has been indefeasibly paid in full to Secured Party or its assignee. The surrender, upon payment or otherwise, of any Note or any of the other documents evidencing any of the Indebtedness shall not affect the right of Secured Party to retain the Collateral for such other Indebtedness as may then exist or as it may be reasonably contemplated will exist in the future. This Agreement shall automatically be reinstated if Secured Party is ever required to return or restore the payment of all or any portion of the Indebtedness (all as though such payment had never been made).
- (i) DEBTOR AND SECURED PARTY HEREBY UNCONDITIONALLY WAIVES THEIR RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF, DIRECTLY OR INDIRECTLY, THIS AGREEMENT, ANY OF THE DEBT DOCUMENTS, ANY DEALINGS BETWEEN DEBTOR AND SECURED PARTY RELATING TO THE SUBJECT MATTER OF THIS TRANSACTION OR ANY RELATED TRANSACTIONS, AND/OR THE RELATIONSHIP THAT IS BEING ESTABLISHED BETWEEN DEBTOR AND SECURED PARTY. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT (INCLUDING, WITHOUT LIMITATION, CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS). THIS WAIVER IS IRREVOCABLE MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND THE WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT, ANY OF THE OTHER DEBT DOCUMENTS OR ANY OTHER DOCUMENTS RELATING TO THIS TRANSACTION OR ANY RELATED TRANSACTION. IN THE EVENT OF ANY LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.
- (j) This Agreement shall continue in full force and effect until all of the Indebtedness has been indefeasibly paid in full to Secured Party. This Agreement shall automatically be reinstated in the event that Secured Party is ever required to return or restore the payment of all or any portion of the Indebtedness (all as though such payment had never been made).
- (k) THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL IN ALL RESPECTS BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF CONNECTICUT (WITHOUT REGARD TO THE CONFLICT OF LAWS PRINCIPLES OF SUCH STATE), INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, REGARDLESS OF THE LOCATION OF THE COLLATERAL. DEBTOR IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE STATE OF CONNECTICUT TO HEAR AND DETERMINE ANY SUIT, ACTION OR PROCEEDING AND TO SETTLE ANY DISPUTES, WHICH MAY ARISE OUT OF OR IN CONNECTION HEREWITH AND WITH THE DEBT DOCUMENTS (COLLECTIVELY, THE "PROCEEDINGS"), AND DEBTOR FURTHER IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE TO REMOVE ANY SUCH PROCEEDINGS FROM ANY SUCH COURT (EVEN IF REMOVAL IS SOUGHT TO ANOTHER OF THE ABOVE-NAMED COURTS). DEBTOR IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MIGHT NOW OR HEREAFTER HAVE TO THE ABOVE-NAMED COURTS BEING NOMINATED AS THE EXCLUSIVE FORUM TO HEAR AND DETERMINE ANY SUCH PROCEEDINGS AND AGREES NOT TO CLAIM THAT IT IS NOT PERSONALLY SUBJECT TO THE JURISDICTION OF THE ABOVE-NAMED COURTS FOR ANY REASON WHATSOEVER, THAT IT OR ITS PROPERTY IS IMMUNE FROM LEGAL PROCESS FOR ANY REASON WHATSOEVER, THAT ANY SUCH COURT IS NOT A CONVENIENT OR APPROPRIATE FORUM IN EACH CASE WHETHER ON THE GROUNDS OF VENUE OR FORUM NON-CONVENIENS OR OTHERWISE. DEBTOR ACKNOWLEDGES THAT BRINGING ANY SUCH SUIT, ACTION OR PROCEEDING IN ANY COURT OTHER THAN THE COURTS SET FORTH ABOVE WILL CAUSE IRREPARABLE HARM TO SECURED PARTY WHICH COULD NOT ADEQUATELY BE COMPENSATED BY MONETARY DAMAGES, AND, AS SUCH, DEBTOR AGREES THAT, IN ADDITION TO ANY OF THE REMEDIES TO WHICH SECURED PARTY MAY BE ENTITLED AT LAW OR IN EQUITY, SECURED PARTY WILL BE ENTITLED TO AN INJUNCTION OR INJUNCTIONS (WITHOUT THE POSTING OF ANY BOND AND WITHOUT PROOF OF ACTUAL DAMAGES) TO ENJOIN THE PROSECUTION OF ANY SUCH PROCEEDINGS IN ANY OTHER COURT. Notwithstanding the foregoing, each of Debtor and Secured Party shall have the right to apply to a court of competent jurisdiction in the United States of America or abroad for equitable relief as is necessary to preserve, protect and enforce its respective rights under this Agreement and any other Debt Document, including, but not limited to orders of attachment or injunction necessary to maintain the status quo pending litigation or to enforce judgments against Debtor, any Guarantor or the collateral pledged to Secured Party pursuant to any Debt Document or to gain possession of such collateral.
 - (l) This Agreement and any amendments, waivers, consents or supplements hereto in connection herewith may be executed in any number of counterparts and by different parties hereto in separate counterparts, all of which taken together shall constitute one and the same instrument; signature pages may be detached from multiple separate counterparts and attached to a single counterpart so that all signature pages are physically

attached to the same document. Delivery of an executed signature page of this Agreement or any delivery contemplated hereby by facsimile or electronic transmission shall be as effective as delivery of a manually executed counterpart thereof.

(m) Secured Party's obligations hereunder and under the transactions contemplated hereby are contingent upon Debtor being directly controlled by Mastec, Inc.

IN WITNESS WHEREOF, Debtor and Secured Party, intending to be legally bound hereby, have duly executed this Agreement in one or more counterparts, each of which shall be deemed to be an original, as of the day and year first aforesaid.

SECURED PARTY:

DEBTOR:

General Electric Capital Corporation

PUMPCO, INC.

By: /s/ Jessica A. Ernst

By: /s/ C. Robert Campbell

Name: Jessica A. Ernst Title: Senior Risk Analyst Name: C. Robert Campbell
Title: Vice President & Treasurer

CORPORATE GUARANTY

Date: May 30, 2008

General Electric Capital Corporation 11175 Cicero Drive Suite 600 Alpharetta, GA 30022

To induce you to enter into, purchase or otherwise acquire, now or at any time hereafter, any promissory notes, security agreements, chattel mortgages, pledge agreements, conditional sale contracts, lease agreements, and/or any other documents or instruments evidencing, or relating to, any lease, loan, extension of credit or other financial accommodation (collectively "Account Documents" and each an "Account Document") to PUMPCO, Inc, a corporation organized and existing under the laws of the State of Texas ("Customer"), but without in any way binding you to do so, the undersigned, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, does hereby guarantee to you, your successors and assigns, the due regular and punctual payment of any sum or sums of money which the Customer may owe to you now or at any time hereafter, whether evidenced by an Account Document, on open account or otherwise, and whether it represents principal, interest, rent, late charges, indemnities, an original balance, an accelerated balance, liquidated damages, a balance reduced by partial payment, a deficiency after sale or other disposition of any leased equipment, collateral or security, or any other type of sum of any kind whatsoever that the Customer may owe to you now or at any time hereafter, and does hereby further guarantee to you, your successors and assigns, the due, regular and punctual performance of any other duty or obligation of any kind or character whatsoever that the Customer may owe to you now or at any time hereafter (all such payment and performance obligations being collectively referred to as "Obligations"). The undersigned does hereby further guarantee to pay upon demand all losses, costs, reasonable attorneys' fees and expenses which may be suffered by you by reason of Customer's default or default of the undersigned. As used in this Guaranty, "you" shall mean General Electric Capital Corporation and all its subsidiaries, parent entities, successors and assigns.

This Guaranty is a guaranty of prompt payment and performance (and not merely a guaranty of collection). Nothing herein shall require you to first seek or exhaust any remedy against the Customer, its successors and assigns, or any other person obligated with respect to the Obligations, or to first foreclose, exhaust or otherwise proceed against any leased equipment, collateral or security which may be given in connection with the Obligations. It is agreed that you may, upon any breach or default of the Customer, or at any time thereafter, make demand upon the undersigned and receive payment and performance of the Obligations, with or without notice or demand for payment or performance by the Customer, its successors or assigns, or any other person. Suit may be brought and maintained against the undersigned, at your election, without joinder of the Customer or any other person as parties thereto. The obligations of each signatory to the Guaranty, and each other guarantor of the Obligations, shall be joint and several.

The undersigned agrees that its obligations under this Guaranty shall be primary, absolute, continuing and unconditional, irrespective of and unaffected by any of the following actions or circumstances (regardless of any notice to or consent of the undersigned) and the undersigned hereby affirmatively and irrevocably waives as a defense to the payment or performance of its obligations hereunder each and every one of the following defenses: (a) the genuineness, validity, regularity and enforceability of the Account Documents or any other document; (b) any extension, renewal, amendment, change, waiver or other modification of the Account Documents or any other document; (c) the absence of, or delay in, any action to enforce the Account Documents, this Guaranty or any other document; (d) your failure or delay in obtaining any other guaranty of the Obligations (including, without limitation, your failure to obtain the signature of any other guarantor hereunder); (e) the release of, extension of time for payment or performance by, or any other indulgence granted to the Customer or any other person with respect to the Obligations by operation of law or otherwise; (f) the existence, value, condition, loss, subordination or release (with or without substitution) of, or failure to have title to or perfect and maintain a security interest in, or the time, place and manner of any sale or other disposition of any leased equipment, collateral or security given in connection with the Obligations, or any other impairment (whether intentional or negligent, by operation of law or otherwise) of the rights of the undersigned; (g) the Customer's voluntary or involuntary bankruptcy, assignment for the benefit of creditors, reorganization, or similar proceedings affecting the Customer or any of its assets; (h) any merger or consolidation of Customer, any change in control of Customer or any sale of all or substantially all of the assets of Customer; or (i) any other action or circumstances which might otherwise constitute a legal or

This Guaranty, the Account Documents and the Obligations may be assigned by you, without the consent of the undersigned. The undersigned agrees that if it receives written notice of an assignment from you, the undersigned will pay all amounts due hereunder to such assignee or as instructed by you. The undersigned also agrees to confirm in writing receipt of the notice of assignment as may be reasonably requested by assignee. The undersigned hereby waives and agrees not to assert against any such assignee any of the defenses set forth in the immediate preceding paragraph.

This Guaranty may be terminated upon delivery to you (at your address shown above) of a written termination notice from the undersigned. However, as to all Obligations (whether matured, unmatured, absolute, contingent or otherwise) incurred by the Customer prior to your receipt of such written termination notice (and regardless of any subsequent amendment, extension or other modification which may be made with respect to such Obligations), this Guaranty shall nevertheless continue and remain undischarged until all such Obligations are indefeasibly paid and performed in full.

The undersigned agrees that this Guaranty shall remain in full force and effect or be reinstated (as the case may be) if at any time payment or performance of any of the Obligations (or any part thereof) is rescinded, reduced or must otherwise be restored or returned by you, all as though such payment or performance had not been made. If, by reason of any bankruptcy, insolvency or similar laws affecting the rights of creditors, you shall be prohibited from exercising any of your rights or remedies against the Customer or any other person or against any property, then, as between you and the undersigned, such prohibition shall be of no force and effect, and you shall have the right to make demand upon, and receive payment from, the undersigned of all amounts and other sums that would be due to you upon a default with respect to the Obligations.

Notice of acceptance of this Guaranty and of any default by the Customer or any other person is hereby waived. Presentment, protest demand, and notice of protest, demand and dishonor of any of the Obligations, and the exercise of possessory, collection or other remedies for the Obligations, are hereby waived. The undersigned warrants that it has adequate means to obtain from the Customer on a continuing basis financial data and other information regarding the Customer and is not relying upon you to provide any such data or other information. Without limiting the foregoing, notice of adverse change in the Customer's financial condition or of any other fact which might materially increase the risk of the undersigned is also waived. All settlements, compromises, accounts stated and agreed balances made in good faith between the Customer, its successors or assigns, and you shall be binding upon and shall not affect the liability of the undersigned.

Payment of all amounts now or hereafter owed to the undersigned by the Customer or any other obligor for any of the Obligations is hereby subordinated in right of payment to the indefeasible payment in full to you of all Obligations and is hereby assigned to you as a security therefor. The undersigned hereby irrevocably and unconditionally waives and relinquishes all statutory, contractual, common law, equitable and all other claims against the Customer, any other obligor for any of the Obligations, any collateral therefor, or any other assets of the Customer or any such other obligor, for subrogation, reimbursement, exoneration, contribution, indemnification, setoff or other recourse in respect of sums paid or payable to you by the undersigned hereunder, and the undersigned hereby further irrevocably and unconditionally waives and relinquishes any and all other benefits which it might otherwise directly or indirectly receive or be entitled to receive by reason of any amounts paid by, or collected or due from, it, the Customer or any other obligor for any of the Obligations, or realized from any of their respective assets.

THE UNDERSIGNED HEREBY UNCONDITIONALLY WAIVES ITS RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF, DIRECTLY OR INDIRECTLY, THIS GUARANTY, THE OBLIGATIONS GUARANTEED HEREBY, ANY OF THE RELATED DOCUMENTS, ANY DEALINGS BETWEEN US RELATING TO THE SUBJECT MATTER HEREOF OR THEREOF, AND/OR THE RELATIONSHIP THAT IS BEING ESTABLISHED BETWEEN US. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT (INCLUDING, WITHOUT LIMITATION, CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS). THIS WAIVER IS IRREVOCABLE MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS GUARANTY, THE OBLIGATIONS GUARANTEED HEREBY, OR ANY RELATED DOCUMENTS. IN THE EVENT OF LITIGATION, THIS GUARANTY MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

As used in this Guaranty, the word "person" shall include any individual, corporation, partnership, joint venture, association, joint-stock company, trust, limited liability company, unincorporated organization, or any government or any political subdivision thereof.

This Guaranty is intended by the parties as a final expression of the guaranty of the undersigned and is also intended as a complete and exclusive statement of the terms thereof. No course of dealing, course of performance or trade usage, nor any paid evidence of any kind, shall be used to supplement or modify any of the terms hereof. Nor are there any conditions to the full effectiveness of this Guaranty. This Guaranty and each of its provisions may only be waived, modified, varied, released, terminated or surrendered, in whole or in part, by a duly authorized written instrument signed by you. No failure by you to exercise your rights hereunder shall give rise to any estoppel against you, or excuse the undersigned from

performing hereunder. Your waiver of any right to demand performance hereunder shall not be a waiver of any subsequent or other right to demand performance hereunder.

This Guaranty shall be governed by, or construed in accordance with, the laws of the State of Connecticut. This Guaranty shall bind the undersigned's successors and assigns and the benefits thereof shall extend to and include your successors and assigns. The undersigned will deliver to you its complete financial statements, certified by a recognized firm of certified public accountants, within one hundred and twenty (120) days of the close of each fiscal year of the undersigned. If you request, the undersigned will deliver to you copies of its quarterly financial reports certified by its chief financial officer, within ninety (90) days after the close of each fiscal quarter of the undersigned and copies of its most current tax returns. The undersigned will deliver to you copies of all Forms 10-K and 10-Q, if any, within 30 days after the dates on which they are filed with the Securities and Exchange Commission. In addition, in the eventof defaulthereunder, you may at any time inspect undersigned's records. The undersigned represents, warrants and covenants that all financial statements delivered to you in connection with this Guaranty have been (and will be) prepared in accordance with generally accepted accounting principles, and since the date of the most recent financial statements or other financial information delivered to you, there has been no material adverse change in the undersigned's financial condition.

The undersigned hereby represents and warrants to you as of the date hereof that (i) the undersigned's execution, delivery and performance hereof does not and will not violate any judgment, order or law applicable to the undersigned, or constitute a breach of or default under any indenture, mortgage, deed of trust, or other agreement entered into by the undersigned with the undersigned's creditors or any other party other than (A) those that would not have a material adverse effect in the property, assets, business, operations or financial conditions of the undersigned, or materially impair the right or ability of the undersigned to carry on its operations substantially as now conducted or anticipated to be conducted in the future, and (B) those that have been, or will be prior to the Closing Date, obtained or made; (ii) no approval, consent or withholding of objections is required from any governmental authority or any other entity with respect to the execution, delivery and performance by the undersigned of this Guaranty; (iii) this Guaranty constitutes a valid, legal and binding obligation of the undersigned, enforceable in accordance with its terms; (iv) there are no proceedings presently pending or, to the best of the undersigned's knowledge, having made reasonable inquiry, threatened against the undersigned which will materially impair its ability to perform under this Guaranty; (v) since the date of the undersigned's most recent financial statement, there has been no material adverse change in the financial condition of the undersigned; and (vi) the undersigned is and will remain in (A) material compliance with all laws and regulations solely to the extent applicable to it and (B) in full compliance (solely to the extent applicable to it) with all laws and regulations requiring that it neither is nor shall be (Y) listed on the Specially Designated Nationals and Blocked Person List maintained by the Office of Foreign Assets Control ("OFAC"), Department of the Treasury, and/or any other similar lists maint

If any provisions of this Guaranty are in conflict with any applicable statute, rule or law, then such provisions shall be deemed null and void to the extent that they may conflict therewith, but without invalidating any other provisions hereof.

THE UNDERSIGNED IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE STATE OF CONNECTICUT TO HEAR AND DETERMINE ANY SUIT, ACTION OR PROCEEDING AND TO SETTLE ANY DISPUTES, WHICH MAY ARISE OUT OF OR IN CONNECTION HEREWITH AND WITH THE ACCOUNT DOCUMENTS (COLLECTIVELY, THE "PROCEEDINGS"), AND THE UNDERSIGNED FURTHER IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE TO REMOVE ANY SUCH PROCEEDINGS FROM ANY SUCH COURT (EVEN IF REMOVAL IS SOUGHT TO ANOTHER OF THE ABOVE-NAMED COURTS). THE UNDERSIGNED IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MIGHT NOW OR HEREAFTER HAVE TO THE ABOVE-NAMED COURTS BEING NOMINATED AS THE EXCLUSIVE FORUM TO HEAR AND DETERMINE ANY SUCH PROCEEDINGS AND AGREES NOT TO CLAIM THAT IT IS NOT PERSONALLY SUBJECT TO THE JURISDICTION OF THE ABOVE-NAMED COURTS FOR ANY REASON WHATSOEVER, THAT IT OR ITS PROPERTY IS IMMUNE FROM LEGAL PROCESS FOR ANY REASON WHATSOEVER, THAT ANY SUCH COURT IS NOT A CONVENIENT OR APPROPRIATE FORUM IN EACH CASE WHETHER ON THE GROUNDS OF VENUE OR FORUM NON-CONVENIENS OR OTHERWISE. THE UNDERSIGNED ACKNOWLEDGES THAT BRINGING ANY SUCH SUIT, ACTION OR PROCEEDING IN ANY COURT OTHER THAN THE COURTS SET FORTH ABOVE WILL CAUSE IRREPARABLE HARM TO YOU WHICH COULD NOT ADEQUATELY BE COMPENSATED BY MONETARY DAMAGES, AND, AS SUCH, THE UNDERSIGNED AGREES THAT, IN ADDITION TO ANY OF THE REMEDIES TO WHICH YOU MAY BE ENTITLED AT LAW OR IN EQUITY, YOU WILL BE ENTITLED TO AN INJUNCTION OR INJUNCTIONS (WITHOUT THE POSTING OF ANY BOND AND WITHOUT PROOF OF ACTUAL DAMAGES) TO ENJOIN THE PROSECUTION OF ANY SUCH PROCEEDINGS IN ANY OTHER COURT. Notwithstanding the foregoing, you and the undersigned shall have the right to apply to a court of competent jurisdiction in the United States of America or abroad for equitable relief as is necessary to preserve, protect and enforce its respective rights under this Guaranty and the Account Documents, including, but not limited to orders of attachment or injunction necessary to maintain the status quo pending litigation or to enforce

judgments against the undersigned, the Customer or the collateral pledged to you pursuant to any Account Document or to gain possession of such collateral.

Each person signing this Guaranty on behalf of the undersigned company warrants that (i) it is to the benefit of the undersigned company to execute this Guaranty, (ii) the benefit to be received by the undersigned company from this Guaranty is reasonably worth the obligations thereby guaranteed, and (iii) he/she/it has authority to sign on behalf of such undersigned and by so signing, to bind said company hereunder.

IN WITNESS WHEREOF, this Guaranty is executed the day and year above written.

MasTec, Inc.

By: /s/ C. R. Campbell

(Signature)

Title: Executive Vice President and CFO

(Officer's Title)

Federal Tax ID: 65-0829355

ATTEST: /s/ Alberto de Cardenas

Alberto de Cardenas Secretary/Assistant Secretary